Case 11-13603-PJW Doc 2153 5164 01/30/14 Docket #2152 Date Filed: 1/30/2014

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re:

Chapter 11

BLITZ U.S.A., Inc., et al., 1

Case No. 11-13603 (PJW)

Debtors.

(Jointly Administered)

Re: Docket Nos. 2007, 2008 \$ 2005-2014

FINDINGS OF FACT, CONCLUSIONS OF LAW AND ORDER CONFIRMING DEBTORS' AND OFFICIAL COMMITTEE OF UNSECURED CREDITORS' FIRST AMENDED JOINT PLAN OF LIQUIDATION

A HEARING HAVING BEEN HELD BEFORE THE COURT AND CONCLUDED on January 28, 2014 (the "Confirmation Hearing"), to consider confirmation of the Debtors' and Official Committee of Unsecured Creditors' First Amended Joint Plan of Liquidation, dated December 18, 2013 (including all exhibits thereto and as modified, the "Plan"), proposed by the debtors and debtors-in-possession in the above-captioned cases (collectively, the "Debtors"), together with the Official Committee of Unsecured Creditors (the "Committee," and collectively, the "Proponents);

IT APPEARING TO THE COURT that the Disclosure Statement for the Debtors' and Official Committee of Unsecured Creditors' First Amended Joint Plan of Liquidation, dated December 18, 2013 [Docket No. 2000-2] (the "Disclosure Statement") was approved by the

² Capitalized terms used herein without definition have the meanings provided for in the Plan. In addition, any term used in the Plan or this Confirmation Order that is not defined in the Plan or this Confirmation Order, but that is used in the Bankruptcy Code or the Bankruptcy Rules, shall have the meaning assigned to that term in the Bankruptcy Code or the Bankruptcy Rules.



PAC 1137770v.3

1113603140130000000000000

¹ The Debtors in these chapter 11 cases, along with the last four digits of each Debtor's federal tax identification number, include: LAM 2011 Holdings, LLC (8742); Blitz Acquisition Holdings, Inc. (8825); Blitz Acquisition, LLC (8979); Blitz RE Holdings, LLC (9071); Blitz U.S.A., Inc. (8104); and MiamiOK LLC (2604). The location of the Debtors' corporate headquarters and the Debtors' service address is: 309 North Main Street, Miami, OK 74354.

Court on December 18, 2013, pursuant to the Order (A) Approving the Disclosure Statement; (B) Approving Form and Manner of Notice of Confirmation Hearing; (C) Approving Procedures for the Solicitation and Tabulation of Votes to Accept or Reject the Plan; (D) Estimating Each Blitz Personal Injury Claim at \$1.00 for Voting Purposes; (E) Approving Notice and Objection Procedures in Respect Thereof and (F) Granting Related Relief [Docket No. 2005] (the "Disclosure Statement Order"), which, among other things, also established procedures for the solicitation and tabulation of votes to accept or reject the Plan, scheduled a hearing on confirmation of the Plan and approved related notice procedures;

IT FURTHER APPEARING TO THE COURT that solicitation and noticing procedures with respect to the Plan approved by the Court pursuant to the Disclosure Statement Order have been followed as set forth in the Certification of P. Joseph Morrow IV with Respect to the Tabulation of Votes on the Debtors' and Official Committee of Unsecured Creditors First Amended Joint Plan of Liquidation sworn to on January 23, 2014 [Docket No. 2111] (the "Voting Declaration"); and affidavits of service having been executed by Kurtzman Carson Consultants, LLC (the "Balloting Agent") with respect to the mailing of notice of the Confirmation Hearing and solicitation materials in respect of the Plan in accordance with the Disclosure Statement Order (collectively, the "Affidavits of Service") and having been filed with the Court [Docket Nos. 2025, 2026, 2027, 2028, 2029, 2037, 2054 and 2082];

IT FURTHER APPEARING TO THE COURT that on January 6, 2014, the Proponents filed with the Court, as Exhibit 3 to the Plan, the form of the Blitz Liquidating Trust Agreement [Docket No. 2041], together with any other documents necessary to implement the Plan, the "Plan Documents");

IT FURTHER APPEARING TO THE COURT that on January 15, 2014, the Proponents filed the *Notice of Rescheduling of Confirmation Hearing Date and Time* [Docket No. 2065] stating that the hearing to consider confirmation of the Plan had been adjourned to January 28, 2014 at 9:30 a.m. (Eastern Time) before the Honorable Peter J. Walsh, Judge of the Bankruptcy Court, in the United States Courthouse, at 824 Market Street, Wilmington, Delaware 19801.³

IT FURTHER APPEARING TO THE COURT that the deadline for filing objections to the Plan has passed;

IT FURTHER APPEARING TO THE COURT that the deadline for casting ballots to accept or reject the Plan has passed and that the results of voting have been certified by the Balloting Agent, acting as voting agent for the balloting permitted for Classes 3(a) and 3(b) (Blitz Personal Injury Trust Claims), and Classes 4(a) and 4(b) (General Unsecured Claims), pursuant to the Disclosure Statement Order and as set forth in the Voting Declaration;

IT FURTHER APPEARING TO THE COURT that the following have been filed or offered in support of the Plan: (a) the Declaration of Rocky Flick in Support of Confirmation of the Debtors' and Official Committee of Unsecured Creditors' First Amended Joint Plan of Liquidation [D.I. 2133] (the "Flick Declaration"); (b) the Declaration of Ken McClain in Support of Confirmation of the Debtors' and Official Committee of Unsecured Creditors' First Amended Joint Plan of Liquidation [D.I. 2135] (the "McClain Declaration"); (c) Declaration of Brian Mottet on behalf of Wal-Mart Stores, Inc. with respect to the Debtors' and Official Committee of Unsecured Creditors' First Amended Joint Plan of Liquidation [D.I. 2114] (the "Mottet Declaration"); (d) the Declaration of Christian Michalik in Support of he Debtors' and Official Committee of Unsecured Creditors' First Amended Joint Plan of Liquidation [D.I. 2104]

³ The confirmation hearing time was subsequently moved to 2:30 p.m. at the request of the parties to allow for a mediation session before Chief Judge Kevin Gross, Judge of the Bankruptcy Court, among the Proponents and the remaining objectors.

(the "Michalik Declaration"); (e) the Declaration of Timothy P. Harkness in Support of the Debtors' and Official Committee of Unsecured Creditors' First Amended Joint Plan of Liquidation [D.I. 2105] (the "Harkness Declaration"); (f) the Declaration of Robert Elmburg in Support of the Debtors' and Official Committee of Unsecured Creditors' First Amended Joint Plan of Liquidation [D.I. 2108] (the "Elmburg Declaration"); (g) the Declaration of James A. Pearson in Support of Confirmation of the Debtors' and Official Committee of Unsecured Creditors' First Amended Joint Plan of Liquidation [D.I. 2134] (the "Pearson Declaration"); (h) the Declaration of Jeffrey Roberts in Support of Confirmation of the Debtors' and Official Committee of Unsecured Creditors' First Amended Joint Plan of Liquidation [D.I. 2118] (the "Roberts Declaration"); (i) the Declaration of Steve Snyder in Support of Confirmation of Debtors' and Official Committee of Unsecured Creditor's First Amended Joint Plan of Liquidation [D.I. 2147] (the "Snyder Declaration," and with the Flick Declaration, McClain Declaration, Mottet Declaration, Michalik Declaration, Harkness Declaration, Elmburg Declaration, Pearson Declaration, Roberts Declaration, collectively, the "Declarations"); (j) the Expert Report of James Marshall, Jr. [D.I. 2113] (the "Marshall Report"); (k) the Expert Report of Denise Neumann Martin [D.I. 2112] (the "Martin Report"); (1) that certain Bates White Report Regarding the Participating Insurer Settlement admitted as evidence at the hearing on the Insurance Settlement Motion (the "Bates White Report," and with the Marshall Report and the Martin Report, collectively the "Expert Reports"); and (m) the record from the December 18, 2013 hearing approving the Insurance Settlement Motion (the "December 18, **2013 Record**");

IT FURTHER APPEARING TO THE COURT that the following objections to confirmation of the Plan have been filed: (a) Objection of Newby Claimant to Confirmation of

the Debtors' and Official Committee of Unsecured Creditors' First Amended Plan of Liquidation [D.I. 2090]; (b) Objection of Estate of Joseph M. Cataldi and Lori Cataldi (as Guradian, Parent, and Natural Guardian for Minors Michael Cataldi and Brianna Cataldi) to Debtors' and Official Committee of Unsecured Creditors' First Amended Joint Plan of Liquidation [D.I. 2091]; (c) Objection to Debtors' and Official Committee of Unsecured Creditors' First Amended Joint Plan of Liquidation [D.I. 2092]; and (d) The United States Trustee's Objection to Debtors' and Official Committee of Unsecured Creditors' First Amended Joint Plan of Liquidation [D.I. 2016] (collectively, the "Objections"), each of which was withdrawn and resolved on the record at the Confirmation Hearing.

NOW, THEREFORE, based upon the Court's review of (a) the Disclosure Statement, (b) the Plan, (c) the Plan Documents, (d) the Declarations and the Expert Reports, each of which was admitted into the record without objection, (e) the Voting Declaration, (f) all of the other evidence proffered or adduced, filings and arguments of counsel during these Chapter 11 Cases and the Confirmation Hearing, including the December 18, 2013 Record; and after due deliberation thereon and good cause appearing therefor, and for the reasons set forth on the record at the Confirmation Hearing:

IT IS HEREBY FOUND AND DETERMINED THAT:

A. <u>Findings of Fact and Conclusions of Law</u>. The findings of fact and the conclusions of law stated in this Confirmation Order and on the record at the Confirmation Hearing constitute the Court's findings of fact and conclusions of law pursuant to Bankruptcy Rule 7052, made applicable to the proceeding by Bankruptcy Rule 9014. To the extent any finding of fact shall be determined to be a conclusion of law, it shall be so deemed, and to the extent any conclusion of law shall be determined to be a finding of fact, it shall be so deemed.

- B. <u>Jurisdiction</u>; <u>Venue</u>; <u>Core Proceeding</u>. This Court has jurisdiction over the Debtors' Chapter 11 Cases pursuant to 28 U.S.C. §§ 157 and 1334. Venue in the District of Delaware was proper as of the Petition Date and continues to be proper pursuant to 28 U.S.C. §§ 1408 and 1409. Confirmation of the Plan is a core proceeding pursuant to 28 U.S.C. § 157(b)(2)(L) over which this Court has exclusive jurisdiction.
- C. <u>Transmittal and Mailing of Solicitation Materials and Notices</u>. The solicitation materials and notices prescribed by the Disclosure Statement Order were served in compliance with the Disclosure Statement Order, and such service was adequate and sufficient. Adequate and sufficient notice of the Confirmation Hearing and the other deadlines and matters required to be noticed pursuant to the Disclosure Statement Order was given in compliance with the Bankruptcy Rules and the Disclosure Statement Order, and no other or further notice is or shall be required.
- D. Adequacy of Voting Procedures. All procedures used to distribute the solicitation materials to the appropriate creditors entitled to vote on the Plan and to tabulate the ballots returned by creditors were fair and were conducted in accordance with the applicable provisions of the Bankruptcy Code, the Bankruptcy Rules and the Disclosure Statement Order. Votes for acceptance or rejection of the Plan were solicited and cast in good faith, and only after transmittal of the approved Disclosure Statement, and otherwise in compliance with Bankruptcy Code sections 1125 and 1126 and Bankruptcy Rules 3017 and 3018. As evidenced by the Voting Declaration, all Ballots were properly tabulated.
- E. <u>Good Faith Solicitation 11 U.S.C. § 1125(e)</u>. Based on the record before the Court in the Chapter 11 Cases, the Proponents, and any of their respective directors, officers, employees, members, participants, agents, representatives, partners, affiliates, counsel, other

advisors, successors or assigns, have acted in good faith within the meaning of Bankruptcy Code sections 1125(e) and 1129(a)(3), and in compliance with the applicable provisions of the Bankruptcy Code, the Bankruptcy Rules and the Disclosure Statement Order in connection with all of their respective activities relating to the solicitation of acceptances of the Plan and their participation in the activities described in Bankruptcy Code section 1125, and are entitled to the protections afforded by Bankruptcy Code section 1125(e).

- F. Impaired Class that Has Voted to Accept or Reject the Plan. Classes 3(a), 3(b), 4(a) and 4(b) are impaired and, as evidenced by the Voting Declaration, which certified both the method and results of the voting, have voted to accept the Plan pursuant to the requirements of Bankruptcy Code sections 1125 and 1126. Thus, at least one impaired Class of Claims has voted to accept the Plan.
- G. <u>Classes Deemed to have Accepted or Rejected the Plan</u>. Classes 1(a), 1(b), 2(a) and 2(b) are not impaired under the Plan and are deemed to have accepted the Plan pursuant to Bankruptcy Code section 1126(f). Classes 5(a), 5(b), 6(a) and 6(b) are not expected to receive any distributions under the Plan and are deemed to have rejected the Plan pursuant to Bankruptcy Code section 1126(g).
- H. Releases, Channeling Injunction, Exculpation and Indemnification Provisions. The Releases and Channeling Injunction with respect to each Protected Party, as set forth in Sections 4.3.3, 4.34 and 7.2 and (ii) exculpation and limitation of liability with respect to each Exculpated Party, as set forth in Section 15.4 of the Plan are appropriate under applicable law. The Court's findings of fact to support the approval of the Releases and Channeling Injunction, based on the record established at the Confirmation Hearing, including the Declarations, the Expert Reports, and the December 18, 2013 Record are set forth below.

(1) Fairness.

- (a) The Participating Insurers, on behalf of themselves and all entities covered under the Participating Insurer Policies, are contributing \$137,540,639.36 to the Blitz Personal Injury Trust, through the buy-back of the Participating Insurer Policies. Moreover, the Participating Insurers have previously provided the Debtors' Estates with approximately \$200,000 to fund a portion of the notice costs in connection with the Blitz Personal Injury Claim Bar Date Order. This payment, and the compromises by the Participating Insurers providing for the payment, is a substantial contribution and is critical to the Plan.
- (b) In connection with the Insurance Settlement, Wal-Mart is contributing \$26,429,360.64 to the Blitz Personal Injury Trust. Wal-Mart contends that the amount of cash it contributed to fund the Blitz Personal Injury Trust is in excess of the contribution, if any, needed to satisfy the self-insured retention obligations and activate coverage under the Participating Insurance Policies. In addition, Wal-Mart is contributing \$1.54 million to the Blitz Liquidating Trust to fund distributions to holders of Administrative Expense Claims and General Unsecured Claims against the USA Debtors' Estates. Moreover, Wal-Mart has previously provided the Debtors' Estates with approximately \$100,000 to fund a portion of the notice costs in connection with the Blitz Personal Injury Claim Bar Date Order. Finally, Wal-Mart has agreed to waive significant claims against the Debtors' Estates and agreed to relinquish valuable insurance rights and against the Participating Insurer Policies consented to the Insurance Buy-Back in connection with the Insurance Settlement. The consideration provided by Wal-Mart constitutes a substantial contribution and is critical to the Plan.
- (c) In connection with the BAH Settlement, Blitz Acquisitions Holdings, Inc., on behalf of the BAH Released Parties, is paying \$6.25 million (plus up to an additional \$250,000 to pay the Flick Claim, to the extent it is allowed) to the USA Debtors' Estates. Without this infusion of funds, the USA Debtors would be administratively insolvent. The BAH Settlement Payment would not be possible without the consent of all the BAH Released Parties. As additional consideration, Mr. Michalik, Mr. Aurelio, Kinderhook Capital Fund II, LP and Kinderhook Industries II, LP are also waiving their claims against the USA Debtors. The consideration provided by the BAH Released Parties constitutes a substantial contribution and is critical to the Plan.
- (d) As reflected in the Voting Declaration, 95.29% (81 out of 85) of the holders of Blitz Personal Injury Claims in Class 4(a) of the Plan that voted on the Plan and 100% (3 out of 3) of the holders of Blitz Personal Injury Claims in Class 4(b) of the Plan have voted to accept the Plan. On the record at the Confirmation Hearing, three of the parties that had originally voted against confirmation of the Plan announced that they were changing their votes to votes in support of the Plan. As a result of those announcements, only one vote against confirmation of the Plan in Class 4(a) remains, and that claimant did not object to confirmation of the Plan. Accordingly, the Plan has been overwhelmingly accepted by the classes of creditors affected by the Releases and the Channeling Injunction and there is no remaining objection to confirmation of the Plan.

- Expert testimony as to the valuation of the pending Covered Blitz Personal Injury Claims in the tort system was provided at the Confirmation Hearing by The Court finds this evidence credible. The Court finds this evidence consistent with the evidence adduced in the December 18, 2013 Record. No contrary evidence was provided in the record. The aggregate value of the Covered Blitz Personal Injury Claims would be \$102.9 million if those claims were litigated in the tort system. The Blitz Personal Injury Trust is being funded with \$163,970,000 in funds from the Insurance Settlement Payment. Further, of the 61 Covered Blitz Personal Injury Claims projected to be included in the Non-Appealing Fund, the aggregate value of those claims, if litigated in the tort system would be \$96.8 million. Similarly, of the 32 pending Covered Blitz Personal Injury Claims projected to be eligible for participation in the Special Circumstances Fund, the aggregate value of those claims, if litigated in the tort system, would be \$6.1 million. The Blitz Personal Injury Trust will have sufficient assets on the Effective Date to process and pay Covered Blitz Personal Injury Trust Claims in accordance with the terms of the Plan, the Blitz Liquidating Trust Agreement and the Blitz Personal Injury TDP. The Plan, through the Blitz Personal Injury Trust Agreement and the Blitz Personal Injury TDP, has criteria, rules and procedures that are reasonable and equitable to process, evaluate, allow or disallow, and pay (where applicable) Covered Blitz Personal Injury Trust Claims in a similar and fair manner. Because the allocations contained in the Blitz Personal Injury TDP provide substantially greater recoveries to holders of Covered Blitz Personal Injury Claims than their projected recoveries in the tort system, the holders of Covered Blitz Personal Injury Claims are receiving fair consideration in exchange for the Releases and the Channeling Injunction.
- (f) The Blitz Personal Injury TDP provides a fair and equitable mechanism for holders of Covered Blitz Personal Injury Claims to reject the offers they receive from the Blitz Personal Injury Trust and proceed to mediation/arbitration, and ultimately the tort system, to liquidate their respective Covered Blitz Personal Injury Claims.
- (g) Holders of Blitz Personal Injury Trust Claims arising prior to the Release Date retain their rights under the Assigned Blitz Insurance Policies issued by the Non-Participating Insurers. Accordingly, these claimants are receiving fair consideration in exchange for the Releases and the Channeling Injunction.
- (2) Necessity to the Reorganization. The Releases and the Channeling Injunction are critical to the success of the Plan. Without the Releases and the Channeling Injunction, the Protected Parties are not willing to make their contributions under the Plan. Moreover, the BAH Settlement and the Insurance Settlement are mutually dependent on each other and neither can be consummated without the other. Absent those contributions, the Plan is not feasible.
- (3) Record Supports Specific Findings. The record of the Confirmation Hearing and the Chapter 11 Cases is sufficient to support the Release, Channeling Injunction and exculpation provisions contained in the Plan.

- **(4)** Extraordinary Circumstances. These Chapter 11 Cases were precipitated by a significant number of personal injury claims arising out of accidents involving the Debtors' products. The increasing number of Blitz Personal Injury Claims ultimately drove the Debtors into bankruptcy where they were forced to sell substantially all their assets. The Debtors' primary remaining assets, the proceeds from the Blitz Insurance Policies, are the subject of numerous substantial coverage issues, including whether all of the self-insured retentions had been exhausted to trigger coverage under certain of the Participating Insurer Policies. Moreover, Wal-Mart vigorously defended its rights as an additional insured under the Participating Insurer Policies both pre- and post-petition. After months of arms-length, contentious negotiations before Chief Judge Kevin Gross and Judge Richard A. Cohen (Ret.), the Debtors, the Committee and the Protected Parties came to terms on a global settlement that formed the foundation of the Plan. These extensive efforts and substantial contributions by each of the Protected Parties, along with the magnitude of the recoveries, provided under the Plan to holders of Allowed Claims constitute extraordinary circumstances warranting the Releases and Channeling Injunction set forth in the Plan.
- (5) *Identity of Interest*. There is a substantial identity of interest between the Debtors and the Protected Parties, such that a Blitz Personal Injury Claim asserted against a Protected Party is essentially a claim against the Debtors.
 - (a) Participating Insurers. The Participating Insurers are the Debtors' insurance providers under the Participating Insurer Policies. The Blitz Insurance Policies are the primary remaining assets of the Debtors' Estates. Accordingly, any Blitz Personal Injury Claim asserted against the Debtors, if successful, would ultimately be paid from the proceeds of the Blitz Insurance Policies (subject to the policies' terms and conditions and any applicable coverage defenses). To the extent any third party has a claim against the Participating Insurers, such claim must arise out of an incident subject to coverage under one or more of the Participating Insurer Policies. Because any claim successfully asserted against a Participating Insurer will be paid from the proceeds of the Participating Insurer Policies, a claim against a Participating Insurer will directly reduce the coverage available to the Debtors and any other insureds under those policies. Thus, a claim against a Participating Insurer is essentially a claim against the Debtors.
 - (b) Wal-Mart. Each of the supplier agreements in place between Wal-Mart and Blitz U.S.A., Inc. in effect during the period from July 31, 2007 through July 31, 2012 contains a clause that requires (i) Blitz U.S.A., Inc. to defend and indemnify Wal-Mart from all claims and litigation relating to products manufactured by Blitz U.S.A, Inc.; and (ii) that Blitz U.S.A., Inc. is required to obtain or extend its general liability insurance policies to cover Wal-Mart as a named insured or as an insured under a general endorsement. As a result, Wal-Mart is an additional insured under the Participating Insurer Policies. Accordingly, suits against Wal-Mart are covered by the Debtors' insurance and would deplete the assets of the Debtors' Estates.
 - (c) BAH Released Parties. Each of the BAH Released Parties have direct or indirect indemnification and/or coverage rights against the Debtors in connection with any alleged liability in connection with Blitz Personal Injury Claims, arising out of one of more of the following: (i) the Management Services Agreement

dated as of September 21, 2007 among Blitz USA, Inc., Blitz Holdings, LLC (n/k/a LAM 2011 Holdings, LLC) and Kinderhook Industries II, L.P.; (ii) applicable insurance policies; (iii) specific actions or resolutions of the Debtors' Boards of Directors; (iv) certificates of incorporation, certificates of limited partnership, articles of organization or certificates of formation (as applicable); (v) bylaws and operating agreements of the Debtors; or (vi) statutory or common law. Claims asserted by any of the BAH Released Parties under any of the foregoing bases would deplete the assets of the Debtors' Estates.

- (d) Unified Interest. Each of the Protected Parties share a common goal of resolving their competing and interrelated claims and achieving a fair and equitable distribution of the Debtors' remaining assets through the Plan. Absent the involvement of each of the Protected Parties, none of the settlements required for, or entered into in connection with, the confirmation of the Plan would be possible, and few, if any, assets would be available for distribution to the Debtors' Creditors.
- I. <u>Compromise and Settlement.</u> Pursuant to Bankruptcy Rule 9019, and in consideration for the classification, distribution and other benefits provided under the Plan, the provisions of the Plan constitute a good faith compromise and settlement of all Claims and controversies resolved pursuant to the Plan, including, without limitation, all Claims arising prior to the Petition Date, whether known or unknown, foreseen or unforeseen, asserted or unasserted, by or against the Debtors, arising out of, relating to or in connection with the business or affairs of or transactions with the Debtors. The Court's specific findings of fact to support the approval of the Insurance Settlement and the BAH Settlement, based on the record established at the Confirmation Hearing, including the Declarations, the Expert Reports and the December 18, 2013 Record, are set forth below.
- (1) The Debtors and the Committee performed sufficient due diligence into the scope of the Debtors' available insurance coverage and the likelihood of recovery with respect to Blitz Personal Injury Claims against the Debtors.
- (2) The Participating Insurers have asserted the position that certain policies cannot be accessed to satisfy Blitz Personal Injury Claims if litigated in the tort system due to failure to exhaust the self-insured retentions ("SIRs") of these certain policies. The Blitz Personal Injury Claims present serious questions concerning the liability of each respective Blitz Insurer and the availability of coverage under the Blitz Insurance Policies.
- (3) In the absence of each of the Protected Parties being protected by the Releases and the Channeling Injunction, the continuation of litigation of Blitz Personal Injury

Claims in the tort system would result in the incurrence of costs that would materially diminish the assets available to satisfy Blitz Personal Injury Trust Claims.

- (4) The attorneys representing the holders of Blitz Personal Injury Claims appointed to the Committee represent an appropriate cross-section of diverse types of injuries and are adequately representative of all holders of Blitz Personal Injury Claims.
- (5) The Committee communicated with holders of Blitz Personal Injury Claims not appointed to the Committee regarding the terms of Blitz Personal Injury TDP and TDP Scoring System.
- (6) All communications between the Committee and its representatives, on the one hand, and any holder of a Blitz Personal Injury Claim, on the other, were undertaken in good faith.
- (7) A Vendor, officer, director and/or shareholder of Blitz sued on a Blitz Personal Injury Claim might be entitled to coverage under the Participating Insurer Policies. If any of the foregoing are insureds under the respective policies, they could tender the Blitz Personal Injury Claims to the Participating Insurers for defense and indemnity. Accordingly, any settlement relating to, and the buy-back of, the Participating Insurer Policies must also resolve all claims against Vendors, officers, directors and shareholders to the same extent such claims are being resolved against the Debtors, which justifies the inclusion of the Vendors, officers, directors and shareholders as Protected Parties under the Plan.
- (8) The potential indemnification claims of Wal-Mart with respect to the Participating Insurer Policies, if successful, would materially deplete the Debtors' available insurance coverage and reduce the assets otherwise available to satisfy Blitz Personal Injury Claims.
- (9) Based on the settlements and compromises set forth in the Plan, including the Insurance Settlement, Wal-Mart consented to the Participating Insurers' buyback of the Participating Insurer Policies under 11 U.S.C. § 363(f)(2) thus avoiding any need for litigation of issues under other subparagraphs of 11 U.S.C. § 363(f).
- (10) Wal-Mart's contributions to the Plan facilitate/provide for payment to Blitz Personal Injury Claims which do not assert liability against Wal-Mart.
- (11) The Participating Insurer Policies are property of the Debtors and the Debtors' estates pursuant to 11 U.S.C. § 541(a).
- (12) With the exception of the Blitz Insurance Policies, the Debtors' estates possess few to no assets to fund a recovery for Blitz Personal Injury Trust Claims.
- (13) The Debtors appropriately exercised their business judgment in connection with the negotiation and solicitation of the Plan, including considering the input of the Committee with respect to the terms of the Insurance Settlement, the BAH Settlement, the Blitz Personal Injury Trust Agreement and the Blitz Personal Injury TDP.

- (14) The distributions to be provided on account of Allowed Covered Blitz Personal Injury Claims represent a better opportunity to recover more funds at a faster pace than would be available by liquidation through the tort system.
- (15) The USA Debtors, the BAH Debtors and the Committee performed sufficient due diligence into the claims by and between the USA Debtors on the one hand, and the BAH Debtors and their non-debtor subsidiaries and affiliates on the other, including consideration of the complexity, expense and delay of any litigation resolving such claims and the likelihood of success in such litigation.
- (16) The compromises and settlements, including the Insurance Settlement and the BAH Settlement, are in the best interests of the Debtors, the Estates, creditors and other parties in interest, and are fair, equitable and within the range of reasonableness.
- J. <u>Deemed Consolidation</u>. Based on, among other things, the Disclosure Statement, the Flick Declaration and the record of the Confirmation Hearing, no class of creditors or interest holders is disadvantaged by the deemed substantive consolidation of the USA Debtors and the BAH Debtors, respectively.
- K. <u>Plan Compliance with Bankruptcy Code -- 11 U.S.C. § 1129(a)(1)</u>. The Plan complies with the applicable provisions of the Bankruptcy Code, thereby satisfying Bankruptcy Code section 1129(a)(1).
- (1) Proper Classification—11 U.S.C. §§ 1122, 1123(a)(1). Aside from Administrative Expense Claims and Priority Claims, which need not be classified, the Plan designates twelve (12) Classes of Claims and Equity Interests. The Claims and Equity Interests placed in each Class are substantially similar to other Claims and Equity Interests, as the case may be, in each such Class. Valid business, factual and legal reasons exist for separately classifying the various Classes of Claims and Equity Interests created under the Plan, and such Classes do not unfairly discriminate among holders of Claims and Equity Interests. Thus, the Plan satisfies Bankruptcy Code sections 1122 and 1123(a)(1).
- (2) Specified Unimpaired Classes—11 U.S.C. § 1123(a)(2). Article III of the Plan specifies that Classes 1(a), 1(b), 2(a), and 2(b) are unimpaired under the Plan, thereby satisfying Bankruptcy Code sections 1123(a)(2).
- (3) Specified Treatment of Impaired Classes—11 U.S.C. § 1123(a)(3). Article III of the Plan designates Classes 3(a), 3(b), 4(a), 4(b), 5(a), 5(b), 6(a), and 6(b) as impaired and specifies the treatment of Claims and Equity Interests in those Classes, thereby satisfying Bankruptcy Code section 1123(a)(3).

- (4) No Discrimination—11 U.S.C. § 1123(a)(4). The Plan provides for the same treatment by the Debtors for each Claim or Equity Interest in each respective Class unless the holder of a particular Claim or Equity Interest has agreed to a less favorable treatment of such Claim or Equity Interest, thereby satisfying Bankruptcy Code sections 1123(a)(4).
- (5) Implementation of Plan—11 U.S.C. § 1123(a)(5). Articles IV, V, VI, VII and XII of the Plan provide adequate and proper means for its implementation, thereby satisfying Bankruptcy Code sections 1123(a)(5).
- (6) Non-Voting Equity Securities—11 U.S.C. § 1123(a)(6). The Plan, as a plan of liquidation, contemplates that all Equity Interests of the USA Debtors shall be cancelled as of the Effective Date and that a single share of BAH shall be issued to Kinderhook Capital Fund II, L.P. for purposes of complying with Delaware corporate law and effectuating the BAH Debtors' liquidation under the Plan. As a result thereof, the Plan complies with section 1123(a)(6) of the Bankruptcy Code.
- (7) Selection of Officers and Directors—11 U.S.C. § 1123(a)(7). The initial trustees of both the Blitz Liquidating Trust and the Blitz Personal Injury Trust have been identified. Similarly, the BAH Plan Administrator has been identified and will be vested with authority to act on behalf of the BAH Debtors post-confirmation. These appointments are consistent with the best interests of holders of Claims and Equity Interests and public policy. Accordingly, the requirements of Bankruptcy Code section 1123(a)(7) are satisfied.
- (8) Additional Plan Provisions—11 U.S.C. § 1123(b). The Plan's additional provisions are appropriate and not inconsistent with the applicable provisions of the Bankruptcy Code. The failure to specifically address a provision of the Bankruptcy Code in this Confirmation Order shall not diminish or impair the effectiveness of this Confirmation Order.
 - (a) Impairment of Claims and Interests and Assumption, Assumption and Assignment or Rejection of Executory Contracts and Unexpired Leases—11 U.S.C. § 1123(b)(1)-(2). In accordance with Bankruptcy Code section 1123(b)(1), Article III of the Plan impairs or leaves unimpaired, as the case may be, each Class of Claims and Equity Interests. In accordance with Bankruptcy Code section 1123(b)(2), Article VIII of the Plan provides for the rejection of all executory contracts and unexpired leases of the Debtors that have not expired by their own terms as of the Effective Date, except for those executory contracts and unexpired leases that have been assumed, assumed and assigned or rejected pursuant to previous orders of the Bankruptcy Court. The Plan is therefore consistent with Bankruptcy Code section 1123(b)(1)-(2).
 - (b) Modification of the Rights of Holders of Claims—11 U.S.C. § 1123(b)(5). Article III of the Plan modifies or leaves unaffected, as the case may be, the rights of holders of each Class of Claims, and therefore, the Plan is consistent with Bankruptcy Code section 1123(b)(5).
 - (c) Other Provisions Not Inconsistent with Applicable Provisions of the Bankruptcy Code—11 U.S.C. § 1123(b)(6). The Plan includes additional appropriate provisions that are not inconsistent with applicable provisions of the Bankruptcy Code,

- including: (a) the provisions of Article XII of the Plan regarding the means for executing and implementing the Plan; (b) the provisions of Article VIII of the Plan governing the treatment of executory contracts and unexpired leases; (c) the provisions of Article VI concerning the preservation of certain rights and defenses with respect to certain insurance policies; (d) the provisions of Article XII of the Plan with respect to the separate substantive consolidation of the USA Debtors and the BAH Debtors; and (e) the provisions of Article XIII of the Plan regarding retention of jurisdiction by the Court over certain matters after the Effective Date. The Plan is therefore consistent with Bankruptcy Code section 1123(b)(6).
- (9) Plan Compliance with Fed. R. Bankr. P. 3016. The Plan is dated and identifies the entities submitting it, thereby satisfying Bankruptcy Rule 3016(a). The filing of the Disclosure Statement with the Court satisfies Bankruptcy Rule 3016(b). Further, the Plan and Disclosure Statement describe in specific and conspicuous language all acts to be enjoined and identify the entities that are subject to the injunction, satisfying Bankruptcy Rule 3016(c) to the extent applicable.
- (10) Compliance with Fed. R. Bankr. P. 3017. The Proponents have given notice of the Confirmation Hearing as required by Bankruptcy Rule 3017(d) and the Disclosure Statement Order. The solicitation materials prescribed by the Disclosure Statement Order were transmitted to the creditors entitled to vote on the Plan in accordance with Bankruptcy Rule 3017(d).
- (11) Compliance with Fed. R. Bankr. P. 3018. The solicitation of votes to accept or reject the Plan satisfies Bankruptcy Rule 3018. The Plan was transmitted to all creditors entitled to vote on the Plan, sufficient time was prescribed for such creditors to accept or reject the Plan, and the solicitation materials used and solicitation procedures followed comply with Bankruptcy Code sections 1125 and 1126, thereby satisfying the requirements of Bankruptcy Rule 3018.
- L. <u>Compliance with Bankruptcy Code—11 U.S.C. § 1129(a)(2)</u>. The Debtors have complied with the applicable provisions of the Bankruptcy Code, thereby satisfying Bankruptcy Code section 1129(a)(2). Specifically:
 - (1) The Debtors are proper debtors under Bankruptcy Code section 109.
- (2) The Debtors have complied with all applicable provisions of the Bankruptcy Code, except as otherwise provided or permitted by order of the Court.
- (3) The Proponents have complied with the applicable provisions of the Bankruptcy Code, the Bankruptcy Rules and the Disclosure Statement Order, including, but not limited to, the provisions of Bankruptcy Code sections 1125 and 1126 in transmitting the Plan, the Disclosure Statement, the ballots and related documents and notices and in soliciting and tabulating votes to accept or reject the Plan.

- M. Plan Proposed in Good Faith—11 U.S.C. § 1129(a)(3). The Proponents have proposed the Plan in good faith and not by any means forbidden by law, thereby satisfying Bankruptcy Code section 1129(a)(3). The Proponents' good faith is evident from the Flick Declaration, the McClain Declaration and the record of these Chapter 11 Cases, including the record of the hearing to approve the Disclosure Statement and the record of the Confirmation Hearing. Based upon the evidence proffered at the Confirmation Hearing, the Court finds and concludes that the Plan has been proposed with the legitimate and honest purpose of liquidating the Debtors' assets and distributing any proceeds among the creditors. Moreover, the sufficiency of disclosure, the support of the Debtors' primary constituencies, and the overwhelming acceptance of the Plan by holders of Claims who voted on it, all provide independent evidence of the Proponents' good faith in proposing the Plan in compliance with Bankruptcy Code section 1129(a)(3). Further, the Plan's classification and treatment of Claims and Equity Interests, its compromise provisions, and its provisions respecting the Releases, Channeling Injunction and exculpation have been negotiated in good faith and at arms'-length and are consistent with Bankruptcy Code sections 105, 1122, 1123(b)(3)(A), 1123(b)(6), 1129, and 1142.
- N. Payments for Services or Costs and Expenses—11 U.S.C. § 1129(a)(4). Except as otherwise provided in the Plan or certain prior orders of the Court, any payments made or to be made for services or for costs and expenses incurred in connection with the Chapter 11 Cases are subject to the Court's approval. Fees to be paid to the Debtors' retained professionals are subject to the Court's approval, through the fee application process.
- O. <u>Directors, Officers and Insiders—11 U.S.C. § 1129(a)(5)</u>. Under the Plan, the Debtors' directors and officers are deemed to resign as of the Effective Date of the Plan. The trustees of both the Blitz Liquidating Trust and the Blitz Personal Injury Trust have been

identified. Similarly, the BAH Plan Administrator has been identified and will be vested with authority to act on behalf of the BAH Debtors post-confirmation. The Blitz Personal Injury Trustee and the Blitz Liquidating Trustee are not "insiders," as defined in section 101(31) of the Bankruptcy Code. As required by Section 12.4.2 of the Plan, the board of directors of BAH has chosen a current director of BAH, Louis Aurelio, as the BAH Plan Administrator. The nature of the BAH Plan Administrator's compensation, if any, shall be determined by the board of directors of BAH on or prior to the Effective Date. The Plan therefore complies with Bankruptcy Code section 1129(a)(5).

- P. <u>No Rate Changes—11 U.S.C. § 1129(a)(6)</u>. This section is inapplicable because there is no governmental regulatory commission that has jurisdiction over the rates that the Debtors charge.
- Q. Best Interests of Creditors—11 U.S.C. § 1129(a)(7). The Plan satisfies Bankruptcy Code section 1129(a)(7). The Flick Declaration and the McClain Declaration and other evidence proffered or adduced at the Confirmation Hearing: (1) are persuasive and credible, (2) have not been controverted by other evidence, and (3) establish that each holder of an impaired Claim or Equity Interest either has accepted the Plan or will receive or retain under the Plan, on account of such Claim or Equity Interest, property of a value, as of the Effective Date, that is not less than the amount that such holder would receive or retain if the Debtors were liquidated under chapter 7 of the Bankruptcy Code on such date. Furthermore, a liquidation under chapter 7 would adversely affect the ultimate proceeds available for distribution to all holders of Allowed Claims in the Chapter 11 Cases. Moreover, the increased costs associated with a liquidation under chapter 7 would substantially reduce the proceeds available for distribution. These costs would include, among other things, administrative fees and costs

payable to a trustee in bankruptcy and professional advisors to such trustee. Based upon the foregoing, the Plan satisfies the requirements of section 1129(a)(7) of the Bankruptcy Code. Therefore, the "best interests" test is satisfied with respect to each of these classes.

R. Deemed Acceptance or Rejection by Certain Classes—11 U.S.C. § 1129(a)(8).

- (1) Classes 1(a), 1(b), 2(a) and 2(b) are unimpaired and are conclusively presumed to have accepted the Plan under Bankruptcy Code section 1126(f). As to these Classes, Bankruptcy Code Section 1129(a)(8) has been satisfied.
- (2) Classes 3(a), 3(b), 4(a) and 4(b) are impaired by the Plan. At least two-thirds in amount and more than one-half in number of the Claims held by Creditors in Classes 3(a), 3(b), 4(a) and 4(b) have voted to accept the Plan, as established by the Voting Declaration, in accordance with Bankruptcy Code Section 1126(c). As to these Classes, Bankruptcy Code section 1129(a)(8) has been satisfied.
- (3) Classes 5(a), 5(b), 6(a) and 6(b) are impaired and not entitled to receive or retain any property under the Plan and, therefore, are deemed to have rejected the Plan pursuant to Bankruptcy Code section 1126(g). Although Bankruptcy Code section 1129(a)(8) has not been satisfied with respect to Classes 5(a), 5(b), 6(a) and 6(b), the Plan is confirmable because the Plan satisfies Bankruptcy Code section 1129(b) with respect to those Classes of Claims and Equity Interests, as set forth below.
- S. <u>Treatment of Administrative, Priority and Tax Claims—11 U.S.C. § 1129(a)(9)</u>. Through the provisions of the Plan governing Administrative Expense Claims and Priority Claims, the Plan provides for the payment in full, on the later of (a) the Effective Date, or as soon as reasonably practicable thereafter, and (b) as soon as reasonably practicable after the date such Claim becomes an Allowed Claim, of all Claims entitled to priority under Bankruptcy Code section 507(a). Accordingly, Bankruptcy Code section 1129(a)(9) is satisfied.
- T. Acceptance by Impaired Class—11 U.S.C. § 1129(a)(10). Classes 3(a), 3(b), 4(a) and 4(b) have voted to accept the Plan within the meaning of Bankruptcy Code section 1126(c), without the need to include any acceptance of any insider. Therefore, Bankruptcy Code section 1129(a)(10) is satisfied.

- U. <u>Feasibility—11 U.S.C. § 1129(a)(11)</u>. The evidence proffered or adduced at, or prior to the Confirmation Hearing: (a) is reasonable, persuasive, credible and accurate as of the dates such analysis or evidence was prepared, presented or proffered; (b) utilizes reasonable and appropriate methodologies and assumptions; (c) has not been controverted by other evidence; and (d) establishes that the Debtors will have sufficient funds, including the receipt of the Insurance Settlement Payment and the BAH Settlement Payment upon the Effective Date of the Plan, to satisfy their obligations under the Plan. As a result, the requirements of section 1129(a)(11) of the Bankruptcy Code have been satisfied.
- V. Payment of Fees—11 U.S.C. § 1129(a)(12). In accordance with 1129(a)(12) of the Bankruptcy Code, section 2.2 of the Plan provides for the payment of all fees payable under 28 U.S.C. § 1930(a). As a result, the requirements of section 1129(a)(12) of the Bankruptcy Code have been satisfied.
- W. <u>Continuation of Retiree Benefits—11 U.S.C. § 1129(a)(13).</u> No retiree benefits existed in the Chapter 11 Cases. As such, the Debtors are not obligated to pay any such benefits and Bankruptcy Code section 1129(a)(13) is inapplicable.
- X. <u>Domestic Support Obligations, Individuals and Certain Transfers—11 U.S.C. §</u>

 1129(a)(14)-(16). The Debtors are not required to pay any domestic support obligations and, therefore, Bankruptcy Code section 1129(a)(14) is satisfied. The Debtors are not individuals and, accordingly, Bankruptcy Code section 1129(a)(15) is inapplicable in these Chapter 11 Cases. The Debtors are moneyed, business or commercial corporations or trusts, as the case may be and, accordingly, Bankruptcy Code section 1129(a)(16) is inapplicable in these Chapter 11 Cases.

- Y. Fair and Equitable; No Unfair Discrimination—11 U.S.C. § 1129(b). Pursuant to Bankruptcy Code section 1129(b), as to any impaired class of unsecured claims or equity interests that rejects a plan, such plan must be "fair and equitable" with respect to each such class. Classes 5(a) and 5(b) Intercompany Claims, and Classes 6(a) and 6(b) Equity Interests will not receive any property under the Plan. No Classes junior to Classes 5(a), 5(b), 6(a) and 6(b) are receiving any recovery and, thus, the "fair and equitable" test has been satisfied. Furthermore, the Plan does not discriminate unfairly against any Class that is deemed to reject the Plan.
- Z. Only One Plan—11 U.S.C. § 1129(c). Other than the Plan (including previous versions thereof), no other plan has been presented for confirmation or ever presented and approved for solicitation in the Chapter 11 Cases. Accordingly, the requirements of Bankruptcy Code section 1129(c) have been satisfied.
- AA. <u>Principal Purpose—11 U.S.C. § 1129(d)</u>. The principal purpose of the Plan is neither the avoidance of taxes nor the avoidance of the requirements of section 5 of the Securities Act of 1933, and no governmental unit has objected to the confirmation of the Plan on any such grounds. The Plan therefore satisfies the requirements of Bankruptcy Code section 1129(d).
- BB. <u>Findings Required as Conditions to Confirmation of the Plan.</u> Section 11.1.8 of the Plan requires that certain findings of fact and conclusions of law be contained within this Confirmation Order. While many of these findings of fact and conclusions of law are addressed elsewhere in this Confirmation Order, out of an abundance of caution, and without diminishing any other provision of this Confirmation Order, the Court makes the following additional findings of fact, conclusions of law, orders and/or decrees based on the record established at the Confirmation Hearing:

- (1) As of the Petition Date, certain of the Debtors have been named as defendants in personal injury or wrongful death actions seeking recovery for damages allegedly caused by Blitz Products;
- (2) The Blitz Personal Injury Trust, as of the Effective Date, will assume the liabilities of the Debtors and the Protected Parties with respect to the Blitz Personal Injury Trust Claims and shall assume all obligations of the Blitz Personal Injury Trust set forth in the Plan Documents;
- (3) The Blitz Personal Injury Trust, upon the Effective Date, shall have received the Blitz Personal Injury Trust Assets or the right to receive such assets in accordance with the terms of the Plan and the Plan Documents, as applicable;
- (4) Pursuit of the Blitz Personal Injury Trust Claims outside the procedures prescribed by the Plan is likely to threaten the Plan's purpose to deal equitably with Blitz Personal Injury Trust Claims;
- (5) The Blitz Personal Injury Trust shall use its assets and income to pay Blitz Personal Injury Trust Claims and Blitz Personal Injury Trust Expenses in accordance with the Plan and the Plan Documents;
- (6) The actual amounts and timing of payment to the Blitz Personal Injury Trust Claims cannot be determined at the time of the entry of this Confirmation Order;
- (7) Unless otherwise provided in the Plan and this Confirmation Order, the sole and exclusive remedy of holders of Blitz Personal Injury Trust Claims against the Debtors and the Protected Parties shall be against the Blitz Personal Injury Trust, and no such Blitz Personal Injury Trust Claims may be asserted against the Debtors or any Protected Party;
- (8) An identity of interests exists between the Debtors and the non-debtor Protected Parties such that a Claim asserted against any non-debtor Protected Party gives rise to a claim against the Debtors by contract and/or operation of the law of indemnity and/or contribution;
- (9) The terms of the Channeling Injunction, including any provisions barring actions against third parties, are set out in conspicuous language in the Plan and in the Disclosure Statement and are essential elements of the Plan and the Debtors' liquidating efforts, and are appropriate under the circumstances;
- (10) The procedures and payment mechanisms set forth in the Blitz Personal Injury Trust Agreement and the Blitz Personal Injury TDP are fair and equitable to the holders of Blitz Personal Injury Trust Claims and provide reasonable assurance that the Blitz Personal Injury Trust will value, and be in a financial position to pay, Blitz Personal Injury Trust Claims that involve similar claims in substantially the same manner;
- (11) The Plan provides a mechanism to address and resolve all, Blitz Personal Injury Trust Claims, as allowed;

- (12) The transfer of Blitz Personal Injury Trust Assets to the Blitz Personal Injury Trust does not violate any obligations of the Debtors;
- (13) Any transfers of property by the Debtors (i) to the Blitz Liquidating Trust and/or the Blitz Personal Injury Trust (a) are or will be legal, valid and effective transfers of property, (b) vest or will vest the Blitz Liquidating Trust and/or the Blitz Personal Injury Trust, as the case may be, with good title to such property free and clear of all Liens, Claims, encumbrances or Equity Interests, except as expressly provided in the Plan or Confirmation Order, (c) do not and will not constitute avoidable transfers under the Bankruptcy Code or under applicable bankruptcy or nonbankruptcy law, and (d) except as provided otherwise in the Plan Documents, do not and will not subject the Debtors, the Blitz Liquidating Trust, the Blitz Personal Injury Trust and/or the BAH Plan Administrator to any liability by reason of such transfer under the Bankruptcy Code or under applicable nonbankruptcy law, including, without limitation, any laws affecting successor, transferee or stamp or recording tax liability, and (ii) to holders of Claims under the Plan are for good consideration and value;
- (14) The Bankruptcy Court has subject-matter jurisdiction to issue the Channeling Injunction; and the Channeling Injunction is authorized by section 105 of the Bankruptcy Code and/or by the Bankruptcy Court's inherent authority or other statutory authority;
- (15) Approval of the Insurance Settlement, the BAH Settlement and any other settlement agreement between the Debtors and any other Entities not previously approved by the Bankruptcy Court is appropriate under Bankruptcy Rule 9019 and the applicable law governing approval of such settlements and compromises and is granted as part of this Confirmation Order;
- (16) Upon satisfaction of the conditions set forth in section 14.2.7 of the Plan, each of the Participating Insurers who have paid its share of the Insurance Settlement Payment shall be released and its Participating Insurer Policy was bought back pursuant to the Insurance Policy Buy-Back Order free and clear of all rights, claims, liens, interest, and/or encumbrances in the Participating Insurer Policy pursuant to section 363 of the Bankruptcy Code.
- (17) Upon satisfaction of the conditions set forth in section 14.2.7 of the Plan, each of the Protected Parties who has paid its share of the Insurance Settlement Payment, or who had its share of the Insurance Settlement Payment made by another Entity on its behalf, shall be deemed dismissed in any action and any complaint, petition, counterclaim, cross-claim, cause of action or other request for relief shall be deemed withdrawn with prejudice without any further action of such Protected Party. For the avoidance of doubt, those parties excepted from the Releases and Channeling Injunction in paragraphs 10(b)(4), 53 and 54 of this Confirmation Order are not released and any pending complaint, petition, counterclaim, cross-claim, cause of action or other request for relief against such non-released party shall survive;
- (18) The Plan complies with all applicable sections of the Bankruptcy Code, and the Debtors have complied with all applicable sections of the Bankruptcy Code;
- (19) The substantive consolidation of the USA Debtors for purposes of Distributions as set forth in section 12.1.5 of the Plan and the substantive consolidation of the

BAH Debtors for purposes of Distributions as set forth in section 12.1.6 of the Plan are approved;

- (20) In light of the respective direct and indirect benefits provided, or to be provided, to the Blitz Personal Injury Trust or the USA Debtors, as applicable, by, or on behalf of, each Participating Insurer, Wal-Mart or any other Protected Party pursuant to the Insurance Settlement Term Sheet, the identification of the Participating Insurers, Wal-Mart or any other Protected Party in the Channeling Injunction is fair and equitable with respect to any Entity that might subsequently assert Blitz Personal Injury Trust Claims or any other Claim against a Participating Insurer, Wal-Mart or any other Protected Party pursuant to the Insurance Settlement Term Sheet. The Insurance Settlement constitutes a reasonable settlement and fair resolution of the Participating Insurers' alleged liabilities and obligations under the Participating Insurer Policies. The Insurance Settlement constitutes a reasonable settlement and fair resolution of Wal-Mart's alleged liabilities and obligations with respect to any Blitz Personal Injury Trust Claim and Wal-Mart's rights under the Blitz Insurance Policies;
- (21) In light of the respective direct and indirect benefits provided, or to be provided, to the Blitz Personal Injury Trust or the USA Debtors, as applicable, by, or on behalf of, each BAH Released Party or any other Protected Party pursuant to the BAH Settlement, the identification of the BAH Released Parties or any other Protected Party in the Channeling Injunction is fair and equitable with respect to any Entity that might subsequently assert Blitz Personal Injury Trust Claims or any other Claim against a BAH Released Party or any other Protected Party pursuant to the BAH Settlement. The BAH Settlement constitutes a reasonable settlement and fair resolution of the BAH Settling Parties' alleged liabilities by the Creditors' Committee on behalf of the USA Debtors' Estates and for the Blitz Personal Injury Claims;
- (22) The contributions to be made by Protected Parties to the Blitz Personal Injury Trust and the USA Debtors' Estates are substantial and are a fundamental, integral and essential component of the success and implementation of the Plan;
- (23) The Channeling Injunction and Releases, as applied to Blitz Personal Injury Trust Claims against the Protected Parties, are essential and necessary for the Debtors because, among other reasons, the Protected Parties would not be willing to make their contributions to the Blitz Personal Injury Trust or the USA Debtors' Estates without the protection provided by the Channeling Injunction and Releases;
- (24) The Plan would not be able to be confirmed and the Insurance Settlement would not be able to be consummated without the BAH Settlement Payment to the USA Debtors' Estates; the BAH Settling Parties would not have agreed to the terms of the BAH Settlement if they did not include the Channeling Injunction and Releases for the BAH Released Parties;
- (25) The identification and designation of each Protected Party is fair and equitable with respect to Entities that might subsequently assert Blitz Personal Injury Trust Claims against any such Protected Party, in light of the benefits provided, or to be provided, to the Blitz Personal Injury Trust by or on behalf of the Protected Parties;

- (26) In view of the substantial contributions to the Blitz Personal Injury Trust and the USA Debtors' Estates made by or on behalf of the Protected Parties, it is reasonable and fair for the Plan to provide that holders of Blitz Personal Injury Trust Claims be enjoined from pursuing any action against the Protected Parties as set forth in the Plan or this Confirmation Order;
- (27) The Plan does not violate any consent to assignment, or similar clause in any Assigned Blitz Insurance Policy;
- (28) The Plan does not materially increase any Non-Participating Insurer's risk of providing coverage for any Blitz Personal Injury Trust Claims under the relevant Assigned Blitz Insurance Policy as compared to the risk that was otherwise being borne by such Non-Participating Insurer prior to the Effective Date;
- (29) Except for an agreed upon joint press statement with respect to the Insurance Settlement to be released to the public at an agreed upon time, no Insurance Settling Party shall make any statement to the media concerning the Insurance Settlement other than referring the media to such press statement and any court filings not under seal. This paragraph shall not preclude plaintiff's counsel from identifying on their respective websites and in other materials describing their respective law firms, the fact that they were one of the counsel involved in the Insurance Settlement; and
- (30) The provisions of this Confirmation Order are non-severable and mutually dependent.
- CC. <u>No Objection to Deemed Rejection of Contracts and Leases</u>. No party to an executory contract or unexpired lease to be rejected by the Debtors pursuant to the Plan has objected to such rejection.
- DD. <u>Plan Modifications</u>. The modifications reflected in this Confirmation Order (collectively, the "<u>Modifications</u>") do not materially or adversely affect or change the treatment of any holder of a Claim or Equity Interest. Accordingly, pursuant to Bankruptcy Rule 3019, such Modifications do not require additional disclosure under Bankruptcy Code section 1125 or resolicitation of acceptances or rejections under Bankruptcy Code section 1126, nor do they require that holders of Claims be afforded an opportunity to change previously cast acceptances or rejections of the Plan.

- EE. <u>Burden of Proof</u>. The Debtors and the Committee, as proponents of the Plan, have met their burden of proving the elements of Bankruptcy Code sections 1129(a) and 1129(b) by a preponderance of the evidence.
- FF. <u>Satisfaction of Confirmation Requirements</u>. Based upon the foregoing, the Plan satisfies the requirements for confirmation set forth in Bankruptcy Code section 1129.
- GG. <u>Retention of Jurisdiction</u>. The Court properly may retain jurisdiction over the matters set forth in Article XIII of the Plan.

IT IS THEREFORE ORDERED, ADJUDGED, AND DECREED that:

- 1. The Insurance Policy Buy-Back Order, attached hereto as **Exhibit A**, is incorporated in full as a part of this Confirmation Order.
- 2. Approval of Modifications. The Modifications are approved. In accordance with Bankruptcy Code section 1127 and Bankruptcy Rule 3019, all holders of Claims who voted to accept the Plan or who are conclusively presumed to have accepted the Plan are deemed to have accepted the Plan as modified by the Modifications. No Holder of a Claim shall be permitted to change its vote as a consequence of the Modifications. The Plan as modified by the Modifications shall constitute the Plan and all references herein to the Plan shall mean the Plan as so modified.
- 3. <u>Confirmation of Plan</u>. All requirements for confirmation of the Plan have been satisfied. The Plan, a copy of was filed on the docket at Docket Number 2007, and each of its provisions (whether or not specifically mentioned herein) and all exhibits and schedules thereto as amended to the date hereof are **CONFIRMED** in each and every respect, pursuant to section 1129 of the Bankruptcy Code.
 - 4. <u>Confirmation Hearing Record</u>. The record of the Confirmation Hearing is closed.

- 5. Formal and Informal Objections to Confirmation. Except as otherwise expressly set forth below in this paragraph, the Objections, as well as any and all other formal and informal objections to Confirmation, are hereby overruled in their entirety. All parties have had a full and fair opportunity to be heard with respect to all issues raised, and all issues that may have been raised, by the objections. Any objection that has been withdrawn is deemed to be withdrawn with prejudice. All objections have been resolved and withdrawn on the terms and subject to the conditions set forth below. The compromises and settlements contemplated by the resolution of such objections described herein are fair, equitable and reasonable, are in the best interests of the Debtors, their respective Estates and Creditors and are expressly approved pursuant to Bankruptcy Rule 9019.
- a. Cataldi Claimants: The Cataldi Claimants' motion to allow a late filed Blitz Personal Injury Claim is granted because of the agreement and compromises of the parties and the submission of the motion prior to commencement of the Confirmation Hearing. The Cataldi Claimants (i) shall be eligible to participate in a distribution under the Blitz Personal Injury TDP as Covered Claimants and qualify to apply to the Special Circumstances Fund and (ii) as to the Non-Appealing Fund, agree to accept the offer amount made by the Blitz Personal Injury Trust upon compliance with the TDP and TDP Scoring System, which Offer Amount is estimated to be \$138,846.
- b. McKenzie Claimants: The McKenzie Claimants' motion to allow a late filed Blitz Personal Injury Claim is granted because of the agreement and compromises of the parties and the submission of the motion prior to commencement of the Confirmation Hearing. The McKenzie Claimants (i) shall be eligible to participate in a distribution under the Blitz Personal Injury TDP as Covered Claimants and (ii) agree to accept the offer amount made by the Blitz Personal Injury Trust upon compliance with the TDP and TDP Scoring System, which Offer Amount is estimated to be \$2,300,000.
- c. Bauman Claimants: The Bauman Claimants' objection to confirmation of the Plan is withdrawn and the Bauman Claimants shall be deemed to vote in favor of the Plan. The Bauman Claimants shall be deemed eligible for the Special Circumstances Fund as if the Bauman Claimants had met the enumerated criteria set forth in paragraph H of the TDP Scoring System for compensation from the Special Circumstances Fund.
- d. Newby Claimants: The Newby Claimants' objection to confirmation of the Plan is withdrawn and the Newby Claimants shall be deemed to vote in favor of the Plan. The Newby Claimants agree to accept a fixed offer amount of \$6.2 Million Dollars in full and final satisfaction of their Blitz Personal Injury Claim.

- e. Bosse Claimant: The Bosse Claimant shall be deemed to vote in favor of the Plan and shall qualify to participate in the \$650,000 allocated to reimburse fees and expenses of objecting claimants pursuant to Section 2.3(g) of the Blitz Personal Injury Trust Agreement.
- f. United States Trustee: The United States Trustee's objection is resolved by amending the Plan's definition of Exculpated Parties as set forth in paragraph 49 of this Confirmation Order.
- g. Anthony Torres, Andrew Torres, Valerie Torres, Colton Warren, Cindy Nichols, Tiffany St. John, Marion "B.J." Jones, Dalan Jones (a minor represented by his parents, Tiffany St. John and Marion Jones), the estate of Leiya Jones (deceased, represented by her parents, Tiffany St. John and Marion Jones), Irene Perez, Aliha Perez (a minor represented by her mother Irene Perez), and the Mims Claimant affirmed on the record their support for confirmation of the Plan as modified by this Confirmation Order.
- 6. <u>Incorporation of Terms and Provisions of Plan.</u> Subject to paragraph 33 of this Confirmation Order, the terms and provisions of the Plan are incorporated by reference into and are an integral part of this Confirmation Order. Each term and provision of the Plan is valid, binding, and enforceable as though fully set forth herein. The provisions of the Plan and this Confirmation Order, including the findings of fact and conclusions of law set forth herein, are non-severable and mutually dependent. The failure specifically to include or reference any particular term or provision of the Plan in this Confirmation Order shall not diminish or impair the effectiveness of such term and provision, it being the intent of the Court that the Plan be confirmed in its entirety.
- 7. Plan Classification Controlling. The terms of the Plan shall solely govern the classification of Claims and Equity Interests for purposes of the distributions to be made thereunder. The classifications set forth on the Ballots tendered to or returned by the holders of Claims and Equity Interests in connection with voting on the Plan: (a) were set forth on the Ballots solely for purposes of voting to accept or reject the Plan; (b) do not necessarily represent, and in no event shall be deemed to modify or otherwise affect, the actual classification of such Claim or Equity Interest under the Plan for distribution purposes; (c) may not be relied upon by

any holder of a Claim or Equity Interest as representing the actual classification of such Claim or Equity Interest under the Plan for distribution purposes; and (d) shall not bind the Debtors, the Blitz Liquidating Trust or the Blitz Personal Injury Trust.

- 8. Binding Effect. Effective on the Effective Date or any other date if so provided in the Plan, and except as expressly provided otherwise in this Confirmation Order, the Plan, and its provisions shall be binding to the fullest extent of the law upon the Debtors, any party in interest, any entity acquiring or receiving property or a distribution under the Plan and any holder of a Claim against or Equity Interest in the Debtors, including all governmental entities, whether or not the Claim or Equity Interest of such holder is impaired under the Plan and whether or not such holder or entity has accepted the Plan. The Plan Documents constitute the legal, valid, binding, enforceable, and authorized obligations of the respective parties thereto and shall be enforceable in accordance with their terms. Pursuant to Bankruptcy Code sections 1123(a) and 1142(a) and the provisions of this Confirmation Order, the Plan, the Plan Documents, and all other Plan-related documents shall apply and be enforceable notwithstanding any otherwise applicable non-bankruptcy law.
- 9. <u>Propriety of Releases and Channeling Injunction</u>. Based on the findings of fact set forth in this Confirmation Order, including paragraph H, and the record established at the Confirmation Hearing, the Releases and Channeling Injunction with respect to each Protected Party is appropriate under applicable law:
 - a. Substantial contributions by or on behalf of each of the Protected Parties are being made to fund the distributions provided under the Plan.
 - b. The Plan has been overwhelmingly accepted by the classes of creditors affected by the Releases and the Channeling Injunction.
 - c. The holders of Blitz Personal Injury Claims are receiving fair consideration in exchange for the Releases and the Channeling Injunction.

- d. The Plan provides holders of Blitz Personal Injury Claims with the opportunity to reject the settlement offers they receive from the Blitz Personal Injury Trust and resort to the tort system to liquidate their claims.
- e. The Releases and the Channeling Injunction are necessary and critical provisions of the Plan, without which the Plan cannot be confirmed.
- f. The record of the Confirmation Hearing and the Chapter 11 Cases is sufficient to support the Releases and the Channeling Injunction.
- g. The Plan is the product of the global settlement that was reached in the extraordinary circumstances of these Chapter 11 Cases. These extraordinary circumstances support the approval of the Releases and the Channeling Injunction.
- h. There is a substantial identity of interest between the Debtors and each of the Protected Parties, such that a Blitz Personal Injury Claim asserted against a Protected Party is essentially a claim against the Debtors.
- i. This Court has subject matter jurisdiction over the third party claims against the Protected Parties because of the unity of interest between the Protected Parties and the Debtors.
- Indemnification Obligations. Each of the release, exculpation, injunction and indemnification provisions set forth in the Plan, including without limitation the provisions in Sections 3.8.2, 4.3.2.5, 4.3.3, 4.3.4, 4.14, 7.2, 7.3, 11.1.8, 14.2.5, and 15.4 of the Plan, is hereby approved in all respects in its entirety, is so ordered and shall be immediately effective on the Effective Date of the Plan without further action or notice by this Court, any of the parties to such releases, exculpation, injunction and indemnifications or any other party. For the avoidance of doubt, upon satisfaction of the conditions set forth in section 14.2.7 of the Plan, (a) Westchester shall be released from all Blitz Personal Injury Trust Claims occurring on or after the July 31, 2005 and on or before July 31, 2012 with the exception of the Calder and Bosse claims (which shall be treated in accordance with paragraphs 28 and 29 of the Insurance Settlement Term Sheet, as applicable); and (b) each of the Participating Insurers who have paid its share of the Insurance

Settlement Payment shall be released from any and all liability under the Participating Insurer Policies, except that Westchester shall not be released from the *Calder* and *Bosse* claims. Upon the resolution of the *Calder* and *Bosse* claims with finality either by compromise or adjudication, Westchester Fire Insurance Company or Westchester Surplus Lines Insurance Company, as the case may be, shall receive the remainder of the release and protections for Policies bearing Nos. CUW788371001 and G22053504001 that are afforded to the Participating Insurer Policies and Policies Nos. CUW788371001 and G22053504001 shall be exhausted. Also for the avoidance of doubt, the sole and exclusive remedy of holders of Blitz Personal Injury Trust Claims against the Debtors and the Protected Parties shall be against the Blitz Personal Injury Trust, and no such Blitz Personal Injury Trust Claims may be asserted against the Debtors or any Protected Party. As required by Section 4.3.3 of the Plan, and without diminishing any other provision of this Confirmation Order, the Court orders and decrees that:

Imposition of Channeling Injunction: In order to preserve and promote the settlements contemplated by and provided for in the Plan and to supplement, where necessary, the injunctive effect of the Channeling Injunction and the Releases described in section 7.2 of the Plan, and pursuant to the exercise of the equitable jurisdiction and power of the Bankruptcy Court under section 105(a) of the Bankruptcy Code, all Entities that have held or asserted, or that hold or assert any Blitz Personal Injury Trust Claim against the Protected Parties, or any of them, shall be permanently stayed, restrained and enjoined from taking any action for the purpose of directly or indirectly collecting, recovering, or receiving payments, satisfaction, or recovery from any such Protected Party with respect to any such Blitz Personal Injury Trust Claim, including, but not limited to:

(1) commencing, conducting or continuing, in any manner, whether directly or indirectly, any suit, action or other proceeding of any kind with

respect to any such Blitz Personal Injury Trust Claim, against any of the Protected Parties, or against the property of any Protected Party with respect to any such Blitz Personal Injury Trust Claim;

- (2) enforcing, levying, attaching, collecting or otherwise recovering, by any manner or means, or in any manner, either directly or indirectly, any judgment, award, decree or other order against any of the Protected Parties or against the property of any Protected Party with respect to any such Blitz Personal Injury Trust Claim;
- (3) creating, perfecting or enforcing in any manner, whether directly or indirectly, any Lien of any kind against any Protected Party or the property of any Protected Party with respect to any such Blitz Personal Injury Trust Claim;
- (4) asserting, or accomplishing any setoff, right of subrogation, indemnity, contribution or recoupment of any kind, whether directly or indirectly, against any obligation due any Protected Party or against the property of any Protected Party with respect to any such Blitz Personal Injury Trust Claim; and
- (5) taking any act, in any manner, in any place whatsoever, that does not conform to, or comply with, the provisions of the Plan Documents, with respect to such Blitz Personal Injury Trust Claim.
- (b) <u>Reservations</u>: Notwithstanding anything to the contrary in paragraph 10(a) above or Section 4.3.3 of the Plan, the Channeling Injunction shall not enjoin:
 - (1) any claim for damages on account of bodily injury and/or property damage that occurred on or after 12:01 A.M. CST on July 31, 2012;
 - (2) the rights of Entities to the treatment afforded them under the Plan, including the right of Entities holding Blitz Personal Injury Trust Claims to assert such Claims in accordance with the Blitz Personal Injury Trust Agreement and the Blitz Personal Injury TDP solely against the Blitz Personal Injury Trust whether or not there are funds to pay such Blitz Personal Injury Trust Claims;
 - (3) the rights of Entities to assert any Claim, debt, litigation, or liability for payment of Blitz Personal Injury Trust Expenses solely against the Blitz Personal Injury Trust whether or not there are funds to pay such Blitz Personal Injury Trust Expenses;
 - (4) the rights of holders of Blitz Personal Injury Trust Claims that arose prior to the Release Date to pursue and/or prosecute any Insurance Actions, including, but not limited to, a Direct Action Claim, against any Non-Participating Insurer;
 - (5) if and to the extent necessary to preserve rights against any Non-Participating Insurer, the rights of the Blitz Personal Injury Trust to prosecute any

Direct Action Claim, as attorney in fact for a holder of Blitz Personal Injury Trust Claim, including the participation of such claimant in such Direct Action Claim;

- (6) the rights of any Entity to assert any claim, debt, obligation or liability for payment against a Non-Participating Insurer;
- (7) the rights of any Entity (other than the Debtors and Additional Insureds, in their capacity as such) to assert against a Protected Party any claim, debt, or obligation for payment that is not in any way based upon, related to, or arising out of, any Blitz Insurance Policy, Blitz Product or otherwise subject to the Insurance Settlement and BAH Settlement;
- (8) the Blitz Personal Injury Trust from enforcing its rights under the Insurance Settlement Term Sheet, the Blitz Personal Injury Trust Agreement and the Blitz Personal Injury TDP; and
- (9) the rights of any Indemnified Party to demand the Blitz Personal Injury Trust to fulfill its obligations to enforce the terms of the Channeling Injunction consistent with section 4.14 of the Plan, the Insurance Settlement Term Sheet and the BAH Settlement Term Sheet.
- (c) <u>Modifications</u>: There can be no modification, dissolution or termination of the Channeling Injunction, which shall be a permanent injunction.
- (d) <u>Non-Limitation Channeling Injunction</u>: Nothing in this

 Confirmation Order, the Plan or the Plan Documents shall be construed in any way to limit the scope, enforceability, or effectiveness of the Channeling Injunction issued in connection with the Plan or the Blitz Personal Injury Trust's assumption of all liability with respect to Blitz Personal Injury Trust Claims.
- (e) <u>Bankruptcy Rule 3016 Compliance</u>: The Proponents' compliance with the requirements of Bankruptcy Rule 3016 shall not constitute an admission that the Plan provides for an injunction against conduct not otherwise enjoined under the Bankruptcy Code.
- (f) <u>Releases of Liabilities of Blitz Personal Injury Claims</u>: Except as provided in the Plan, the transfer to, vesting in, and assumption by the Blitz Personal Injury Trust of the Blitz Personal Injury Trust Assets as contemplated by the Plan shall, as of the Effective

Date, release all obligations and liabilities of and bar recovery or any action against the Protected Parties and their respective estates, Affiliates and subsidiaries, for or in respect of all Blitz Personal Injury Claims. The Blitz Personal Injury Trust shall, as of the Effective Date, assume sole and exclusive responsibility and liability for all Blitz Personal Injury Trust Claims, and such Claims shall be liquidated, resolved and, where appropriate, paid by the Blitz Personal Injury Trust from the Blitz Personal Injury Trust Assets or as otherwise directed in the Blitz Personal Injury Trust Documents. For the avoidance of doubt, holders of Vendor Claims, holders of a Co-Defendant Claims, and Direct Action Claims shall not receive distributions or other payment of any funds attributable to the Insurance Settlement Payment or the BAH Settlement Payment on behalf of, related to or with respect to such Claims. As set forth in section 4.14 of the Plan and the Blitz Personal Injury Trust Agreement, the Blitz Personal Injury Trust shall defend and indemnify the Indemnified Parties in connection with any proceeding involving, relating to or arising out of, in whole or in part, the enforcement or enforceability of the Channeling Injunction.

11. Enjoinment of Judicial Actions. Effective on the Effective Date, without any further action of the Court or other entity, any suit, legal action, or other proceeding (including a judicial, arbitration, administrative or other proceeding) in any forum in the United States against or affecting any Protected Party, shall be stayed, enjoined or otherwise prohibited from continuation as to any Protected Party except for those parties excepted from the channeling injunction provided in paragraph 10(b) of this Confirmation Order. Without limiting the foregoing, on or after the Effective Date, without any further action of the Court or any entity, the Blitz Personal Injury Trustee or any Protected Party, as applicable, is authorized and empowered to file or record this Confirmation Order with respect to any suit, legal action, or

other proceeding (including a judicial, arbitration, administrative or other proceeding) in any forum in the United States, and such Confirmation Order shall be binding upon all entities and governmental authorities, including, all filing agents, filing officers, administrative agencies or units, governmental departments or units, secretaries of state, federal, state and local officials, with respect to the stay or enjoinment of each such action. Each and every governmental authority, and any other person or entity, is hereby authorized and directed to accept this Confirmation Order as independent and conclusive authorization to effectuate a stay or enjoinment of any suit, legal action, or other proceeding described herein.

12. Plan Implementation Authorization. The Debtors, the Blitz Personal Injury Trust and the Blitz Liquidating Trust, as the case may be, and their respective directors, officers, members, agents, and attorneys, are authorized and empowered from and after the date hereof to negotiate, execute, issue, deliver, implement, file, or record any contract, instrument, release, or other agreement or document, including, without limitation, the Plan Documents, as the same may be modified, amended and supplemented (including such modifications to the Blitz Personal Injury Trust and/or the Blitz Liquidating Trust that are substantially consistent with the terms and provisions of such form and necessary to satisfy the conditions to the effectiveness of the Plan), and to take any action necessary or appropriate to implement, effectuate, consummate, or further evidence the Plan in accordance with its terms, or take any or all corporate actions authorized to be taken pursuant to the Plan, whether or not specifically referred to in the Plan or any exhibit thereto, without further order of the Court. To the extent applicable, any or all such documents shall be accepted upon presentment by each of the respective state filing offices and recorded in accordance with applicable state law and shall become effective in accordance with their terms and the provisions of state law. Pursuant to section 303 of the General Corporation Law of the State of Delaware and any comparable provision of the business corporation laws of any other state, as applicable, no action of the Debtors' Boards of Directors or the trustee of the Blitz Personal Injury Trust and/or the Blitz Liquidating Trust will be required to authorize the Debtors to enter into, execute and deliver, adopt or amend, as the case may be, the Plan Documents, and following the Effective Date, each of the Plan Documents will be a legal, valid and binding obligation of the Debtors, enforceable against the Debtors in accordance with the respective terms thereof.

- 13. <u>Debtors' Status Prior to Effective Date</u>. The Debtors shall remain Debtors-in-Possession under the Bankruptcy Code until the Effective Date. The Blitz Liquidating Trustee, the Blitz Personal Injury Trustee and/or the BAH Plan Administrator shall wind-up the affairs of the Debtors and may make distributions to creditors after the Effective Date in accordance with the Plan, the Blitz Liquidating Trust, Blitz Personal Injury Trust, and the Blitz Personal Injury TDP.
- 14. <u>Dissolution of Securities</u>. In accordance with the terms of the Plan, as soon as practicable after the Effective Date, the Equity Interests in the Debtors shall be dissolved for all purposes without the necessity for any other or further actions to be taken by or on behalf of the Debtors or payments to be made in connection therewith.
- 15. <u>Continued Corporate Existence</u>. As of the Effective Date, the USA Debtors, other than Blitz U.S.A., Inc., shall be deemed to have been dissolved and terminated in accordance with Section 12.4.1 of the Plan. Upon the Effective Date, or as soon as reasonably practicable thereafter, (i) LAM shall merge with and into BAH; (ii) the number of directors constituting the entire board of directors of BAH shall be fixed at one; (iii) the BAH Plan Administrator shall be deemed to be elected as the sole officer and sole director of BAH and each existing officer and

member of the board of directors of the BAH Debtors shall be deemed to have been removed as of the occurrence of the Effective Date; (iv) all existing stock in BAH shall be cancelled; (v) a single share of the stock of BAH shall be issued to Kinderhook Capital Fund II, L.P. (which single share shall be issued for the sole purpose of allowing BAH to comply with any annual meeting or election of director requirements after the Effective Date, but shall not entitle the Kinderhook Capital Fund II, L.P. to receive, make, or call for any distribution, dividends or redemptions from BAH); (vi) all actions in furtherance of the Plan shall be deemed authorized, approved, and, to the extent taken prior to the Effective Date, ratified without any requirement for further action by the BAH Debtors, holders of Claims or Equity Interests, directors, members, managers, or officers of the BAH Debtors, or any other Entity, including the dissolution or winding up of BAH; and (vii) the BAH Plan Administrator shall be authorized to execute, deliver, file, or record such contracts, instruments, release, and other agreements or documents and take such other actions as the BAH Plan Administrator may deem, in the BAH Plan Administrator's sole discretion, necessary or appropriate to effectuate and implement the provisions of the Plan. As soon as reasonably practicable after making distributions provided for under the Plan, BAH shall be dissolved and the BAH Plan Administrator, without further action of the directors or stockholders of BAH, shall be authorized to file a certificate of dissolution and take any other action that may be necessary to terminate the corporate existence of BAH.

16. <u>Transfer of Records</u>. Notwithstanding anything to the contrary in the Plan or the Plan Documents, the Blitz Liquidating Trustee shall be and is hereby appointed as custodian of all books and records in the possession and control of the USA Debtors, which shall include but not be limited to, (i) all electronic documents and data, (ii) hard drives, (iii) software, (iv) cloud storage, (v) archives, (vi) hard copies, (vii) molds, gas cans and prototypes (the "<u>Books and</u>

Records") and is authorized to testify (i) that the Books and Records or other tangible property in his possession were once kept in the ordinary course of business in the files and/or possession of the USA Debtors, and (ii) for purposes of establishment of business records foundation under state or federal evidence laws, all other objections to the admissibility of such records shall be preserved. For the avoidance of doubt, the provisions of sections 4.10 and 5.12 of the Plan apply to the Books and Records unless otherwise specifically provided herein.

- (a) All Books and Records shall be transferred to the Blitz Liquidating Trustee. The Blitz Liquidating Trustee shall preserve all Books and Records for a period of not less than six (6) months from the Effective Date (the "Retention Period"). At the end of the Retention Period, the Blitz Liquidating Trustee may file a motion to abandon or destroy some or all of the Books and Records.
- (b) Wal-Mart, the Blitz Insurers and any Vendors or holders of a Co-Defendant Claims against whom Blitz Personal Injury Claims are asserted prior to the Release Date (the "Review Parties") shall have four (4) months from the commencement of the Retention Period to review the Books and Records for privileged documents and communications, with all such privileges or protections being preserved and not waived pursuant to Federal Rule of Evidence 502(d), with such review to be performed where the Books and Records are located, unless otherwise agreed to by the Blitz Liquidating Trustee. There shall be a presumption of privilege for (i) all Books and Records as to which the Debtors previously asserted a privilege or similar protection, (ii) all documents authored by counsel, and (iii) all communications between any of the USA Debtors and

their counsel. Retrieval of electronically stored information shall be performed by a nationally recognized vendor with all costs to born among the Review Parties.

- (c) All Books and Records as to which the Debtors previously asserted a privilege or similar protection or for which a privilege is asserted by a Review Party shall be transferred to the possession of a neutral third party (the "Privileged Records Custodian"), agreed to by the Review Parties, the Blitz Liquidating Trustee and the Blitz Personal Injury Trustee, with the Review Parties asserting privileges responsible for the fees and costs of the Privileged Records Custodian.
- (d) The Privileged Records Custodian, at the conclusion of the four month review period shall provide a log of all documents sufficient to analyze whether a privilege exists (the "Privilege Log"). The Privilege Log will be provided to the Blitz Liquidating Trustee and the Blitz Personal Injury Trustee.
- (e) Any party seeking discovery in litigation relating to the USA Debtors may challenge the assertion of any privilege in the court where the litigation is pending. The Privileged Records Custodian shall give notice to the Review Party asserting the privilege so that they may defend the assertion of privilege in the court where the litigation is pending.
- (f) The Review Parties shall be responsible for reimbursing the Blitz Liquidating Trustee for any fees and costs incurred in the exercise of the Blitz Liquidating Trustee's duties under the provisions of this paragraph to the extent such fees and costs exceed \$100,000. Prior to any reimbursement under the

foregoing sentence, Wal-Mart shall be entitled to audit the books and records of the Blitz Liquidating Trustee.

- 17. The Blitz Personal Injury Trust. The Blitz Personal Injury Trust shall be established on the Effective Date and shall be maintained thereafter in accordance with the terms of the Plan and the Blitz Personal Injury Trust Agreement. The Blitz Personal Injury Trust Agreement, in the final form annexed hereto as Exhibit B, and all of the provisions therein, are hereby approved by this Confirmation Order. The designation of Hon. Richard Cohen as the initial Blitz Personal Injury Trustee and Hank Anderson, Diane Breneman, Richard Denney, and Daniel Haltiwanger as initial members of the Blitz Personal Injury TAC is approved. Pending the occurrence of the Effective Date, the Proponents are authorized to take all actions as may be necessary to facilitate the creation and implementation of the Blitz Personal Injury Trust. On the Effective Date, pursuant to Section 4.3 of the Plan, a newly-formed trust which qualifies as a "Qualified Settlement Fund" within the meaning of section 468B of the Internal Revenue Code shall be created to hold the Blitz Personal Injury Trust Assets, make certain distributions pursuant to Article IV of the Plan and liquidate or otherwise administer the Blitz Personal Injury Trust Assets. The Blitz Personal Injury Trust is authorized and empowered, pursuant to the Plan, including, without limitation, Article IV of the Plan, and the Blitz Personal Injury Trust Agreement, to liquidate or otherwise administer the Blitz Personal Injury Trust Assets.
- 18. <u>Transfer of Assets to Blitz Personal Injury Trust</u>. Pursuant to Section 4.5.1 of the Plan, and upon the Effective Date, the Debtors shall be deemed to have transferred all Blitz Personal Injury Trust Assets, other than the Insurance Settlement Payment, the Supplemental Insurance Settlement Payment and \$1.54 million (which shall be remitted by the Blitz

Liquidating Trust) to the Blitz Personal Injury Trust subject to all rights, defenses and setoffs in accordance with the Plan.

- 19. The Blitz Liquidating Trust. The Blitz Liquidating Trust shall be established on the Effective Date and shall be maintained thereafter in accordance with the terms of the Plan and the Blitz Liquidating Trust Agreement. The Blitz Liquidating Trust Agreement, in the final form annexed hereto as **Exhibit C**, and all of the provisions therein, are hereby approved by this Confirmation Order. The designation of Steven D. Sass as the initial Blitz Liquidating Trustee is approved. Pending the occurrence of the Effective Date, the Debtors are authorized to take all actions as may be necessary to facilitate the creation and implementation of the Blitz Liquidating Trust. On the Effective Date, pursuant to Section 5.6.1 of the Plan, a newly-formed trust which qualifies as a "grantor trust" for federal income tax purposes shall be created to hold the Blitz Liquidating Trust Assets, make certain distributions pursuant to Article V of the Plan and liquidate or otherwise administer the Blitz Liquidating Trust Assets. The Blitz Liquidating Trust is authorized and empowered, pursuant to the Plan, including, without limitation, Article V of the Plan, and the Blitz Liquidating Trust Agreement, to liquidate or otherwise administer the Blitz Liquidating Trust Assets.
- 20. <u>Transfer of Assets to the Blitz Liquidating Trust</u>. Pursuant to Section 5.5 of the Plan, and upon the Effective Date, (i) the Debtors shall be deemed to have transferred the Blitz Liquidating Trust Assets to the Blitz Liquidating Trust, and (ii) Wal-Mart shall waive its secured setoff claim against Blitz U.S.A., Inc., and will pay to the Blitz Liquidating Trust the sum of \$1.54 million.
- 21. <u>Special Provisions from the BAH Settlement</u>. On the Effective Date, any proofs of claim filed by any members of the Committee and any holder of a Blitz Personal Injury Trust

Claim against the BAH Debtors shall be deemed withdrawn with prejudice as against the BAH Debtors and all Blitz Personal Injury Trust Claims against the BAH Debtors are released and shall be channeled to the Blitz Personal Injury Trust and may thereafter be asserted exclusively against the Blitz Personal Injury Trust and resolved in accordance with the terms, provisions and procedures of the Blitz Personal Injury Trust Agreement and the Blitz Personal Injury TDP. Furthermore, the Committee and/or the Blitz Personal Injury Trustee shall support any objection by the BAH Debtors to Claims by any Blitz Insurer against the BAH Debtors. For avoidance of doubt, and notwithstanding anything that may be construed to the contrary herein, in the Plan, the Plan Documents or any other document, the Releases of the BAH Released Parties are, and shall be construed to be, as broad as the broadest Releases provided to any other Entity.

- 22. <u>Proofs of Claim of the Participating Insurers</u>. Upon the Effective Date, the Participating Insurers shall be deemed to withdraw their proofs of claim filed against the Debtors.
- 23. <u>Ratification</u>. All actions taken by the Debtors from the Petition Date through the Confirmation Date are hereby ratified. All actions taken by the Debtors from the Confirmation Date through the Effective Date shall be deemed automatically ratified on the occurrence of the Effective Date.
- 24. <u>Post-Effective Date Fees and Expenses</u>. From and after the Effective Date, the Blitz Personal Injury Trustee and the Blitz Liquidating Trustee shall, in accordance with the Blitz Personal Injury Trust Agreement and the Blitz Liquidating Trust Agreement, respectively; but without the necessity for any approval by the Court, pay the reasonable fees and expenses of the professional Persons thereafter incurred by each of the respective trustees related to the implementation and consummation of the Plan (<u>i.e.</u>, fees and expenses that are not

Administrative Expense Claims). From and after the Effective Date, BAH shall, without the necessity for any approval by the Court, pay the reasonable fees and expenses of any professionals retained by BAH and related to the implementation and consummation of the Plan. For the avoidance of doubt, this Paragraph is not intended to limit or preclude the payment of any post effective date fees or expenses incurred at the request of the Blitz Personal Injury Trust or the Blitz Liquidating Trust or defending fees and expenses incurred in defending or pursuing fee applications relating to fees and expenses incurred prior to the Effective Date.

- 25. Approval of Compromise and Settlement. The settlements, compromises, releases and injunctions set forth in Article XIV of the Plan, including those described in the Insurance Settlement and the BAH Settlement, are approved and incorporated into this Confirmation Order in all respects.
- Approval of Deemed Consolidation. Pursuant to Sections 12.5.1 and 12.6.1 of the Plan, the terms of the separate substantive consolidation of the USA Debtors and the BAH Debtors are approved. On the Effective Date, without further action of the Debtors, the cases of the following Debtors shall be closed: (i) LAM 2011 Holdings, LLC (Case No. 11-13605); (ii) Blitz Acquisition, LLC (Case No. 11-13606); (iii) Blitz RE Holdings, LLC (Case No. 11-13607); and (iv) MiamiOK, LLC f/k/a F3 Brands LLC (Case No. 11-13604). In the event the Clerk of the Court requests an additional order or orders regarding the closing of these cases, the Debtors may submit such order or orders under certification of counsel.
- 27. <u>Governmental Approvals Not Required</u>. This Confirmation Order shall constitute all approvals and consents required, if any, by the laws, rules or regulations of any State or any other governmental authority with respect to the implementation or consummation of the Plan, the Plan Documents, and any other documents, instruments, or agreements, and any amendments

or modifications thereto, and any other acts referred to in or contemplated by the Plan, the Disclosure Statement, the Plan Documents, and any other documents, instruments, or agreements, and any amendments or modifications thereto.

- Exemption from Certain Taxes. Pursuant to Bankruptcy Code section 1146(a), the issuance, transfer or exchange of any security or the making or delivery of any instrument of transfer under the Plan may not be taxed under any law imposing a stamp tax, use tax, sales tax or similar tax. Bankruptcy Code section 346 shall apply to any taxes that may potentially result from, or may be related to, the events, transactions and occurrences of the Chapter 11 Cases and, in particular, pursuant to Bankruptcy Code section 346, no state or local tax imposed on, or measured by, income shall be imposed on the Debtors.
- 29. <u>Applicable Non-Bankruptcy Law.</u> Pursuant to Bankruptcy Code sections 1123(a) and 1142(a), the provisions of this Confirmation Order, the Plan, or any other amendments or modifications thereto shall apply and be enforceable notwithstanding any otherwise applicable non-bankruptcy law.
- 30. Approval of Deemed Rejection of Remaining Contracts and Leases. Unless otherwise provided in an order of or in proceedings before the Court specifically dealing with an executory contract or unexpired lease that is subject to rejection pursuant to Section 8.1 of the Plan, the rejection of such contract or lease is hereby approved as of the Effective Date as proposed in the Plan. If the rejection pursuant to Plan Section 8.1 results in a Claim for damages, then such Claim shall be forever barred and shall not be enforceable against the Estates, the Blitz Liquidating Trust, their successors or properties, unless a Proof of Claim is filed and served on the Blitz Liquidating Trust within thirty (30) days after the Effective Date.

- 31. Governing Law. Except to the extent that the Bankruptcy Code or Bankruptcy Rules or other federal laws are applicable, and subject to the provisions of any contract, instrument, release, or other agreement or document entered into in connection with the Plan, the construction, implementation, and enforcement of the Plan and all rights and obligations arising under the Plan shall be governed by, and construed and enforced in accordance with, the laws of the State of Delaware, without giving effect to conflicts of law principles which would apply the law of a jurisdiction other than the State of Delaware.
- Administrative Expense Claims Bar Date. Unless required to be filed by an 32. earlier date by another order of this Court, all (i) holders of Administrative Expense Claims that have not been Allowed by Final Order of the Bankruptcy Court and (ii) Bankruptcy Professionals requesting compensation or reimbursement of expenses pursuant to sections 327, 328, 330, 331, 503(b) or 1103 of the Bankruptcy Code for services rendered during the Chapter 11 Cases (including, without limitation, any compensation requested by any Bankruptcy Professional or any other Entity for making a substantial contribution to the Chapter 11 Cases), shall file with the Bankruptcy Court, as applicable, a request for Allowance of their Administrative Expense Claim (which request shall specify whether the Claim is asserted against the USA Debtors or the BAH Debtors), or, in the case of Bankruptcy Professionals, an application for final allowance of compensation and reimbursement of expenses on or before the date that is forty-five (45) days after the Effective Date. Any request for payment of a Administrative Expense Claim that is not timely filed as set forth above will be forever barred, and holders of such Claims will not be able to assert such Claims in any manner against the Debtors, the Estates, the Blitz Liquidating Trust or the Blitz Liquidating Trustee.

- 33. <u>Effect of Conflict Between Plan and Confirmation Order.</u> This Confirmation Order shall control over the terms of the Plan and the Plan Documents in the event of any modification, clarification or supplement to the Plan and/or in the event of any inconsistency or ambiguity between the Plan and this Confirmation Order.
- 34. Reversal. If any or all of the provisions of this Confirmation Order are hereafter reversed, modified, or vacated by subsequent order of the Court or any other court, in the absence of a stay of this Confirmation Order, such reversal, modification, or vacatur shall not affect the validity of the acts or obligations incurred or undertaken in good faith under or in connection with the Plan prior to the Debtors' receipt of written notice of entry of any such order. Notwithstanding any such reversal, modification, or vacatur of this Confirmation Order, in the absence of a stay of this Confirmation Order, any such act or obligation incurred or undertaken in good faith pursuant to, and in reliance on, this Confirmation Order prior to the effective date of such reversal, modification, or vacatur shall be governed in all respects by the provisions of this Confirmation Order and the Plan or any amendments or modifications thereto.
- 35. Retention of Jurisdiction. Pursuant to section 105(a) and 1142 of the Bankruptcy Code, the Court hereby retains jurisdiction of the Chapter 11 Cases and all matters arising under, arising out of, or related to, the Chapter 11 Cases and the Plan (i) as provided for in the Plan, (ii) as provided for in this Confirmation Order, and (iii) for the purposes set forth in sections 1127 and 1142 of the Bankruptcy Code.
- 36. <u>Continuation of the Automatic Stay.</u> Unless otherwise provided in this Confirmation Order or a separate order from the Court, until the Effective Date, all injunctions or stays provided for in the Chapter 11 Cases under Bankruptcy Code sections 105(a) or 362, or otherwise, and in existence on the Confirmation Date, shall remain in full force and effect. After

the Effective Date, all injunctions or stays provided for in the Chapter 11 Cases under Bankruptcy Code sections 105(a) or 362, or otherwise, and in existence on the Confirmation Date, shall remain in full force and effect to the extent provided in section 7.1 of the Plan and in this Confirmation Order.

- 37. Payment of Statutory Fees. With respect to the period prior to the Effective Date, all statutory fees pursuant to 28 U.S.C. § 1930(a)(6) shall be paid by the Debtors on the Effective Date or other required payment date. From and after the Effective Date, the Blitz Liquidating Trust shall be liable for and shall pay the fees assessed against the USA Debtors' estate under 28 U.S.C. § 1930 until entry of a final decree closing the USA Debtors' Chapter 11 Case and the Blitz Liquidating Trust shall file post-confirmation quarterly reports in conformity with the U.S. Trustee guidelines until entry of an order closing or converting the USA Debtors' Chapter 11 Cases. From and after the Effective Date, BAH shall be liable for and shall pay the fees assessed against the BAH Debtors' Estates under 28 U.S.C. § 1930 and BAH and/or the BAH Plan Administrator shall file post-confirmation quarterly reports in conformity with the U.S. Trustee guidelines until entry of an order closing or converting the BAH Debtors' Chapter 11 Cases.
- 38. <u>Provisions Governing Distributions</u>. The Blitz Personal Injury Trustee, the Blitz Liquidating Trustee and/or the BAH Plan Administrator, as applicable, shall make all distributions required under the Plan and the distribution provisions of Articles IV and V of the Plan, the Blitz Liquidating Trust Agreement, the Blitz Personal Injury Trust Agreement, and the Blitz Personal Injury TDP, shall be, and hereby are, approved in their entirety.
- 39. <u>Procedures for Resolution of Claims</u>. The claims resolution procedures and reserves to be established in accordance with Articles IV and V of the Plan, the Blitz Liquidating

Trust Agreement, the Blitz Personal Injury Trust Agreement, and the Blitz Personal Injury TDP, shall be, and hereby are, approved in their entirety.

- 40. <u>Setoff Rights</u>. In the event that the Debtors have a Claim of any nature whatsoever against the holder of a Claim against the Debtors, then the Debtors, the Blitz Liquidating Trustee and/or the BAH Plan Administrator, as the case may be, are authorized, but not required to, set off against the Claim (and any payments or other Plan distributions to be made in respect of such Claim under the Plan) the respective Debtor's Claim against such holder, subject to the provisions of Bankruptcy Code sections 553, 556 and 560. Neither the failure to set off nor the allowance of any Claim under the Plan shall constitute a waiver or release of any Claims that the Debtors may have against the holder of any Claim. Such setoff rights of the USA Debtors shall inure to the benefit of the Blitz Liquidating Trust and any successors thereto.
- 41. <u>Dissolution of the Committee</u>. On the Effective Date, the Committee shall be dissolved in accordance with Section 15.3 of the Plan.
- 42. Final Order; Authorization to Consummate Plan. This Confirmation Order is a Final Order and the period in which an appeal must be filed shall commence upon the entry hereof. Notwithstanding Bankruptcy Rule 3020(e), this Confirmation Order shall take effect immediately upon its entry and the Debtors are authorized to consummate the Plan immediately after entry of this Confirmation Order and the satisfaction or waiver of all other conditions to the Effective Date of the Plan, in accordance with the terms of the Plan. The Proponents, the Blitz Personal Injury Trust and the Blitz Liquidating Trust, as applicable, are authorized to consummate the Plan after entry of this Confirmation Order subject to satisfaction of the conditions precedent to the occurrence of the Effective Date set forth in Section 11.2 of the Plan, or waiver of such conditions pursuant to Section 11.5 of the Plan.

- 43. <u>Substantial Consummation.</u> The substantial consummation of the Plan, within the meaning of Bankruptcy Code section 1127, is deemed to occur on the first date, on or after the Effective Date, on which any distributions are made in accordance with the terms of the Plan to holders of any Allowed Claims.
- 44. <u>Post-Confirmation Modification of the Plan</u>. Section 10.1 of the Plan shall govern any post-confirmation Plan modifications.
- 45. Filing and Recording. This Confirmation Order (a) is and shall be effective as a determination that, on the Effective Date, all Claims and Equity Interests existing prior to such date have been unconditionally released, discharged and terminated, and (b) is and shall be binding upon and shall govern the acts of all entities including, without limitation, all filing agents, filing officers, title agents, title companies, recorders of mortgages, recorders of deeds, registrars of deeds, administrative agencies, governmental departments, secretaries of state, federal, state and local officials and all other Persons and entities who may be required, by operation of law, the duties of their office, or contract, to accept, file, register or otherwise record or release any document or instruments. Each and every federal, state and local government agency is hereby directed to accept any and all documents and instruments necessary, useful or appropriate (including Uniform Commercial Code financing statements) to effectuate, implement and consummate the transactions contemplated by the Plan and this Confirmation Order without payment of any recording tax, stamp tax, transfer tax or similar tax imposed by state or local law.
- 46. <u>Notice of Entry of Confirmation Order.</u> No later than [five] business days following the date of entry of this Confirmation Order, the Creditors' Committee or the Blitz Liquidating Trustee shall serve notice of the entry of this Confirmation Order pursuant to Bankruptcy Rules 2002(f)(7), 2002(k) and 3020(c) on all holders of Claims and Equity Interests,

- the U.S. Trustee, and the parties named on the Master Service List maintained in these Chapter 11 Cases, by causing notice substantially in the form attached hereto as **Exhibit D** to be delivered to such parties by first-class mail, postage prepaid.
- 47. <u>Notice of Effective Date.</u> Within five business days following the occurrence of the Effective Date, the Blitz Liquidating Trustee shall file notice of the Effective Date with the Court.
- 48. <u>Judicial Notice</u>. The Court takes judicial notice of the docket of these Chapter 11 Cases maintained by the clerk of the Court and/or its duly appointed agent (the "<u>Docket</u>") including, without limitation, all pleadings and other documents on file, all orders entered and all evidence and arguments made, proffered or adduced at the hearings held before the Court during the pendency of the Chapter 11 Case.
- 49. The definition of Exculpated Parties in Exhibit 1 to the Plan is deleted and replaced with the following:

Exculpated Parties: Each of (i) the Debtors and any of their respective Representatives; and (ii) the Creditors' Committee, its members and any of their respective Representatives.

- 50. The Plan and Plan Documents are modified to reflect that Wal-Mart shall make an additional \$2.0 Million contribution to the Insurance Settlement Payment. The \$2.0 Million contribution shall be allocated as follows: (a) \$1.2 Million to the Non- Appealing Fund; (b) \$700,000 to the Special Circumstances Fund and (c) \$100,000 to the Blitz Liquidating Trust.
- 51. Section 7.2.1 of the Plan is modified to include the Creditors' Committee, its members and any of their respective Representatives as receiving releases from each of the Debtors, on behalf of themselves and their respective Estates and their respective Affiliates,

members, officers, directors, and employees, and any person claiming by or through them as contemplated by such paragraph.

- 52. Section 7.2.2 of the Plan is modified to include the Creditors' Committee, its members and any of their respective Representatives as receiving releases from each present and former holder of a Claim or Equity Interest as contemplated by such paragraph.
- Parties, (iii) the USA Debtors' Representatives, (iv) the Participating Insurers and, (v) all Vendors and Additional Insureds under the Participating Policies, no other parties shall be released from, and the Releases and Channeling Injunction provisions of the Plan shall not apply to, any claim against any party that arose prior to 12:01 A.M. CST on July 31, 2007; provided however, that Westchester's rights and protections, including with regard to the Releases and the Channeling Injunction, shall be as stated in, and consistent with paragraph 10 of this Confirmation Order.
- (b) Between July 31, 2007 and July 31, 2012, holders of a Co-Defendant Claim, which holder is not a Vendor, BAH Released Party or an Additional Insured, shall not receive a release. In addition to all of the other rights and remedies afforded to (i) Wal-Mart, (ii) the BAH Released Parties, (iii) the USA Debtors' Representatives, (iv) the Participating Insurers and, (v) all Vendors and Additional Insureds under the Participating Policies under the Plan, the Plan Documents or this Confirmation Order, in the event that any of (i) Wal-Mart, (ii) the BAH Released Parties, (iii) the USA Debtors' Representatives, (iv) the Participating Insurers and, (v) all Vendors and Additional Insureds under the Participating Policies is named as a defendant or co-defendant in any action by the holder of a surviving Co-Defendant Claim, (i) Wal-Mart, (ii) the BAH Released Parties, (iii) the USA Debtors' Representatives, (iv) the Participating Insurers

and, (v) all Vendors and Additional Insureds under the Participating Policies, as the case may be, shall be entitled to recover, with priority, liability and/or reasonable defense costs from any proceeds received on account of such surviving Co-Defendant Claim.

- 54. Paragraph 2.3(b) of the Blitz Personal Injury Trust is clarified as follows: (i) holders of Blitz Personal Injury Claims that arose prior to July 31, 2005 may pursue their claims against Vendors (except for Wal-Mart), holders of Co-Defendant Claims (except for Wal-Mart and the BAH Released Parties) and other third parties liable for the Blitz Personal Injury Claim and the automatic stay is lifted as of the Effective Date and (ii) any recoveries obtained by the holder of a Blitz Personal Injury Claim shall be directly payable to such holder and shall not be payable to the Blitz Personal Injury Trust and no holder of a Blitz Personal Injury Trust Claim shall have an interest in or be entitled to a distribution from any such recoveries.
- of all materials required by the Personal Injury POC form (including medical records and expert reports) and any materials to be considered in support of a Special Circumstances claim to thirty (30) calendar days after the Confirmation Hearing. All such materials are required to be sent electronically to the Blitz Personal Injury Trust, at blitzclaim@litigationkc.com, dhaltiwanger@rpwb.com and adr@richardcohen.net so as to be received on or before February 28, 2014 by 5:00 p.m. Central Standard Time. Materials received after this time will not be considered by the Blitz Personal Injury Trust.
- 56. Notwithstanding anything to the contrary in this Confirmation Order, the Plan or the Plan Documents, the automatic stay provided for in these Chapter 11 Cases, whether pursuant to section 105, section 362 or any other provision of the Bankruptcy Code or other applicable law, in existence immediately prior to the Confirmation Date shall expire on the

Effective Date solely with respect to Blitz Personal Injury Claims that arose prior to the Release Date.

- 57. Nothing in this Confirmation Order, the Plan or the Blitz Personal Injury TDP shall be interpreted as releasing or impairing any Blitz Personal Injury Claim against any Vendor, Wal-Mart or holder of a Co-Defendant Claim for incidents occurring on or after July 31, 2102 at 12:01 CST.
- The claims of Melissa Weeks and Paul Beadore, Individually and as parent of 58. Landon Beadore, as set forth in the Complaint filed in the case of Beadore v. Stewart's Shops Corp. et al, Complaint Index No. 10062122, pending in the Supreme Court of the State of New York, Saratoga County, (the "Beadore Action"), which arise from an incident which occurred on September 16, 2003, constitute Blitz Personal Injury Claims which are excepted from the Release provisions of sections 7.2.1 and 7.2.2 by virtue of Section 7.2.3.6 of the Plan. Accordingly, and notwithstanding anything to the contrary in the Plan or any other provision of this Confirmation Order, (1) neither Central Hudson Gas & Electric Corporation nor Central Hudson Energy Services, Inc., nor CH Energy Group (collectively, "Central Hudson"), nor any other co-defendant in the Beadore Action, constitute a Protected Party entitled to a Release under the Plan; (2) the cross-claims for contribution and indemnification asserted against the Debtors by Central Hudson as well as any other co-defendant in the Beadore Action are likewise excepted from the Release and Injunction provisions of sections 7.2.1,7.2.2, and 7.3; (3) nothing in this Confirmation Order or the Plan shall prejudice or in any way alter or restrict the rights of the co-defendants in the Beadore Action to assert their cross claims and affirmative defenses as set forth in their Amended Answers; (4) the rights of the co-defendants in the Beadore Action to a recovery on their cross claims, if any, from the Debtors or the Blitz Personal Injury Trust, shall

Case 11-13603-PJW Doc 2152 Filed 01/30/14 Page 53 of 53

be limited to the coverage available from the Assigned Blitz Insurance Policies assigned to the Blitz Personal Injury Trust; (5) for the avoidance of doubt, the rights of Central Hudson, pursuant to Article 16 of the New York Civil Practice Laws and Rules, New York State General Obligations Law §15-108, as well as any other rights claimed in the affirmative defenses and cross claims contained in the Amended Verified Answer filed on behalf of Central Hudson, if any, are unaffected by this Confirmation Order or the Plan, and (6) nothing in this paragraph is intended to in any way limit the right of the plaintiffs in the above-described Beadore Action to collect the full amount of any judgment in such action against any defendant in such action under their joint and several liability as determined under New York law, including Article 16 of the CPLR. For the avoidance of doubt, none of the BAH Released Parties party to the Beadore Action.

Wilmington, Delaware

THE HONORABLE PETER J. WALSH UNITED STATES BANKRUPTCY JUDGE

EXHIBIT A

Insurance Settlement Buy-Back Order

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re:

Chapter 11

BLITZ U.S.A., Inc., et al., 1

Case No. Case No. 11-13603 (PJW)

Debtors.

(Jointly Administered)

Re: Docket No. 1537

ORDER UNDER 11 U.S.C. §§ 105 AND 363 AND FED. R. BANKR. P. 2002, 6004 AND 9019 APPROVING SETTLEMENT AND AUTHORIZING DEBTORS TO (I) COMPROMISE, SETTLE, RELEASE AND DISMISS CERTAIN CLAIMS OF AND AGAINST THE DEBTORS PURSUANT TO INSURANCE TERM SHEET AND (II) SELL CERTAIN INSURANCE POLICIES BACK TO THE PARTICIPATING INSURERS FREE AND CLEAR OF ALL LIENS, CLAIMS, INTERESTS AND OTHER ENCUMBRANCES

This matter, coming before the Court on the Motion for Order Under 11 U.S.C. §§ 105 and 363 and Fed. R. Bankr. P. 2002, 6004 and 9019 Approving Settlement and Authorizing Debtors to: (I) Compromise, Settle, Release and Dismiss Claims of and Against the Debtors Pursuant to Insurance Term Sheet and (II) Sell Certain Insurance Policies Back to the Participating Insurers Free and Clear of Liens, Claims, Interests and Other Encumbrances [Docket No. 1537] filed on July 24, 2013 (the "Insurance Settlement Motion"); and the Court having entered on August 14, 2013, the Scheduling Order With Respect to Evidentiary Hearing on Motion for Order Under 11 U.S.C. §§ 105 and 363 and Fed. R. Bankr. P. 2002, 6004 and

The Debtors in these chapter 11 cases, along with the last four digits of each Debtors' federal tax identification number, include: LAM 2011 Holdings, LLC (8742); Blitz Acquisition Holdings, Inc. (8825); Blitz Acquisition, LLC (8979); Blitz RE Holdings, LLC (9071); Blitz U.S.A., Inc. (8104); and F3 Brands LLC (2604). The location of the Debtors' corporate headquarters and the Debtors' service address is 394 North Main Street, Miami, OK 74354.

Capitalized terms not otherwise defined herein shall have the meaning given to such terms in the Insurance Settlement Motion.

9019 Approving Settlement and Authorizing Debtors to (I) Compromise, Settle Release and Dismiss Claims of and Against the Debtors Pursuant to Insurance Term Sheet and (II) Sell Certain Insurance Policies Back to the Participating Insurers Free and Clear of Liens, Claims, Interests and Other Encumbrances [Docket No. 1671] (the "Insurance Motion Scheduling Order"), pursuant to which the Court set an evidentiary hearing, initially for November 19, 2013, but thereafter adjourned to December 18, 2013 (the "Hearing"), on the following requests for relief in connection with the Motion (collectively, the "Insurance Settlement"):

- (a) Whether the settlement set forth and described in the Insurance Settlement Motion between Debtors and the Participating Insurers satisfies the standards for settlements under Fed. R. Bankr. P. 9019, as set forth in *In re Martin*, 91 F.3d 389, 393 (3d Cir. 1996); *In re RFE Indus., Inc.*, 283 F.3d 159, 165 (3d Cir. 2002); and *In re Nutraquest, Inc.*, 434 F.3d 639, 644 (3d Cir. 2006), including: (i) the probability of success in the litigation; (ii) the difficulties associated with collection; (iii) the complexity of the litigation, and attendant expense, inconvenience and delay; and (iv) the paramount interests of creditors; and
- (b) Whether the Subject Policies are property of the Debtors' estates;
- (c) Whether there is sound business justification for Debtors' agreement to sell the Subject Policies to the Participating Insurers as set forth and described in the Insurance Settlement Motion;
- (d) Whether the Participating Insurers' buyback of the Subject Policies is in good faith within the meaning of 11 U.S.C. § 363(m); and
- (e) Whether it is appropriate under the circumstances for Debtors to sell the Subject Policies free and clear of all interests in the Subject Policies under 11 U.S.C. § 363(f).

The Court, after notice and hearing, having determined that the legal and factual bases set forth in the Insurance Settlement Motion and at the Hearing establish just cause for the relief granted herein; and the Court, hereby incorporating the record made at the Hearing on the Insurance Settlement Motion; and the objections of the Texas Claimants to the Insurance Settlement having been withdrawn without prejudice upon and subject to the terms of the Plan Support Agreement (the "Plan Support Agreement") attached to the Disclosure Statement for Debtors' and Official Committee of Unsecured Creditors First Amended Plan of Liquidation; and the Court, having considered and overruling any remaining objections to the Insurance Settlement Motion and the Insurance Settlement; hereby finds and concludes as follows:³

<u>Findings</u>: The Court by this Order (this "Insurance Settlement Approval Order") hereby finds that

- (a) it has jurisdiction over the Motion pursuant to 28 U.S.C. §§ 157 and 1334;
- (b) this contested matter is a core proceeding pursuant to 28 U.S.C. § 157(b);
- (c) notice to an attorney for the holder of a Blitz Personal Injury Claim constitutes notice to such holder for purposes of notice of the Insurance Settlement Motion, the Hearing, the Term Sheet, and this Insurance Settlement Approval Order;
- (d) notice of the Insurance Settlement Motion and Hearing was sufficient under the circumstances and in full compliance with 11 U.S.C. § 102(1), Fed. R. Bankr. P. 2002 and the Local Rules of this Court;

This Insurance Approval Order constitutes the Court's findings and conclusions of law under and pursuant to Fed. R. Bankr. P. 7052. To the extent necessary and appropriate, any finding made by the Court in this Insurance Approval Order shall be considered as a conclusion of law; and, to the extent necessary and appropriate, any conclusion of law made by the Court in this Insurance Approval Order shall be considered as a finding.

Case 11-13603-PJW Doc 2152-1 Filed 01/30/14 Page 5 of 8 Case 11-13603-PJW Doc 2011 Filed 12/20/13 Page 4 of 7

(e) each of the following factors has been taken into account by the Settling Parties in

reaching the compromises with respect to the Insurance Settlement, and each factor supports

entry of this Insurance Settlement Approval Order pursuant to Bankruptcy Rule 9019: (i) the

probability of success in the litigation of any claims among the Settling Parties, including the

ranges of recoverable damages, (ii) the complexity of potential litigation among the Settling

Parties, and the expense, inconvenience, and delay necessarily attending it, (iii) the paramount

interest of creditors and proper deference to their reasonable views regarding the Insurance

Settlement, and (iv) whether the Insurance Settlement promotes the integrity of the judicial

system;

(f) the Insurance Settlement (and the Term Sheet) (i) is the product of good faith, arms'

length negotiation among the Settling Parties, without collusion, (ii) is fair, reasonable,

appropriate, and in the best interests of the Debtors' estates, and (iii) represents a sound exercise

of the Debtors' business judgment;

(g) the Insurance Settlement is supported by proper and sufficient consideration provided

by each of the Settling Parties;

(h) the Subject Policies are property of the Debtors and of the Debtors' estates pursuant

to 11 U.S.C. § 541(a),

(i) this Court has the jurisdiction and power to approve the Insurance Settlement as set

forth in the Term Sheet; and

(j) this Insurance Settlement Approval Order is "final" within the meaning of 28 U.S.C. §

158(a)(1).

Conclusions of Law and Order:

IT IS HEREBY ORDERED THAT:

- 1. The Insurance Settlement Motion, to the extent it seeks approval of the Insurance Settlement, is hereby GRANTED, as set forth herein.
- 2. The Insurance Settlement, in accordance with the terms of the Term Sheet as such relates to the Insurance Settlement under and pursuant to Bankruptcy Rule 9019, and the sale by the Debtors of the Subject Policies to the Participating Insurers, free and clear of all interests in the Subject Policies under and pursuant to 11 U.S.C. § 363, is hereby APPROVED.
- 3. The Court hereby finds and concludes that the Participating Insurers' buyback of the Subject Policies is in good faith within the meaning of 11 U.S.C. § 363(m) and, therefore, the protections of section 363(m) apply.
- 4. The entry of this Order is without prejudice to any of the rights of the Debtors, the Committee, Wal-Mart, the Texas Claimants,⁴ or the Participating Insurers with respect to the Subject Policies and their construction in the event that confirmation of the Plan as a condition for implementation of the buyback of the Subject Policies is not met. Further, should the Plan not be confirmed or not go effective, and the buyback not be implemented for whatever reason, none of the evidence adduced at the Hearing or findings made in this Order shall be binding upon the Debtors, the Committee, Wal-Mart, the Texas Claimants or the Participating Insurers.
- 5. If, prior to or at the hearing to consider confirmation of the First Amended Joint Plan of Liquidation (the "Confirmation Hearing"), the Texas Claimants have actually terminated

[&]quot;Texas Claimants," as used herein, mean and refer to Anthony Torres, Andrew Torres, Valerie Olivarez Torres, Colton Warren, Cindy Nichols, Tiffany St. John, Marion "B.J." Jones, Dalan Jones, a minor child represented by his parents Tiffany St. John and Marion Jones, and the Estate of Leiya Jones, a deceased minor child represented by her parents Tiffany Jones, Marion Jones, Irene Perez on behalf of Aliha Perez (Minor), Irene Perez individually, and Jose Perez individually.

Case 11-13603-PJW Doc 2152-1 Filed 01/30/14 Page 7 of 8 Case 11-13603-PJW Doc 2011 Filed 12/20/13 Page 6 of 7

the Plan Support Agreement pursuant to Paragraph 6 thereof, then their right to raise their objections to the Insurance Settlement at the Confirmation Hearing shall be preserved.

- 6. This Court shall retain jurisdiction to (i) enforce and interpret the terms and provisions of this Insurance Settlement Approval Order and the Term Sheet, (ii) resolve any disputes, controversies, or claims arising out of or relating to this Insurance Settlement Approval Order or the Term Sheet, and (iii) interpret, implement, and enforce the provisions of this Insurance Settlement Approval Order or any other Order of this Court entered in these Cases.
- 7. This Insurance Settlement Approval Order is intended to approve the Insurance Settlement in accordance with the Term Sheet and the omission of any term or condition of the Insurance Settlement herein shall not diminish or impair the effectiveness of such term and provision, it being the intent of the Court that the Insurance Settlement, as more specifically set forth and described in the Term Sheet, be authorized and approved in its entirety by this Insurance Settlement Approval Order.

Case 11-13603-PJW Doc 2152-1 Filed 01/30/14 Page 8 of 8 Case 11-13603-PJW Doc 2011 Filed 12/20/13 Page 7 of 7

8. This Insurance Settlement Approval Order shall be effective and enforceable immediately upon entry and its provisions shall be self-executing and shall not be stayed under Bankruptcy Rule 6004(g).

Dated: December **2 4**013 Wilmington, Delaware

The Honorable Peter J. Walsh United States Bankruptcy Judge

EXHIBIT B

Blitz Personal Injury Trust Agreement

BLITZ PERSONAL INJURY TRUST AGREEMENT

TABLE OF CONTENTS

SECTION 1	AGREEMENT OF TRUST	2
1.1	Creation and Name	2
1.2	Purpose	
1.3	Transfer of Assets	
1.4	Acceptance of Assets and Assumption of Liabilities	3
SECTION 2	POWERS AND TRUST ADMINISTRATION	5
2.1	Powers	5
2.2	General Administration	9
2.3	Claims Administration.	13
2.4	MSP Claims Reporting	14
2.5	Payment of MSP Obligations	17
2.6	Indemnification for Medicare Claims Reporting and Payment Obligations.	18
SECTION 3	ACCOUNTS, INVESTMENTS, AND PAYMENTS	18
3.1	Accounts	18
3.2	Investments	
3.3	Source of Payments	
SECTION 4	BLITZ PERSONAL INJURY TRUSTEE	21
4.1	Number	
4.2	Term of Service	
4.3	Appointment of Successor Blitz Personal Injury Trustee	22
4.4	Liability of Blitz Personal Injury Trustee, Blitz Personal Injury TAC	. 22
4.5	Compensation and Expenses of Blitz Personal Injury Trustee	
4.6	Blitz Personal Injury Trustee's Employment of Experts	
4.7	Blitz Personal Injury Trustee's Independence	
4.8	Bond	
	TRUST ADVISORY COMMITTEE	
5.1	Members	
5.2	Duties	
5.3	Term of Office	
5.4	Appointment of Successor Members	
5.5	Blitz Personal Injury TAC's Employment of Professionals	
5.6	Compensation and Expenses of Blitz Personal Injury TAC	
SECTION 6	GENERAL PROVISIONS	29
6.1	Procedures for Consultation with Blitz Personal	
	Injury TAC	29

Case 11-13603-PJW Doc 2152-2 Filed 01/30/14 Page 4 of 47

6.2	Indemnification	30
6.3	Irrevocability	31
6.4	Termination	31
6.5	Amendments	32
6.6	Severability	33
6.7	Notices	
6.8	Successors and Assigns	
6.9	Entire Agreement; No Waiver	35
6.10	Headings	
6.11	Governing Law	35
6.12	Debtors' Representations and Cooperation	35
6.13	Dispute Resolution	
6.14	Enforcement and Administration	35
6.15	Effectiveness	36
6.16	Counterpart Signatures	36

BLITZ PERSONAL INJURY TRUST AGREEMENT¹

This Blitz Personal Injury Trust Agreement (the "Agreement"), dated as of _______, 2013, and effective as of the later of the date on which all parties to this Agreement have executed this Agreement or the Effective Date of the Plan, is entered into by Blitz USA, Inc., Blitz Acquisition, LLC, Blitz RE Holdings, LLC, and MiamiOK LLC f/k/a F3Brands LLC (collectively, the "USA Debtors") and LAM 2011 Holdings, LLC and Blitz Acquisition Holdings, Inc. (the "BAH Debtors", and together with the USA Debtors, the "Debtors" or the "Settlors"), debtors and debtors-in-possession in the Chapter 11 Cases, as Settlors and by the Blitz Personal Injury Trustee and the Blitz Personal Injury Trust Advisory Committee ("Blitz Personal Injury TAC") identified on the signature page hereof.

WHEREAS, on the Petition Date, each of the Debtors filed voluntary petitions for relief under Chapter 11 of the Bankruptcy Code with the Bankruptcy Court. The Chapter 11 Cases, which are captioned *In re Blitz USA*, *Inc. et al.*, Case No. 11-13603 (PJW), are being jointly administered by the Bankruptcy Court.

WHEREAS, as of the Petition Date, some of the Debtors had been named as defendants in Blitz Product Litigation.

WHEREAS, the Plan was confirmed by the Confirmation Order dated ______, 2014 and entered by the Bankruptcy Court.

WHEREAS, the Plan Documents, provide, among things, for the creation of the Blitz Personal Injury Trust.

WHEREAS, pursuant to the Plan, the Blitz Personal Injury Trust shall be established to, among other things, assume liability for all Blitz Personal Injury Trust Claims; administer, process, settle, resolve and liquidate such Blitz Personal Injury Trust Claims; and to use the Blitz Personal Injury Trust Assets to satisfy all such Blitz Personal Injury Trust Claims that may qualify for a recovery in accordance with the terms of this Agreement and the Blitz Personal

¹ All capitalized terms not otherwise defined herein shall have their respective meanings as set forth in the *Debtors'* and Official Committee of Unsecured Creditors' Amended Joint Plan of Liquidation, as may be amended, modified or supplemented from time to time (the "Plan") and such definitions are incorporated herein by reference. All capitalized terms not defined herein or in the Plan, but defined in the Bankruptcy Code or Federal Rules of Bankruptcy Procedure ("Bankruptcy Rules"), shall have the meanings given to them by the Bankruptcy Code and Bankruptcy Rules, and such definitions are incorporated herein by reference.

Injury TDP; preserve, hold, manage, maximize and use the Blitz Personal Injury Trust Assets to administer, process, settle, resolve, liquidate and make payments as may be appropriate to holders of Blitz Personal Injury Trust Claims in accordance with this Agreement, the Blitz Personal Injury TDP and the Plan Documents.

WHEREAS, the Plan provides that on the Effective Date, the Blitz Personal Injury Trust Assets will be transferred to the Blitz Personal Injury Trust.

WHEREAS, on the Effective Date, the Debtors shall assign the Assigned Blitz Insurance Policies to the Blitz Personal Injury Trust.

WHEREAS, pursuant to the Plan, the Blitz Personal Injury Trust is intended to qualify as a "qualified settlement fund" within the meaning of section 1.468B-1 *et seq.* of the Treasury Regulations promulgated under Section 468B of the Internal Revenue Code (the "IRC").

WHEREAS, it is the intent of the Settlors, the Proponents, the Blitz Personal Injury Trustee and the Blitz Personal Injury TAC that the Blitz Personal Injury Trust be administered, maintained, and operated at all times through mechanisms that provide reasonable assurance that the Blitz Personal Injury Trust will administer, process, settle, resolve, liquidate, satisfy and pay, if applicable, all Blitz Personal Injury Trust Claims pursuant to this Blitz Personal Injury Trust Agreement and the Blitz Personal Injury TDP in compliance with the terms of this Agreement.

WHEREAS, it is the intent of the Settlors, the Proponents, the Blitz Personal Injury Trustee and the Blitz Personal Injury TAC that holders of Covered Blitz Personal Injury Claims shall receive Offer Amounts that are substantially in accordance with the amounts listed on Exhibit B to the Blitz Personal Injury TDP assuming that (i) the Covered Claimants can support the level of injury and other compensation sought in their respective proofs of claim (with the understanding that (a) Covered Claimants who provide documentation supporting a higher recovery than sought in their proof of claim may receive Offer Amounts greater than the amounts listed for such claimant on Exhibit B, which would impact all Covered Claimants' Offer Amounts, and (b) Covered Claimants who provide documentation supporting a lower recovery than sought in their proof of claim may receive Offer Amounts less than the amounts listed for such claimant on Exhibit B, which would impact all Covered Claimants' Offer Amounts), and (ii) the Covered Claimants can satisfy the Threshold Components set forth in the TDP Scoring System.

WHEREAS, the Bankruptcy Court has determined that the Blitz Personal Injury Trust and the Plan satisfy all the legal prerequisites for issuing the Channeling Injunction pursuant to section 105 of the Bankruptcy Code, and such Channeling Injunction has been entered in connection with the Confirmation Order.

NOW, THEREFORE, for good and valuable consideration, it is hereby agreed as follows:

SECTION 1

AGREEMENT OF TRUST

- 1.1. <u>Creation and Name.</u> The Debtors, as Settlors, hereby create the Blitz Personal Injury Trust, which is the trust provided for and referred to in Article IV of the Plan. The Blitz Personal Injury Trustee may transact the business and affairs of the Blitz Personal Injury Trust in the name of the Blitz Personal Injury Trust.
- 1.2 Purpose. On the Effective Date, the Blitz Personal Injury Trust shall be established in accordance with the Plan Documents. The Blitz Personal Injury Trust shall be a "Qualified Settlement Fund" within the meaning of section 468B of the Internal Revenue Code and the regulations promulgated thereunder. The Blitz Personal Injury Trust shall assume the liability for all Blitz Personal Injury Trust Claims; shall administer, process, settle, resolve and liquidate such Blitz Personal Injury Trust Claims; and shall use the Blitz Personal Injury Trust Assets and the proceeds and income therefrom to satisfy and make payment to all Blitz Personal Injury Claims that may qualify for a recovery only in accordance with the terms of this Blitz Personal Injury Trust Agreement and the Blitz Personal Injury TDP, all in accordance with the Plan, the Blitz Personal Injury Trust Agreement and the Blitz Personal Injury TDP. In this regard, holders of Covered Blitz Personal Injury Claims shall receive Offer Amounts that are substantially in accordance with the amounts listed on Exhibit B to the Blitz Personal Injury TDP assuming that (i) the Covered Claimants can support the level of injury and other compensation sought in their respective proofs of claim (with the understanding that (a) Covered Claimants who provide documentation supporting a higher recovery than sought in their proof of claim may receive Offer Amounts greater than the amounts listed for such claimant on Exhibit B, which would impact all Covered Claimants' Offer Amounts, and (b) Covered Claimants who provide documentation supporting a lower recovery than sought in their proof of claim may receive Offer

Amounts less than the amounts listed for such claimant on Exhibit B, which would impact all Covered Claimants' Offer Amounts), and (ii) the Covered Claimants can satisfy the Threshold Components set forth in the TDP Scoring System. The Blitz Personal Injury Trust will (i) administer, process, settle, resolve, liquidate, satisfy and/or pay, as applicable, Blitz Personal Injury Trust Claims in such a way that the holders of Blitz Personal Injury Claims are treated equitably and in a substantially similar manner, subject to the terms of the Plan, the Blitz Personal Injury Trust Agreement and the Blitz Personal Injury TDP and (ii) in accordance with section 4.14 of the Plan and the Blitz Personal Injury Trust Agreement, defend and indemnify the Protected Parties, at the Blitz Personal Injury Trust's sole expense, in connection with any proceeding involving, relating to or arising out of, in whole or in part, the enforcement or enforceability of the Channeling Injunction. Blitz Personal Injury Trust Claims shall be channeled to the Blitz Personal Injury Trust pursuant to the Channeling Injunction set forth in section 4.3.3 of the Plan and may thereafter be asserted only and exclusively against the Blitz Personal Injury Trust. All such Blitz Personal Injury Claims shall be liquidated and paid in accordance with this Blitz Personal Injury Trust Agreement, the Blitz Personal Injury TDP, the Plan and the Confirmation Order. For the avoidance of doubt, notwithstanding anything to the contrary in this Blitz Personal Injury Trust Agreement, the Blitz Personal Injury TDP, the Plan or the Confirmation Order, from among the categories of Blitz Personal Injury Trust Claims, only Blitz Personal Injury Claims that are Covered Blitz Personal Injury Claims shall be entitled to payment or compensation from the proceeds of the Insurance Settlement Payment (except for Michael Montgomery, whose sole recovery from the Blitz Personal Injury Trust is set forth below at paragraph 2.2(d)). The Blitz Personal Injury Trust shall be administered and implemented by the Blitz Personal Injury Trustee as provided in this Blitz Personal Injury Trust Agreement.

1.3 <u>Transfer of Assets.</u> Pursuant to Section 4.5 of the Plan, the USA Debtors, the Participating Insurers, Wal-Mart and the Blitz Liquidating Trust will transfer, issue or assign as appropriate and deliver to the Blitz Personal Injury Trust, the Blitz Personal Injury Trust Assets at the time and in the manner contemplated by the Plan Documents, in each case free and clear of any Claims, Liens, Equity Interests or other interests of any creditor, shareholder or other Entity.

1.4 Acceptance of Assets and Assumption of Liabilities.

- (a) In furtherance of the purposes of the Blitz Personal Injury Trust, the Blitz Personal Injury Trustee, on behalf of the Blitz Personal Injury Trust, hereby expressly accepts the transfer and assignment to the Blitz Personal Injury Trust of the Blitz Personal Injury Trust Assets in the time and manner contemplated by the Plan. The Blitz Personal Injury Trust shall receive the benefits of and shall be bound by the Insurance Settlement as if it had been a party thereto at the time of execution of the Insurance Settlement Term Sheet.
- (b) In furtherance of the purposes of the Blitz Personal Injury Trust, the Blitz Personal Injury Trust, expressly assumes liability for all Blitz Personal Injury Trust Claims, subject to and as provided in the Plan and the Plan Documents. The Blitz Personal Injury Trust shall have all defenses, crossclaims, offsets, and recoupments, as well as rights of indemnification, contribution, subrogation, and similar rights, regarding such claims that the Debtors and/or the Protected Parties had, have or would have had under the Plan, applicable law or under any agreement related thereto; *provided, however*, that the Blitz Personal Injury Trust shall not have any cross-claims, rights of recovery, reimbursement, defense, indemnity, offset, recoupment, contribution, subrogation and similar rights against any Protected Party or the Proponents. Regardless of the foregoing, however, to be eligible to receive any payment from the Blitz Personal Injury Trust on account of a Covered Blitz Personal Injury Claim, a Claimant's Covered Blitz Personal Injury Claim must not be barred by any applicable federal, state or foreign statute of limitations or repose.
- (c) No provision herein or in the Blitz Personal Injury TDP shall be construed to mandate distributions on any Blitz Personal Injury Claim or other actions that would contravene the Blitz Personal Injury Trust's compliance with the requirements of a "qualified settlement fund" within the meaning of section 1.468B-1 et seq. of the Treasury Regulations promulgated under section 468B of the IRC.
- (d) The Blitz Personal Injury Trust shall fully and completely defend each of the Debtors, the present and former directors and officers of each of the Debtors, the Participating Insurers, Wal-Mart, the BAH Released Parties, shareholders of the Debtors solely in their capacity as shareholders and solely related to the Blitz Personal Injury

Claims, and the Representatives, Affiliates, subsidiaries and/or successors of each of the foregoing connection with any proceeding involving, relating to or arising out of, in whole or in part, the enforcement or enforceability of the Channeling Injunction as set forth in section 4.14 of the Plan. In the event any person or Entity asserts any claim that is subject to the Channeling Injunction, the Blitz Personal Injury Trust shall, at its own expense, defend the validity of the Channeling Injunction and its application and use its best efforts to establish that such Claim is enjoined by filing a motion to defend the Channeling Injunction in the Bankruptcy Court or the Court where the underlying claim is filed. The Participating Insurers, Wal-Mart and the BAH Settling Parties (i) shall have consultation and approval rights with respect to the selection of counsel hired by the Blitz Personal Injury Trust for such defense obligations and (ii) shall have the right, at their own cost and expense, to associate in the defense in any proceeding concerning the enforcement and application of the Channeling Injunction. If the Blitz Personal Injury Trust breaches its duty to fully and completely defend the Protected Parties, the Blitz Personal Injury Trust is obligated to indemnify the Protected Parties, including advancement of defense costs. All defense and indemnity costs described in this section shall be assessed pro rata against the Non-Appealing Fund and the Special Circumstances Fund as set forth in the TDP Scoring System provided, however, that in the event that either the Non-Appealing Fund or the Special Circumstances Fund is unable to satisfy its pro rata obligations to the Protected Parties, such obligations shall be satisfied in full by the other Fund.

- (e) Nothing in this Agreement shall be construed in any way to limit or expand the scope, enforceability, or effectiveness of the Channeling Injunction issued in connection with the Plan or the Blitz Personal Injury Trust's assumption of all liability for Blitz Personal Injury Trust Claims.
- (f) The sole and exclusive remedy on account of Blitz Personal Injury Trust Claims shall be against the Blitz Personal Injury Trust, and no Blitz Personal Injury Trust Claim that has been channeled to the Blitz Personal Injury Trust may be asserted against any Protected Party or any Covered Claimant individually. Should any action based on or arising from or in connection with any Blitz Personal Injury Trust Claim be commenced against any of the Debtors, the present and former directors and officers of each of the Debtors, the Participating Insurers, Wal-Mart, the BAH Released Parties, shareholders of

the Debtors solely in their capacity as shareholders and solely related to the Blitz Personal Injury Claims, and the Representatives, Affiliates, subsidiaries and/or successors of each of the foregoing Protected Party in any tribunal whatsoever, the Blitz Personal Injury Trust shall seek to enjoin the prosecution of such action in the Bankruptcy Court in connection with these Chapter 11 Cases or in the jurisdiction where the underlying action is filed.

SECTION 2

POWERS AND TRUST ADMINISTRATION

2.1 Powers.

- (a) The Blitz Personal Injury Trustee is and shall act as the fiduciary to the Blitz Personal Injury Trust in accordance with the provisions of this Agreement and the Plan, and shall have the power, on behalf of the Blitz Personal Injury Trust, to exercise all rights and fulfill all obligations of the Blitz Personal Injury Trust hereunder and under the Plan. The Blitz Personal Injury Trustee shall, at all times, administer the Blitz Personal Injury Trust and the Blitz Personal Injury Trust Assets in accordance with the purposes set forth in section 1.2 above. Subject to the limitations set forth in this Agreement, the Blitz Personal Injury Trustee shall have the power to take any and all actions that, in the judgment of the Blitz Personal Injury Trustee, are necessary or proper to fulfill the purposes of the Blitz Personal Injury Trust, including, without limitation, each power expressly granted in this section 2.1, any power reasonably incidental thereto, and any trust power now or hereafter permitted under the laws of the State of Delaware.
- (b) Except as required by applicable law or otherwise specified herein, the Blitz Personal Injury Trustee need not obtain the order or approval of any court in the exercise of any power or discretion conferred on the Blitz Personal Injury Trustee hereunder.
- (c) Without limiting the generality of section 2.1(a) above, and except as limited below, the Blitz Personal Injury Trustee shall have the power to:
 - (i) receive and hold the Blitz Personal Injury Trust Assets;
 - (ii) invest the monies held from time to time by the Blitz Personal Injury
 Trust;

- (iii) enter into leasing and financing agreements with third parties to the extent such agreements are reasonably necessary to permit the Blitz Personal Injury Trust to operate;
- (iv) pay the Blitz Personal Injury Trust Expenses;
- (v) in accordance with section 3.1 below, establish such reserves and accounts for the Blitz Personal Injury Trust Expenses, with the Blitz Personal Injury Trust Assets as deemed by the Blitz Personal Injury Trustee to be useful in carrying out the purposes of the Blitz Personal Injury Trust;
- (vi) sue and be sued and participate, as a party or otherwise, in any judicial, administrative, arbitrative, or other proceeding;
- (vii) establish, supervise and administer the Blitz Personal Injury Trust in accordance with the Plan, this Agreement and the Blitz Personal Injury TDP, and take any and all actions contemplated to be taken by the Blitz Personal Injury Trustee under the Blitz Personal Injury TDP;
- (viii) engage legal, financial, accounting, investment, and auditing consultants and agents as the business of the Blitz Personal Injury Trust may require; and to delegate to such persons power and authority as the fiduciary duties of the Blitz Personal Injury Trustee permit and as the Blitz Personal Injury Trustee, in his or her discretion, deems advisable or necessary to carry out the purposes of the Blitz Personal Injury Trust in accordance with the Plan and this Agreement;
 - (ix) pay reasonable compensation to legal, financial, accounting, investment, and auditing consultants and agents hired or retained by the Blitz Personal Injury Trust, including, without limitation, any such persons hired or retained in connection with the alternative dispute resolution and litigation activities of the Blitz Personal Injury Trust, and allocate such costs, as permitted under the TDP Scoring System, to one or more Covered Claimants as an Individual Cost or as a General Cost (as defined and using the methodology described in the TDP Scoring System);

- (x) compensate the Blitz Personal Injury Trustee and the Blitz Personal Injury TAC member and reimburse the Blitz Personal Injury Trustee and the Blitz Personal Injury TAC members for reasonable out-of-pocket costs and expenses incurred by such persons in connection with the performance of their duties hereunder;
- (xi) execute and deliver such instruments as the Blitz Personal Injury Trustee considers proper in administering the Blitz Personal Injury Trust;
- (xii) enter into such other arrangements with third parties as are deemed by the Blitz Personal Injury Trustee to be reasonably necessary in carrying out the purposes of the Blitz Personal Injury Trust, provided such arrangements do not conflict with any other provision of this Agreement;
- (xiii) defend and indemnify, and if appropriate purchase insurance indemnifying (A) the Blitz Personal Injury Trustee and (B) the Blitz Personal Injury TAC, to the fullest extent that a corporation or trust organized under the law of the State of Delaware is from time to time entitled to indemnify and/or insure its directors, officers, employees, agents, advisors and representatives (as set forth in the TDP Scoring System, the foregoing indemnification costs shall be borne pro rata between the Non-Appealing Fund and the Special Circumstances Fund in the manner described therein);
- (xiv) in accordance with section 1.4(d) above and section 4.14 of the Plan, defend the Channeling Injunction;
- (xv) delegate any or all of the authority herein conferred with respect to the investment of all or any portion of the Blitz Personal Injury Trust Assets to any one or more reputable individuals or recognized institutional investment advisors or investment managers without liability for any action taken or omission made because of any such delegation, except as provided in section 4.4 below;

- (xvi) consult with the Blitz Personal Injury TAC at such times and with respect to such issues relating to the conduct of the Blitz Personal Injury Trust as the Blitz Personal Injury Trustee considers desirable; and
- (xvii) make, pursue (by litigation or otherwise), collect, compromise or settle, in the name of the Blitz Personal Injury Trust, any Claim, right, action, or cause of action, if any, included in the Blitz Personal Injury Trust Assets before any court of competent jurisdiction; provided that the Blitz Personal Injury Trust must seek approval of any settlement of actions that are pending before the Bankruptcy Court from the Bankruptcy Court after notice to parties in interest;
- (xviii) process, resolve and object to Blitz Personal Injury Trust Claims as provided in the Plan, this Agreement and the Blitz Personal Injury TDP.
- (d) The Blitz Personal Injury Trustee shall not have the power to cause the Blitz Personal Injury Trust to guarantee debt of any other Entity.
- (e) The Blitz Personal Injury Trustee shall give the Blitz Personal Injury TAC prompt notice of any act performed or taken pursuant to sections 2.2(c)(i) and (ii), and 2.2(d), and any act proposed to be performed or taken pursuant to section 2.2(f) below.
- (f) The Blitz Personal Injury Trustee shall have the power, but not the obligation, to, at the request and sole cost and expense of a Blitz Personal Injury Claimant whose claim arose prior to the Release Date, seek or sue for insurance coverage proceeds only in connection with the rights transferred to the Blitz Personal Injury Trust pursuant to Article IV of the Plan and the Assigned Blitz Insurance Policies.

2.2 General Administration.

(a) The Blitz Personal Injury Trustee shall act in accordance with this Agreement, the Blitz Personal Injury TDP (including the TDP Scoring System attached thereto and incorporated therein), and the Plan. In the event of an inconsistency between the Plan and this Agreement, the Plan shall govern. In the event of any inconsistency between this Agreement and the Blitz Personal Injury TDP, the Blitz Personal Injury TDP shall control. The Blitz Personal Injury Trust may, at the sole discretion of the Blitz Personal Injury

Trustee, adopt bylaws (if adopted, "Blitz Personal Injury Trust Bylaws"). The Blitz Personal Injury Trust Bylaws cannot supersede this Agreement, the Plan, the Blitz Personal Injury TDP, or the Blitz Personal Injury TDP Scoring System. To the extent not inconsistent with the terms of this Agreement, if and as when adopted the Blitz Personal Injury Trust Bylaws shall govern the affairs of the Blitz Personal Injury Trust. In the event of an inconsistency between Blitz Personal Injury Trust Bylaws and this Agreement, this Agreement shall govern.

- (b) The Blitz Personal Injury Trustee shall: (i) timely file such income tax and other returns and statements required to be filed by the Blitz Personal Injury Trust and shall from the Blitz Personal Injury Trust Assets timely pay all taxes required to be paid by the Blitz Personal Injury Trust, (ii) comply with all withholding obligations, as required under the applicable provisions of the IRC and of any state law and any regulations promulgated thereunder, (iii) meet without limitation all requirements necessary to qualify and maintain qualification of the Blitz Personal Injury Trust as a qualified settlement fund within the meaning of section 1.468B-1 et seq. of the Treasury Regulations promulgated under section 468B of the IRC, and (iv) take no action that could cause the Blitz Personal Injury Trust to fail to qualify as a qualified settlement fund within the meaning of section 1.468B-1 et seq. of the Treasury Regulations promulgated under section 468B of the IRC. As set forth in the TDP Scoring System, the foregoing reporting costs shall be borne pro rata between the Non-Appealing Fund and the Special Circumstances Fund in the manner described therein.
- (c) The Blitz Personal Injury Trustee shall timely account to the Bankruptcy Court as follows:
 - (i) Depending on whether the Blitz Personal Injury Trustee decides, in his or her sole discretion, to operate the Blitz Personal Injury Trust on a calendar or fiscal year basis, the Blitz Personal Injury Trustee shall cause to be prepared and filed with the Bankruptcy Court, as soon as available, but in no event later than one hundred and twenty (120) days following the end of each calendar or fiscal year, an annual report containing, *inter alia*, financial statements of the Blitz Personal Injury Trust (including, without limitation,

- a balance sheet of the Blitz Personal Injury Trust as of the end of such fiscal year and a statement of operations for such fiscal year) audited by a firm of independent certified public accountants selected by the Blitz Personal Injury Trustee and accompanied by an opinion of such firm as to the fairness of the financial statements' presentation of the cash and investments available for the payment of claims and as to the conformity of the financial statements with generally accepted accounting principles. The Blitz Personal Injury Trustee shall provide a copy of such report to the Blitz Personal Injury TAC, when such reports are filed with the Bankruptcy Court.
- (ii) Simultaneously with delivery of each set of financial statements referred to in section 2.2(c)(i) above, the Blitz Personal Injury Trustee shall cause to be prepared and filed with the Bankruptcy Court a report containing a summary regarding the number and type of Blitz Personal Injury Trust Claims disposed of during the period covered by the financial statements. The Blitz Personal Injury Trustee shall provide a copy of such report to the Blitz Personal Injury TAC, when such report is filed with the Bankruptcy Court.
- (iii) All materials required to be filed with the Bankruptcy Court by this section 2.2(c) shall be available for inspection by the public in accordance with procedures established by the Bankruptcy Court.
- (iv) As set forth in the TDP Scoring System, the foregoing reporting costs shall be borne pro rata between the Non-Appealing Fund and the Special Circumstances Fund in the manner described therein.
- (d) The Blitz Personal Injury Trustee shall cause to be prepared as soon as practicable prior to the commencement of each calendar or fiscal year a budget and cash flow projections covering such calendar or fiscal year. The Blitz Personal Injury Trustee shall provide a copy of the budget and cash flow projections to the Blitz Personal Injury TAC.
- (e) The Blitz Personal Injury Trustee shall consult with the Blitz Personal Injury TAC (i) on the general implementation and administration of the Blitz Personal Injury Trust;

- (ii) on the general implementation and administration of the Blitz Personal Injury TDP; and (iii) on such other matters as may be required under this Agreement and the Blitz Personal Injury TDP.
- (f) Without limiting the generality of the foregoing section 2.2(e), the Blitz Personal Injury Trustee shall be required to consult with the Blitz Personal Injury TAC pursuant to the Consultation Process set forth in section 6.1 below:
 - to require that Claimants provide additional kinds of medical or other evidence pursuant to the Blitz Personal Injury TDP or the TDP Scoring System;
 - (ii) to change the form of release to be provided pursuant to section 6.6 of the Blitz Personal Injury TDP, provided that the changed release must be consistent with the requirements of the Plan and the Plan Documents and the Settlement Term Sheet;
 - (iii) to terminate the Blitz Personal Injury Trust;
 - (iv) to settle rights assigned to the Blitz Personal Injury Trust;
 - (v) to change the compensation of the Blitz Personal Injury Trustee, other than
 to reflect cost-of-living increases or changes approved by the Bankruptcy
 Court as otherwise provided herein;
 - (vi) to take structural or other actions to minimize any tax on the Blitz Personal Injury Trust Assets;
 - (vii) to adopt Blitz Personal Injury Trust Bylaws in accordance with section 2.2(a) above or to amend the Blitz Personal Injury Trust Bylaws in accordance with the terms thereof;
 - (viii) to amend any provision of this Agreement or the Blitz Personal Injury TDP in accordance with the terms thereof, provided that any such amendments must be consistent with the requirements of the Plan and the Plan Documents; and
 - (ix) to develop methods for auditing the reliability of medical evidence.

- (g) For the avoidance of doubt, as set forth in Section 6.5(c) below, the Blitz Personal Injury Trustee may not amend (i) the TDP Scoring System, (ii) the enumerated criteria listed in paragraph H of the TDP Scoring System for application to the Special Circumstances Fund, or (iii) the amount of the Insurance Settlement Payment allocated to the Non-Appealing Fund and the Special Circumstances Fund.
- (h) The Blitz Personal Injury Trustee shall meet with the Blitz Personal Injury TAC no less often than quarterly. The Blitz Personal Injury Trustee otherwise may meet with the Blitz Personal Injury TAC as and when deemed advisable by the Blitz Personal Injury Trustee.
- (i) The Blitz Personal Injury Trustee, upon notice from the Blitz Personal Injury TAC, if practicable in view of pending business, shall at the next meeting with the Blitz Personal Injury TAC, consider issues submitted by the Blitz Personal Injury TAC for consideration by the Blitz Personal Injury Trust.
- (j) Periodically, but not less often than once a year, the Blitz Personal Injury Trustee shall make available to Covered Claimants and other interested parties the number of claims that have been resolved by the Blitz Personal Injury Trust and the amounts of the awards in each case.

2.3 Claims Administration.

The bulk of the personal injury claims involving Blitz gasoline containers arise from incidents that occurred between July 31, 2007 (the "Release Date") and July 31, 2012. In contrast, only a discrete number of claims arise from incidents that occurred on or before the Release Date. The monies contributed to the Blitz Personal Injury Trust by the Participating Insurers pursuant to the Insurance Settlement represent consideration for the buy back of Participating Insurance Policies in effect for the period from the Release Date to July 31, 2012, the channeling of Blitz Personal Injury Trust Claims covered by the Participating Insurance Policies to the Blitz Personal Injury Trust, and the releases granted to the Participating Insurers under the Plan. The monies contributed by Wal-Mart under the Insurance Settlement represent consideration for, among other things, the channeling of all Blitz Personal Injury Claims to the Blitz Personal Injury Trust and the release being granted to Wal-Mart under the Plan. As set forth in detail below, in the

Blitz Personal Injury TDP and the TDP Scoring System, a portion of the consideration paid under the Insurance Settlement and the BAH Settlement in the sum of \$131,020,000 shall be allocated to the Non-Appealing Fund and \$30,700,000 to the Special Circumstances Fund to pay claims involving Blitz gasoline containers that arise from incidents that occurred between the Release Date and July 31, 2012. The claims that arose prior to the Release Date are not released against the Non-Participating Insurers and are channeled to the Blitz Personal Injury Trust. Holders of Blitz Personal Injury Claims arising prior to the Release Date will be subject the Channeling Injunction and shall retain all rights to pursue their claims against the Assigned Insurance Policies issued by the Non-Participating Insurers.

Blitz Personal Injury Trust Claims shall be administered as follows:

(a) Covered Blitz Personal Injury Claims. The Blitz Personal Injury TDP shall provide mechanisms such as pro rata and/or percentage distributions of the proceeds of the Insurance Settlement and the BAH Settlement allocable to the Blitz Personal Injury Trust, net of reserves for fees, costs, expenses and indemnification obligations incurred by the Blitz Personal Injury Trust, on account of Allowed Covered Blitz Personal Injury Claims and; periodic review of estimates of the numbers and values of Allowed Covered Blitz Personal Injury Claims or other comparable mechanisms, that provide reasonable assurance that the Blitz Personal Injury Trust will value and be in a financial position to pay similar Allowed Covered Blitz Personal Injury Claims in substantially the same manner. From and after the Effective Date, the Blitz Personal Injury Trust shall liquidate and pay the Allowed Covered Blitz Personal Injury Claims in accordance with the Blitz Personal Injury TDP. Notwithstanding the foregoing, the Blitz Personal Injury Trustee shall use his or her best efforts deliver Offer Amounts to the holders of Covered Blitz Personal Injury Claims that are substantially in accordance with the amounts listed on Exhibit B to the Blitz Personal Injury TDP assuming that (i) the Covered Claimants can support the level of injury and other compensation sought in their respective proofs of claim (with the understanding that (a) Covered Claimants who provide documentation supporting a higher recovery than sought in their proof of claim may receive Offer Amounts greater than the amounts listed for such claimant on Exhibit B, which would impact all Covered Claimants' Offer Amounts, and (b) Covered Claimants who provide

documentation supporting a lower recovery than sought in their proof of claim may receive Offer Amounts less than the amounts listed for such claimant on Exhibit B, which would impact all Covered Claimants' Offer Amounts), and (ii) the Covered Claimants can satisfy the Threshold Components set forth in the TDP Scoring System. All distributions the Blitz Personal Injury Trust makes on account of Allowed Covered Blitz Personal Injury Claims shall be final and, other than for demonstrative mathematical errors, shall not subject to recapture or disgorgement by the Blitz Personal Injury Trust or any other party.

- (b) Blitz Personal Injury Claims Arising Prior to the Release Date. As set forth in Article VII of the Plan, all Blitz Personal Injury Claims that arose prior to the Release Date shall be channeled to the Blitz Personal Injury Trust and will be subject to the Channeling Injunction but shall not be deemed to have released their Blitz Personal Injury Claims against the USA Debtors and the Non-Participating Insurers. Holders of such claims shall retain the right to (i) liquidate their claims and seek payment from the Non-Participating Insurers in accordance with the Plan; (ii) assert their claims against, and, if necessary, prosecute an action against the Non-Participating Insurers, and (iii) may initiate an Insurance Action against the Blitz Personal Injury Trust if necessary to preserve their rights against Non-Participating Insurers.
- (c) Vendor Claims, Co-Defendant Claims and Direct Action Claims. Vendor Claims, Co-Defendant Claims and Direct Action Claims shall be channeled to the Blitz Personal Injury Trust and will be subject to the Channeling Injunction. Holders of Vendor Claims, Co-Defendant Claims and Direct Action Claims shall receive the Releases and the benefit of the Channeling Injunction and shall retain their rights, if any, with respect to the Assigned Blitz Insurance Policies but shall not receive any distributions from the Blitz Personal Injury Trust on account of their claims.
- (d) Claims Related to Michael Montgomery. In satisfaction of the claims of Michael Montgomery: (i) the Blitz Personal Injury Trust shall pay to Michael Montgomery the sum of \$3,075,000; (ii) Michael Montgomery shall retain and be entitled to pursue his Claims against the insurance policies issued by Non-Participating Insurers that were in effect on the date that his injuries occurred and against Home Depot; and (iii)

Michael Montgomery shall have no other or further claims against the Blitz Personal Injury Trust or any of the Protected Parties.

- (e) Claims Related to David Calder. In satisfaction of the claim of David Calder and his co-plaintiffs, Debtors, Westchester Fire Insurance Company ("Westchester Fire") and David Calder and his co-plaintiffs agree to settle and compromise in full the claims of David Calder and his co-plaintiffs by (a) Westchester Fire paying \$2,942,014, and (b) the Debtors or the Blitz Personal Injury Trustee causing to be paid, or directing RLI Insurance Company to pay from the proceeds of the Debtors' bond that is returnable to the Blitz USA estate (Bond Number RSB4174412) the full amount of that bond that the Debtors posted for on appeal (\$1,057,986.31). The forgoing payments shall be made within thirty (30) days of payment of the Insurance Settlement Amount but shall be further subject to the exchange of full and mutual releases, compliance with Medicare secondary payer requirements and the dismissal with prejudice of all appeals. The automatic stay of section 362 of the Bankruptcy Code shall remain in place through the payment of the Insurance Settlement Amount. If the Debtors or the Blitz Personal Injury Trustee are unable to deliver the full amount of their bond (\$1,057,986.31) from the proceeds of the Debtors' bond for reasons beyond their control, the Debtors and/or the David Calder and his co-plaintiffs shall be free to go forward with their appeal(s) and otherwise prosecute their claims and, if they opt to do so, the other parties to the settlement of Calder's Claims shall be relieved of their obligations under this paragraph.
- (f) Claims Related to Jonathan Green. Jonathan Green shall retain and shall not release his claim for sanctions, which is now pending appeal, until the occurrence of the Payment Date (as defined in the Insurance Settlement), and the vacator of the sanctions order by the Green court (to which the Debtors and the Green plaintiffs consent and the Participating Blitz Personal Injury Claimants do not oppose), at which time the \$250,000 that has already been deposited with his counsel in escrow, shall be released and paid to Jonathan Green, and any claim asserted by Green shall be released and the parties to the Green case agree to mutually dismiss their appeals.
- (g) Fees and Expenses of Objecting Claimants. \$650,000 shall be paid to reimburse legal fees and expenses incurred in connection with these Chapter 11 Cases

through and including 10:00 a.m. on December 18, 2013, by the Torres claimants, Jones claimants, Perez claimants, Newby claimants, Bauman claimants, Mims claimants and Bosse claimants provided that such claimants either: (i) sign on to the Plan Support Agreement by not later than 10:00 a.m. December 18, 2013; or (ii) withdraw any written opposition to the Insurance Settlement by not later than 10:00 a.m. on December 18, 2013 and do not prosecute objections to approval of the Insurance Settlement, do not file or prosecute objections to the adequacy of the Disclosure Statement, do not file or prosecute objections to confirmation of the Plan and do not vote against or take any other action to oppose confirmation of the Plan. Qualifying claimants shall submit a request for reimbursement of legal fees and costs to the Blitz Personal Injury Trustee within 10 days of the Effective Date. Claims not timely filed shall be deemed waived. The Blitz Personal Injury Trustee in consultation with the Blitz Personal Injury TAC shall review and determine the amount of legal fees and costs allowable for each Qualifying Claimant. In the event that total allowed legal fees and costs exceed \$650,000, the \$650,000 shall be distributed pro-rata amongst allowed legal fees and costs. extent that allowed legal fees and costs are less than \$650,000, allowed legal fees and costs shall be paid in full, and the remaining balance shall be added to the Non-Appealing Fund and shall be distributed to Covered Claimants in accordance with the procedures for distributions from the Non-Appealing Fund. Allowed legal fees and costs shall be paid within 10 days of allowance on a first in first out basis. For purposes of this provision, "legal fees and expenses" shall include fees billed at the regular hourly rate of bankruptcy counsel for any qualifying claimant. "Legal fees and expenses" shall not include fees incurred by plaintiff's tort counsel, unless such claimant has not retained separate bankruptcy counsel, in which case legal fees incurred by tort counsel only in connection with the Chapter 11 Cases, and not the underlying tort cases, shall be reimbursable under this provision at that attorney's regular hourly rate (or is such attorney does not have a regular hourly rate at the average hourly rate for attorneys of similar experience in the jurisdiction where such attorney's office is located to be determined by the Blitz Personal Injury Trustee in consultation with the Blitz Personal Injury TAC). "Legal fees and expenses" shall include travel expenses, expert fees and deposition transcript costs incurred in connection with the Bankruptcy Cases by

bankruptcy counsel or by tort counsel in the instance where separate bankruptcy counsel has not been retained.

2.4 MSP Claims Reporting.

(a) It is the position of the Proponents that neither Debtors nor any Protected Party will have any reporting obligations in respect of their contributions to the Blitz Personal Injury Trust, or in respect of any payments, settlements, resolutions, awards, or other claim liquidations by the Blitz Personal Injury Trust, under the reporting provisions of section 111 of the Medicare, Medicaid, and SCHIP Extension Act of 2007 (P.L.110-173), or any other similar statute or regulation, and any related rules, regulations, or guidance issued in connection therewith ("MMSEA"). Unless and until there is definitive regulatory, legislative, or judicial authority (as embodied in a final non-appealable decision from the United States Court of Appeals for the Third Circuit or the United States Supreme Court), or a letter from the Secretary of Health and Human Services confirming that the Debtors or any Protected Party have no reporting obligations under MMSEA with respect to any settlements, payments, or other awards made by the Blitz Personal Injury Trust or with respect to contributions made or will make to the Blitz Personal Injury Trust, the Blitz Personal Injury Trust shall, at its sole expense, in connection with the implementation of the Plan, act as a reporting agent for the Debtors and/or the Protected Parties and shall timely submit all reports that would be required to be made by the Debtors and/or the Protected Parties under MMSEA on account of any claims settled, resolved, paid, or otherwise liquidated by the Blitz Personal Injury Trust or with respect to contributions to the Blitz Personal Injury Trust including, but not limited to, reports that would be required if the Debtors and/or any Protected Party were determined to be "applicable plans" for purposes of MMSEA, or any of the Debtors and/or any Protected Party were otherwise found to have MMSEA reporting requirements. The Blitz Personal Injury Trust, in its role as reporting agent for the Debtors and/or any Protected Party shall follow all applicable guidance published by the Centers for Medicare & Medicaid Services of the United States Department of Health and Human Services and/or any other agent or successor Entity charged with responsibility for monitoring, assessing, or receiving reports made under MMSEA (collectively, "CMS") to determine whether or not, and, if so, how, to report to CMS pursuant to MMSEA.

- (b) If the Blitz Personal Injury Trust is required to act as a reporting agent for the Debtors and/or any Protected Party pursuant to the provisions of section 2.4(a) above, the Blitz Personal Injury Trust shall provide a written certification to each of the Debtors and/or any Protected Party within ten (10) days following the end of each calendar quarter, confirming that all reports to CMS required by section 2.4(a) have been submitted in a timely fashion, and identifying (i) any reports that were rejected or otherwise identified as noncompliant by CMS, along with the basis for such rejection or noncompliance, and (ii) any payments to Medicare benefits recipients or Medicare-eligible beneficiaries that the Blitz Personal Injury Trust did not report to CMS.
- (c) With respect to any reports rejected or otherwise identified as noncompliant by CMS, the Blitz Personal Injury Trust shall, upon request by the Debtors and/or any Protected Party, promptly provide copies of the original reports submitted to CMS, as well as any response received from CMS with respect to such reports; *provided, however*, that the Blitz Personal Injury Trust may redact from such copies the names, social security numbers other than the last four digits, health insurance claim numbers, taxpayer identification numbers, employer identification numbers, mailing addresses, telephone numbers, and dates of birth of the injured parties, Claimants, guardians, conservators and/or other personal representatives, as applicable.
- (d) If the Blitz Personal Injury Trust is required to act as a reporting agent for the Debtors and/or any Protected Party pursuant to the provisions of section 2.4(a) above, with respect to each claim of a Medicare benefits recipient or Medicare-eligible beneficiary that was paid by the Blitz Personal Injury Trust and not disclosed to CMS, the Blitz Personal Injury Trust shall, upon request by the Debtors and/or any Protected Party, promptly provide the last four digits of the Claimant's social security number, the year of the Claimant's birth, the Claimant's injury, and any other information that may be necessary in the reasonable judgment of the Debtors and/or any Protected Party to satisfy their obligations, if any, under MMSEA, as well as the basis for the Blitz Personal Injury Trust's failure to report the payment. In the event the Debtors and/or any Protected Party disagrees with the Blitz Personal Injury Trust's decision not to report a claim paid by the Blitz Personal Injury Trust, the Blitz Personal Injury Trust shall promptly report the payment to CMS. All documentation relied upon by the Blitz Personal Injury Trust in

making a determination that a payment did not have to be reported to CMS shall be maintained for a minimum of six years following such determination.

- (e) If the Blitz Personal Injury Trust is required to act as a reporting agent for the Debtors and/or any Protected Party pursuant to the provisions of section 2.4(a) above, the Blitz Personal Injury Trust shall make the reports and provide the certifications required by sections 2.4(a) and (b) above until such time as each of the Debtors and/or any Protected Party all determine, in their reasonable judgment, that they have no further legal obligation under MMSEA or otherwise to report any settlements, resolutions, payments, or liquidation determinations made by the Blitz Personal Injury Trust or contributions to the Blitz Personal Injury Trust. Furthermore, following any permitted cessation of reporting, or if reporting has not previously commenced due to the satisfaction of one or more of the conditions set forth in section 2.4(a) above, and if the Debtors and/or any Protected Party reasonably determine, based on subsequent legislative, administrative, regulatory, or judicial developments, that reporting is required, then the Blitz Personal Injury Trust shall promptly perform its obligations under sections 2.4(a) and (b) above.
- (f) Section 2.4(a) above is intended to be purely prophylactic in nature, and does not imply, and shall not constitute an admission, that the Debtors and/or any Protected Party are in fact "applicable plans" within the meaning of MMSEA, or that they have any legal obligation to report any actions undertaken by the Blitz Personal Injury Trust or contributions to the Blitz Personal Injury Trust under MMSEA or any other statute or regulation.
- (g) In the event that CMS concludes that reporting done by the Blitz Personal Injury Trust in accordance with section 2.4(a) above is or may be deficient in any way, and has not been corrected to the satisfaction of CMS in a timely manner, or if CMS communicates to the Blitz Personal Injury Trust, the Debtors and/or any Protected Party a concern with respect to the sufficiency or timeliness of such reporting, or there appears to the Debtors and/or any Protected Party a reasonable basis for a concern with respect to the sufficiency or timeliness of such reporting or non-reporting based upon the information received pursuant to section 2.4(b), (c) or (d) or other credible information, then each of the Debtors and/or any Protected Party shall have the right to submit its own reports to

CMS under MMSEA, and the Blitz Personal Injury Trust shall provide to any party that elects to file its own reports such information as the electing party may require in order to comply with MMSEA, including, without limitation, the full reports filed by the Blitz Personal Injury Trust pursuant to section 2.4(a) above without any redactions. The Debtors and/or any Protected Party shall keep any information they receive from the Blitz Personal Injury Trust pursuant to this section 2.4(g) confidential and shall not use such information for any purpose other than meeting obligations under MMSEA.

(h) Notwithstanding any other provisions hereof, if the Blitz Personal Injury Trust is required to act as a reporting agent for the Debtors and/or any Protected Party, then such Entities shall take all steps necessary and appropriate as required by CMS to permit any reports contemplated by this section to be filed. Furthermore, until the Debtors and/or any Protected Party provide the Blitz Personal Injury Trust with any necessary information that may be provided by CMS's Coordination of Benefits Contractor (the "COBC") to effectuate reporting, the Blitz Personal Injury Trust shall have no obligation to report under section 2.4(a) above with respect to any such entity that has not provided such information.

2.5 Payment of MSP Obligations.

In connection with the implementation of the Plan, the Blitz Personal Injury Trustee shall obtain to the satisfaction of the Blitz Personal Injury Trustee and the Garretson Group, prior to remittance of funds to Covered Claimants' counsel or the Covered Claimant, if pro se, in respect of any Blitz Personal Injury Trust Claim a certification from the Covered Claimant to be paid that said Covered Claimant has provided for the payment and/or resolution of any obligations owing or potentially owing under 42 U.S.C. § 1395y(b), or any related rules, regulations, or guidance, in connection with, or relating to, such Blitz Personal Injury Trust Claim. The Blitz Personal Injury Trust shall provide a quarterly certification of its compliance with this section to each of the Debtors and each Protected Party, and permit reasonable audits by such entities, no more often than quarterly, to confirm the Blitz Personal Injury Trust's compliance with this section.

For the avoidance of doubt, the Blitz Personal Injury Trust shall be obligated to comply with the requirements of this section regardless of whether the Debtors and/or any Protected Party elects to file its own reports under MMSEA pursuant to section 2.4(g) above.

2.6 Retention of Qualified MSP Compliance Vendor.

In accordance with section 4.12 of the Plan, the Blitz Personal Injury Trust shall, at its expense, retain the Garretson Groupto provide such services as are required to ensure compliance with the Blitz Personal Injury Trust's MSP requirements. The Participating Insurers and Wal-Mart shall have the right to obtain information from the Garretson Group, the Blitz Personal Injury Trust and any holder of a Covered Blitz Personal Injury Claim as they may reasonably request to ensure that the Blitz Personal Injury Trust has complied with MSP requirements or for the purpose of internal audits and insurance or reinsurance claims. The Blitz Personal Injury Trust is prohibited from making a distribution to any holder of a Blitz Personal Injury Claim who refuses to provide the information necessary to meet MSP requirements with regard to that claimant. If a Covered Claimant fails to provide the information necessary to meet MSP requirements by the date by which the Blitz Personal Injury Trustee intends to terminate the Blitz Personal Injury Trust in accordance with section 6.4 of this Agreement, any amount reserved for such Covered Claimant shall be treated as a residual fund to be redistributed to other Covered Claimants pursuant to either paragraph I of the TDP Scoring System (if such funds were reserved from the Special Circumstances Fund for the benefit of a Covered Claimant with a claim against the Special Circumstances Fund) or paragraph K of the TDP Scoring System (if such funds were reserved from the Non-Appealing Fund for the benefit of a Covered Claimant who does not have a claim against the Special Circumstances Fund).

2.7 Indemnification for Medicare Claims Reporting and Payment Obligations.

For the avoidance of doubt, the Blitz Personal Injury Trust shall defend, indemnify save and hold harmless each Protected Party from any claims in respect of Medicare claims reporting and payment obligations in connection with Blitz Personal Injury Trust Claims, including any obligations owing or potentially owing under MMSEA or 42 U.S.C. § 1395y(b) or any related rules, regulations, or guidance issued in connection therewith, or relating thereto, and any claims arising from or related to the Blitz Personal Injury Trust's obligations under sections 2.4 and 2.5 above. All defense and indemnity costs described in this section shall be assessed pro rata

against the Non-Appealing Fund and the Special Circumstances Fund as set forth in the TDP Scoring System provided, however, that in the event that either the Non-Appealing Fund of the Special Circumstances Fund is unable to satisfy its pro rata obligations to the Protected Parties, such obligations shall be satisfied in full by the other Fund.

SECTION 3

ACCOUNTS, INVESTMENTS, AND PAYMENTS

3.1 Accounts. The Blitz Personal Injury Trustee may, from time to time, create such accounts and reserves as he or she may deem necessary, prudent, or useful to (i) provide for the payment, or to make provision for future payment, of expenses incurred or reasonably anticipated to be incurred by the Blitz Personal Injury Trust (the "Cost Reserve") and (ii) provide for the payment, or to make provision for future payment, on account of liquidated Blitz Personal Injury Trust Claims; and may, with respect to any such account or reserve, restrict the use of monies therein; provided, however, that such reserve be consistent with the allocation of expenses between the Non-Appealing Fund and the Special Circumstances Fund as set forth in the TDP Scoring System. With respect to the reserves contemplated under subpart (ii) above, reserves will be set consistent with Section 6.5 of the Blitz Personal Injury Trust Distribution Procedures. Because there are limited funds available in the Blitz Personal Injury Trust for the payment of the Blitz Personal Injury Claims, no Covered Claimant is entitled to payment "in full," payment that is disproportionately larger than similarly situated Covered Claimants with the same injuries and similar characteristics, or payment that is at the expense of other Covered Claimants. The amount of the initial Cost Reserve shall be \$1,000,000.00, deducted pro rata from the Non-Appealing Fund and the Special Circumstances Fund (i.e., Because the \$30,700,000 allocated to the Special Circumstances Fund represents 18.98% of \$161,720,000, \$189,800 of the \$1,000,000 reserve shall come from the Special Circumstances Fund and \$810,200 shall come from the Non-Appealing Fund). After consultation with the Blitz Personal Injury TAC, the Blitz Personal Injury Trustee may increase the size of the Cost Reserve in an amount necessary to provide for expenses incurred or reasonably anticipated to be incurred by the Blitz Personal Injury Trustee. The Blitz Personal Injury Trustee shall include a reasonably detailed description of any account or reserve created in accordance with this section 3.1 in the annual reports described in section 2.2(c)(i) hereof, which description shall include, with respect

to any such account, the transfers made to such account, the proceeds of or earnings on the assets held in each such account, and the payments or disbursements made from each such account.

- **3.2** <u>Investments.</u> Investment of monies held in the Blitz Personal Injury Trust shall be administered in a manner consistent with the standards set forth in the Uniform Prudent Investor Act, subject to the following limitations and provisions:
 - (a) The Blitz Personal Injury Trust may invest in well diversified equity portfolios whose benchmark is a broad equity market index such as, but not limited to, the S&P 500 Index, Russell 1000 Index, S&P ADR Index or MSCI EAFE Index. The Blitz Personal Injury Trust shall not acquire, directly or indirectly, equity in any entity or business enterprise if, immediately following such acquisition, the Blitz Personal Injury Trust would hold more than 5% of the equity in such entity or business enterprise. The Blitz Personal Injury Trust shall not hold, directly or indirectly, more than 10% of the equity in any entity or business enterprise.
 - (b) The Blitz Personal Injury Trust shall not acquire or hold any long-term debt securities unless (i) such securities are included in the Blitz Personal Injury Trust Assets under the Plan, (ii) such securities are rated "Baa" or higher by Moody's, "BBB" or higher by Standard & Poor's ("S&P's"), or have been given an equivalent investment grade rating by another nationally recognized statistical rating agency, or (iii) have been issued or fully guaranteed as to principal and interest by the United States of America or any agency or instrumentality thereof. This restriction does not apply to any pooled investment vehicles where pooled assets receive an investment grade rating (i.e., "BBB" rating or above) by a nationally recognized rating agency.
 - (c) Notwithstanding (b) above, the Blitz Personal Injury Trust may acquire or hold additional non-investment grade debt securities for longer than ninety (90) days if the Blitz Personal Injury Trust holds these securities as part of the Blitz Personal Injury Trust's intermediate-term bond manager portfolio, and such securities represent no more than 5% of the bond manager's portfolio.
 - (d) The Blitz Personal Injury Trust shall not acquire or hold for longer than ninety (90) days any commercial paper unless such commercial paper is rated "Prime-1" or

higher by Moody's or "A-1" or higher by S&P's or has been given an equivalent rating by another nationally recognized statistical rating agency.

- (e) The Blitz Personal Injury Trust shall not acquire any debt securities or other instruments issued by any entity if, following such acquisition, the aggregate market value of all debt securities and instruments issued by such entity held by the Blitz Personal Injury Trust would exceed 5% of the aggregate value of the Blitz Personal Injury Trust Assets. There is no limitation on holding debt securities or other instruments issued or fully guaranteed as to principal and interest by the United States of America or any agency or instrumentality thereof.
- (f) The Blitz Personal Injury Trust shall not acquire or hold any certificates of deposit unless all publicly held, long-term debt securities, if any, of the financial institution issuing the certificate of deposit and the holding company, if any, of which such financial institution is a subsidiary, meet the standards set forth in Section 3.2(b) above.
- (g) The Blitz Personal Injury Trust may acquire and hold any securities or instruments obtained as proceeds of litigation or otherwise to resolve disputes, without regard to the limitations set forth in Subsections (a)-(f) above.
- (h) The Blitz Personal Injury Trust shall not acquire or hold any repurchase obligations unless, in the opinion of the Blitz Personal Injury Trustee, they are adequately collateralized based on the advice and recommendation of its investment managers, the Blitz Personal Injury Trust may allow its investment managers to acquire prudently or hold derivative instruments like options and futures in the normal course of portfolio management. Specifically, the Blitz Personal Injury Trust may acquire or hold derivatives to manage or mitigate portfolio risk, including, but not limited to, interest rate risk and equity market risk. Using derivative instruments to leverage a portfolio to enhance returns (at a much greater risk to the portfolio) is prohibited.
- (i) The Blitz Personal Injury Trust may lend securities on a short-term basis, subject to adequate, normal and customary collateral arrangements.

3.3 Source of Payments. All Blitz Personal Injury Trust Expenses and all other liabilities of the Blitz Personal Injury Trust shall be payable solely by the Blitz Personal Injury Trustee in a manner consistent with the allocation of costs and expenses pro rata between the Non-Appealing Fund and the Special Circumstances Fund as set forth in the TDP Scoring System. Neither any Protected Party or their respective subsidiaries, successors in interest, or the present or former shareholders, directors, officers, employees or agents, or their respective subsidiaries, nor the Blitz Personal Injury Trustee in his or her personal capacity, the Blitz Personal Injury TAC, or any of their respective officers, agents, advisors, or employees shall be liable for the payment of Blitz Personal Injury Trust Expenses or any other liability of the Blitz Personal Injury Trust. The Blitz Personal Injury Trustee shall include a reasonably detailed description of Blitz Personal Injury Trust Expenses and other liabilities paid in accordance with this section 3.3 in the annual reports described in section 2.2(c)(i) above.

SECTION 4

BLITZ PERSONAL INJURY TRUSTEE

4.1 <u>Number.</u> There shall be one (1) Blitz Personal Injury Trustee. The initial Blitz Personal Injury Trustee shall be the person named on the signature page hereof.

4.2 Term of Service.

- (a) The initial Blitz Personal Injury Trustee named pursuant to section 4.1 above shall serve an initial term of three (3) years. Thereafter each term of service shall be five (5) years. The initial Blitz Personal Injury Trustee shall serve from the Effective Date until the earlier of (i) the end of his or her term, (ii) his or her death, (iii) his or her resignation pursuant to section 4.2(b) below, (iv) his or her removal pursuant to section 4.2(c) below, or (v) the termination of the Blitz Personal Injury Trust pursuant to section 7.2 below.
- (b) A Blitz Personal Injury Trustee may resign at any time by written notice to the Blitz Personal Injury TAC. Such notice shall specify a date when such resignation shall take place, which shall not be less than 90 days after the date such notice is given, where practicable.
- (c) A Blitz Personal Injury Trustee may be removed by order of the Bankruptcy Court upon notice and motion filed by the Blitz Personal Injury TAC in the event that the

Blitz Personal Injury Trustee becomes unable to discharge his or her duties hereunder due to accident or physical or mental deterioration, or for other good cause. Good cause shall be deemed to include, without limitation (i) substantial failure to comply with the general administration provisions of section 2.2 above, (ii) a consistent pattern of neglect and failure to perform or participate in performing the duties of the Blitz Personal Injury Trustee hereunder, (iii) repeated nonattendance at scheduled meetings, or (iv) one of the circumstances set forth in section 4.7 of this Agreement. Removal shall take effect at such time as the Bankruptcy Court shall determine.

4.3 Appointment of Successor Blitz Personal Injury Trustee.

- (a) In the event of a vacancy in the position of a Blitz Personal Injury Trustee, the Blitz Personal Injury TAC shall consult concerning appointment of a successor (a "Successor Blitz Personal Injury Trustee"). The vacancy shall be filled by the vote of a majority of the Blitz Personal Injury TAC members. In the event that the Blitz Personal Injury TAC members fail to secure a majority vote for the appointment of a Successor Blitz Personal Injury Trustee, the Bankruptcy Court shall make the appointment. Nothing shall prevent the reappointment of a Blitz Personal Injury Trustee for an additional term or terms except that a Blitz Personal Injury Trustee removed for cause shall not be reappointed.
- (b) Immediately upon the appointment of any Successor Blitz Personal Injury Trustee, all rights, titles, duties, powers and authority of the predecessor Blitz Personal Injury Trustee hereunder shall be vested in, and undertaken by, the Successor Blitz Personal Injury Trustee without any further act. No Successor Blitz Personal Injury Trustee shall be liable personally for any act or omission of his or her predecessor Blitz Personal Injury Trustee.
- (c) Each Successor Blitz Personal Injury Trustee shall serve until the earlier of (i) the end of a full term of five (5) years if the predecessor Blitz Personal Injury Trustee completed his or her term, (ii) the end of the remainder of the term of the Blitz Personal Injury Trustee whom he or she is replacing if said predecessor Blitz Personal Injury Trustee did not complete said term, (iii) his or her death, (iv) his or her resignation

pursuant to section 4.2(b) above, (v) his or her removal pursuant to section 4.2(c) above, or (vi) the termination of the Blitz Personal Injury Trust pursuant to section 7.2 below.

4.4 Liability of Blitz Personal Injury Trustee, Blitz Personal Injury TAC.

Neither the Blitz Personal Injury Trustee nor the members of the Blitz Personal Injury TAC shall have any liability to the Blitz Personal Injury Trust, to any Blitz Personal Injury Trust Claimant, or to any other Entity, for actions taken or not taken in connection with the operation of the Blitz Personal Injury Trust or the administration, processing, settlement, resolution, liquidation, satisfaction and/or payment of Blitz Personal Injury Trust Claims under the Blitz Personal Injury TDP except for a breach of fiduciary duty by any of the foregoing committed through fraud, gross negligence or willful misconduct.

4.5 Compensation and Expenses of Blitz Personal Injury Trustee.

- (a) For all time expended (i) administering the Blitz Personal Injury Trust in accordance with the terms of this Agreement and the Plan, (ii) preparing for and attending Blitz Personal Injury Trust meetings, and (iii) receiving, processing, administering, resolving, liquidating and/or paying Blitz Personal Injury Trust Claims, the Blitz Personal Injury Trustee shall receive the sum of \$600.00 per hour, plus reasonable out-of-pocket costs and expenses. The Blitz Personal Injury Trustee shall record all hourly time to be charged to the Blitz Personal Injury Trust on a daily basis. If any payments received from by the Blitz Personal Injury Trustee are determined by the Blitz Personal Injury Trustee to be properly allocable to a particular Covered Claimant as an Individual Cost (each, a "Reimbursable Payment"), then, at the time the Individual Cost that qualifies as a Reimbursable Payment would have otherwise been deducted from the distribution to be made to such Covered Claimant, an amount equal to such Reimbursable Payment shall be transferred to the Cost Reserve from the Non-Appealing Fund or the Special Circumstances Fund (as applicable, depending on whether the Covered Claimant at issue is to be paid from the Non-Appealing Fund or the Special Circumstances Fund).
- (b) On a monthly basis, by the 15th day of each month for the preceding month, the Blitz Personal Injury Trustee shall submit bills for compensation for such preceding month to the Blitz Personal Injury TAC and to any counsel for a Covered Claimant that

has requested copies of such bills. If no objection is made in writing to such bill within seven (7) days of submission, then the Blitz Personal Injury Trustee shall be entitled to pay such bill from the Retainer or, if the Retainer has been exhausted, from the Blitz Personal Injury Trust Assets. If a written objection is timely submitted to the Blitz Personal Injury Trustee, the objecting party shall discuss its objection in good faith with the Blitz Personal Injury Trustee in an effort to reach a consensual resolution. If no resolution is reached within seven (7) days of submission of a written objection then the Blitz Personal Injury Trustee shall be entitled to payment of the monthly bill to which such objection was submitted unless a written objection is filed with the Bankruptcy Court seeking a determination of the matter. An objection to a monthly bill filed with the Bankruptcy Court and notice of any hearing scheduled on such matter must be served on the Blitz Personal Injury Trust and the members of the Blitz Personal Injury TAC. If a written objection is filed, payment of the bill to which the objection relates shall be made only pursuant to (i) an order of the Bankruptcy Court or (ii) agreement of the Blitz Personal Injury Trustee and the party filing the objection, which agreement also would result in a withdrawal of the objection filed with the Bankruptcy Court.

- (c) The Blitz Personal Injury Trust will promptly reimburse the Blitz Personal Injury Trustee for all reasonable out-of-pocket costs and expenses incurred by the Blitz Personal Injury Trustee in connection with the performance of his or her duties hereunder, provided however, that the Blitz Personal Injury TAC may make such motion to the Bankruptcy Court as it deems advisable to seek disgorgement of any cost or expense for which the Blitz Personal Injury Trustee has received reimbursement that the Blitz Personal Injury TAC believes was not reasonable under the circumstances. The costs and expenses incurred by the Blitz Personal Injury Trustee shall be apportioned between the Non-Appealing Fund and the Special Circumstances Fund as set forth in the TDP Scoring System.
- (d) The hourly compensation payable to the Blitz Personal Injury Trustee hereunder shall be reviewed every year by the Blitz Personal Injury Trustee and, after consultation with the members of the Blitz Personal Injury TAC, appropriately adjusted as may be reasonable under the circumstances including, without limitation, for changes in the cost of living.

- (e) The Blitz Personal Injury Trust shall include a description of the amounts paid under this section 4.5 in the accounts to be filed with the Bankruptcy Court and provided to the Blitz Personal Injury TAC pursuant to section 2.2(c)(i) above.
- 4.6 Blitz Personal Injury Trustee's Employment of Experts. The Blitz Personal Injury Trustee may, but shall not be required to, retain and/or consult with legal counsel, accountants, auditors, experts, or financial and investment advisors on the matters submitted to the Blitz Personal Injury Trustee (a "Trust Professional"). In the absence of gross negligence, the written opinion of, or information provided by, any such Trust Professional on the particular matter in respect of which such Trust Professional is an expert may be relied upon by the Blitz Personal Injury Trustee and shall be full and complete authorization and protection to the Blitz Personal Injury Trustee in respect of any action taken or not taken in good faith by the Blitz Personal Injury Trustee otherwise consistent with this Agreement and in accordance with the written opinion of or information provided by such Trust Professional. The reasonable and necessary fees for any Trust Professional(s) will be paid from the Blitz Personal Injury Trust Assets and allocated, as permitted under the TDP Scoring System, to one or more Covered Claimants as an Individual Cost (as defined and using the methodology described in the TDP Scoring System) to be charged to a particular Covered Claimant or as a General Cost (as defined and using the methodology described in the TDP Scoring System) to
- 4.7 <u>Blitz Personal Injury Trustee's Independence.</u> The Blitz Personal Injury Trustee shall not, during the term of his or her service, hold a financial interest in, act as attorney or agent for, or serve as any other professional for any Entity with a financial interest in the operation of the Blitz Personal Injury Trust. No Blitz Personal Injury Trustee shall act as an attorney for any Claimant either (i) in connection with such Claimant's Blitz Personal Injury Trust Claim, or (ii) otherwise prior to final payment on account of such Blitz Personal Injury Trust Claim, and (iii) the Blitz Personal Injury Trustee shall not make any agreement with the holder of any Blitz Personal Injury Trust Claim or such Claimant's representative prior to final payment on account of such Blitz Personal Injury Trust Claim applicable after such final payment. In addition to the circumstances set forth in section 4.2(c) of this Agreement, any violation of this section 4.7 shall be cause for removal of the Blitz Personal Injury Trustee.

4.8 Bond. The Blitz Personal Injury Trustee shall not be required to post any bond or other form of surety or security unless otherwise ordered by the Bankruptcy Court.

SECTION 5

TRUST ADVISORY COMMITTEE

- 5.1 Formulation and Members. The Blitz Personal Injury TAC shall be formed pursuant to the Plan as of the Effective Date. The Blitz Personal Injury TAC shall consist of four (4) members, who shall initially be the persons named on the signature page hereof. Because there will be an even number of members of the Blitz Personal Injury TAC, in the event a vote of the Blitz Personal Injury TAC results in a tie, the Blitz Personal Injury Trustee shall hold a tiebreaker vote. In the event a member of the Blitz Personal Injury TAC is unable to attend an in-person or telephonic meeting of the Blitz Personal Injury TAC, such member may designate an alternate to attend such meeting on his or her behalf with such alternate having the same rights to be heard and vote on all matters discussed.
- 5.2 <u>Duties.</u> The members of the Blitz Personal Injury TAC shall serve in a fiduciary capacity representing all Blitz Personal Injury Claimants and Protected Parties. The Blitz Personal Injury Trustee must consult with the Blitz Personal Injury TAC on matters identified in section 2.2(e) above and may consult with the Blitz Personal Injury TAC with respect to such other matters relating to the Blitz Personal Injury Trust and the administration, processing, settlement, resolution, liquidation, satisfaction and/or payment, as applicable, of Blitz Personal Injury Trust Claims as the Blitz Personal Injury Trustee and the Blitz Personal Injury TAC deem advisable.

5.3 Term of Office.

(a) The initial members of the Blitz Personal Injury TAC appointed in accordance with section 5.1 above shall serve the staggered three-, four-, or five-year terms shown on the signature pages hereof. Thereafter, each term of office shall be five (5) years. Each member of the Blitz Personal Injury TAC shall serve until the earlier of (i) his or her death, (ii) his or her resignation pursuant to section 5.3(b) below, (iii) his or her removal pursuant to section 5.3(c) below, (iv) the end of his or her term as provided above, or (v) the termination of the Blitz Personal Injury Trust pursuant to section 7.2 below.

- (b) A member of the Blitz Personal Injury TAC may resign at any time by written notice to the other members of the Blitz Personal Injury TAC and the Blitz Personal Injury Trustee. Such notice shall specify a date when such resignation shall take effect, which shall not be less than ninety (90) days after the date such notice is given, where practicable.
- (c) A member of the Blitz Personal Injury TAC may be removed in the event that he or she becomes unable to discharge his or her duties hereunder due to accident, physical deterioration, mental incompetence, or a consistent pattern of neglect and failure to perform or to participate in performing the duties of such member hereunder, such as repeated nonattendance at scheduled meetings, or for other good cause. Such removal shall be made at the recommendation of the remaining members of the Blitz Personal Injury TAC with the approval of the Bankruptcy Court.

5.4 Appointment of Successor Members.

- (a) A vacancy in the Blitz Personal Injury TAC caused by resignation, death or as a result of removal, shall be filled with an individual, not a firm, approved by the majority vote of the Blitz Personal Injury Trustee and all remaining members of the Blitz Personal Injury TAC. Nothing in this Agreement shall prevent the reappointment of an individual serving as a member of the Blitz Personal Injury TAC for an additional term or terms, except that a member removed for cause may not be reappointed, and there shall be no limitation on the number of terms that a Blitz Personal Injury TAC member may serve.
- (b) Each successor member of the Blitz Personal Injury TAC shall serve until the earlier of (i) the end of the full term of five (5) years for which he or she was appointed if his or her immediate predecessor member of the Blitz Personal Injury TAC completed his or her term, (ii) the end of the term of the member of the Blitz Personal Injury TAC whom he or she replaced if his or her predecessor member did not complete such term, (iii) his or her death, (iv) his or her resignation pursuant to section 5.3(b) above, (v) his or her removal pursuant to section 5.3(c) above, or (vi) the termination of the Blitz Personal Injury Trust pursuant to section 7.2 below.
- 5.5 <u>Blitz Personal Injury TAC's Employment of Professionals.</u> The Blitz Personal Injury TAC may at the sole and non-reimbursable cost of its members (but is not required to)

retain and/or consult counsel, accountants, appraisers, auditors, forecasters, experts, and financial and investment advisors, and such other parties deemed by the Blitz Personal Injury TAC to be qualified as experts on matters submitted to the TAC (the "TAC Professionals"). Subject to the provisions of section 6.1 of this Agreement, the Blitz Personal Injury TAC and the TAC Professionals shall at all times have complete access to the Blitz Personal Injury Trust's officers, employees and agents, as well as to the Trust Professionals, and shall also have complete access to all information generated by them or otherwise available to the Blitz Personal Injury Trust or the Blitz Personal Injury Trustee other than the Blitz Personal Injury Privileged Information (as defined in the Plan), provided that any information provided by the Trust Professionals shall not constitute a waiver of any applicable privilege, and provided further that the foregoing provisions shall not amend, modify, or alter in any fashion the provisions regarding and/or restrictions applicable to Blitz Personal Injury Privileged Information or Blitz Personal Injury Confidential Information (each as defined in the Plan), the terms of which apply fully hereunder. In the absence of gross negligence, the written opinion of or information provided by any such TAC Professional on the particular matter in respect of which such TAC Professional is an expert may be relied upon by the Blitz Personal Injury TAC and shall be full and complete authorization and protection to the Blitz Personal Injury TAC in respect of any action taken or not taken in good faith by the Blitz Personal Injury TAC otherwise consistent with this Agreement and in accordance with the written opinion of or information provided by such TAC Professional.

5.6 Compensation and Expenses of Blitz Personal Injury TAC. The members of the Blitz Personal Injury TAC shall receive compensation from the Blitz Personal Injury Trust from the Blitz Personal Injury Trust Assets for their services as members of the Blitz Personal Injury TAC in the form of an hourly rate of no more than \$350 per hour for preparation for and attendance at meetings or other conduct of the Blitz Personal Injury Trust as may be specifically requested by the Blitz Personal Injury Trustee and for which the Blitz Personal Injury Trustee agrees in advance to provide reasonable compensation. The members of the Blitz Personal Injury TAC shall also be reimbursed promptly for all reasonable out-of-pocket costs and expenses incurred by the members of the Blitz Personal Injury TAC in connection with their attendance at meetings and as may be agreed upon by the Blitz Personal Injury Trustee. Any such reimbursement or direct payment shall be deemed a Blitz Personal Injury Trust Expense and allocated, as permitted under the TDP Scoring System, to one or more Covered Claimants as

an Individual Cost (as defined and using the methodology described in the TDP Scoring System) to be charged to a particular Covered Claimant or as a General Cost (as defined and using the methodology described in the TDP Scoring System) to be paid from the Cost Reserve. The Blitz Personal Injury Trust shall include a description of the fees, costs, and expenses incurred under this section 5.6 in the accounts to be filed with the Bankruptcy Court and provided to the Blitz Personal Injury Trustee and the Blitz Personal Injury TAC, pursuant to section 2.2(c)(i) above.

SECTION 6

GENERAL PROVISIONS

6.1 Procedures for Consultation with Blitz Personal Injury TAC. In the event the Blitz Personal Injury Trustee is required to consult with the Blitz Personal Injury TAC as provided herein, the Blitz Personal Injury Trustee shall provide the Blitz Personal Injury TAC with written advance notice of the matter under consideration, and with all relevant information concerning the matter as is reasonably practicable under the circumstances. The Blitz Personal Injury Trustee shall also provide the Blitz Personal Injury TAC with such reasonable access to Trust Professionals and other experts retained by the Blitz Personal Injury Trust and its staff (if any) as the Blitz Personal Injury TAC may reasonably request during the time that the Blitz Personal Injury Trustee is considering such matter, and shall also provide Blitz Personal Injury TAC the opportunity, at reasonable times and for reasonable periods of time, to discuss and comment on such matter with the Blitz Personal Injury Trustee; provided that in no event shall the Blitz Personal Injury TAC or its members (A) have any role, whether by consent, consultation or otherwise, in the Blitz Personal Injury Trust's selection of counsel, experts or other professionals to defend Blitz Personal Injury Trust Claims against the Blitz Personal Injury Trust, or (B) have any right to consult with or obtain information from the Blitz Personal Injury Trust or anyone employed by the Blitz Personal Injury Trust concerning the defense of any such Blitz Personal Injury Trust Claims.

In determining when to take definitive action on any matter subject to the consultation process set forth in this section 6.1, the Blitz Personal Injury Trustee shall take into consideration the time required for the Blitz Personal Injury TAC, if its members so wish, to engage and consult with its own independent financial or investment advisors as to such matter. In any event, the Blitz Personal Injury Trustee shall not take definitive action on any such matter

until at least thirty (30) days after providing the Blitz Personal Injury TAC with the initial written notice that such matter is under consideration by the Blitz Personal Injury Trustee, unless such time period is waived by the Blitz Personal Injury TAC.

6.2 Indemnification.

- (a) The Blitz Personal Injury Trust shall indemnify, hold harmless and defend the Blitz Personal Injury Trustee, and the members of the Blitz Personal Injury TAC in the performance of their respective duties hereunder to the fullest extent that a corporation or trust organized under the laws of the State of Delaware is entitled to indemnify and defend such persons against any and all liabilities, expenses, claims, damages or losses incurred by them in the performance of their duties hereunder. Notwithstanding the foregoing, no individual shall be indemnified or defended in any way for any liability, expense, claim, damage, or loss for which he or she is ultimately held liable as a result of such individual's own breach of fiduciary duty committed through fraud, gross negligence or willful misconduct. As set forth in the TDP Scoring System, the foregoing indemnification costs shall be borne pro rata between the Non-Appealing Fund and the Special Circumstances Fund in the manner described therein.
- (b) The Blitz Personal Injury Trust shall fully and completely defend each of the Debtors, the present and former directors and officers of each of the Debtors, the Participating Insurers, Wal-Mart, the BAH Released Parties, shareholders of the Debtors solely in their capacity as shareholders and solely related to the Blitz Personal Injury Claims, and the Representatives, Affiliates, subsidiaries and/or successors of each of the foregoing in connection with any proceeding involving, relating to or arising out of, in whole or in part, the enforcement or enforceability of the Channeling Injunction. In the event any person or Entity asserts any claim that is subject to the Channeling Injunction, the Blitz Personal Injury Trust shall, at its own expense, defend the validity of the Channeling Injunction and its application and use its best efforts to establish that such Claim is enjoined. The Participating Insurers, Wal-Mart and the BAH Settling Parties (i) shall have consultation and approval rights with respect to the selection of counsel hired by the Blitz Personal Injury Trust for such defense obligations and (ii) shall have the right, at their own cost and expense, to associate in the defense in any proceeding

concerning the enforcement and application of the Channeling Injunction. If the Blitz Personal Injury Trust breaches its duty to fully and completely defend the Protected Parties, the Blitz Personal Injury Trust is obligated to indemnify the Protected Parties, including advancement of defense costs. As set forth in the TDP Scoring System, the foregoing indemnification costs shall be borne pro rata between the Non-Appealing Fund and the Special Circumstances Fund in the manner described therein provided, however, that in the event that either the Non-Appealing Fund of the Special Circumstances Fund is unable to satisfy its pro rata obligations to the Protected Parties, such obligations shall be satisfied in full by the other fund.

- (d) The Blitz Personal Injury Trust may but is not required to (after consulting with the Blitz Personal Injury TAC) purchase and maintain reasonable amounts and types of insurance on behalf of an Indemnified Party to provide for payment of the obligations of the Blitz Personal Injury Trust under this section 6.2 of this Agreement. As set forth in the TDP Scoring System, the costs of any such insurance shall be borne pro rata between the Non-Appealing Fund and the Special Circumstances Fund in the manner described therein.
- **6.3** <u>Irrevocability.</u> The Blitz Personal Injury Trust is irrevocable.

6.4 Termination.

- (a) The Blitz Personal Injury Trust shall automatically terminate on the date that is ninety (90) days after the first to occur of the following events (the "Termination Date"):
 - (i) the Blitz Personal Injury Trustee decides to terminate the Blitz Personal Injury Trust because the Blitz Personal Injury Claims duly filed with the Blitz Personal Injury Trust have been liquidated and paid to the extent provided in this Agreement and the Blitz Personal Injury TDP or disallowed; or
 - (ii) if the Blitz Personal Injury Trustee has procured (after consulting with the Blitz Personal Injury TAC) and has in place irrevocable insurance policies and has established claims handling agreements and other necessary arrangements with suitable third parties adequate to discharge all expected

remaining obligations and expenses of the Blitz Personal Injury Trust in a manner consistent with this Agreement and the Blitz Personal Injury TDP, the date on which the Bankruptcy Court enters an order approving such insurance and other arrangements and such order becomes a Final Order.

(b) On the Blitz Personal Injury Trust Termination Date, after the payment of all the Blitz Personal Injury Claims have been provided for and the liquidation of all properties and other non-cash trust assets then held by the Blitz Personal Injury Trust, all monies remaining in the Non-Appealing Fund (including any funds originating from the Non-Appealing Fund held in the Cost Reserve described in paragraph 3.1 above) shall be distributed pro rata to holders of Allowed Covered Blitz Personal Injury Claims that accepted their Offer Amount from the Blitz Personal Injury Trust and received an award from the Non-Appealing Fund in proportion to the amount each Covered Claimant received from the Non-Appealing Fund. Or, if in the judgment of the Blitz Personal Injury Trustee, such sums are determined to be de minimis such that the costs associated with making such a distribution would outweigh the impact of the distribution, then the excess funds may be given to such organization(s), exempt from federal income tax under section 501(c)(3) of the IRC, at the discretion of the Blitz Personal Injury Trustee. The amount of monies allocated to the Special Circumstances Fund shall be completely exhausted by costs and payments to Covered Claimants making application to the Special Circumstances Fund. All remaining monies originating from the Special Circumstances Fund held in the Cost Reserve described in paragraph 3.1 above shall be distributed pro rata among Covered Claimants receiving payment from the Special Circumstances Fund. Notwithstanding any contrary provision of the Plan and related documents, this section 6.4(b) cannot be modified or amended.

6.5 Amendments.

(a) Except as otherwise provided in sections 6.5(b) and 6.5(c) below, the Blitz Personal Injury Trustee, after consultation with the Blitz Personal Injury TAC, may modify or amend this Agreement and the Blitz Personal Injury Trust By-laws, provided that any such amendments must be consistent with the requirements of the Plan. The Blitz Personal Injury Trustee, after consultation with the Blitz Personal Injury TAC, may modify or

amend the Blitz Personal Injury TDP, provided, however, that no amendment to the Blitz Personal Injury TDP shall be inconsistent with the limitations on amendments provided therein, and provided further that any such amendments must be consistent with the requirements of the Plan and that the Blitz Personal Injury Trustee shall give the Blitz Personal Injury TAC prior notice of the proposed amendments. Any modification or amendment made pursuant to this section 6.5 must be done in writing. Notwithstanding anything contained in this Agreement, any Blitz Personal Injury Trust Bylaws, or the Blitz Personal Injury TDP to the contrary, neither this Agreement, the Blitz Personal Injury Trust Bylaws, nor the Blitz Personal Injury TDP shall be modified or amended in any way that could jeopardize, impair, or modify (i) the efficacy or enforceability of the injunctions entered in connection with confirmation of the Plan, or (ii) the status of the Blitz Personal Injury Trust as a qualified settlement fund under section 468B of the IRC.

- (b) Sections 2.4, 2.5 and 2.6 of this Agreement may not be amended without the prior written consent of the Participating Insurers and Wal-Mart.
- (c) There shall be no amendments to (i) the TDP Scoring System, (ii) the enumerated criteria listed in paragraph H of the TDP Scoring System for application to the Special Circumstances Fund, (iii) the amounts allocated to the Non-Appealing Fund and the Special Circumstances Fund, (iv) the mechanisms for allocating costs and expenses to Covered Claimants (including the Cost Reserve mechanism described in section 3.1 above), (v) the mechanism for review and objection to fees and expenses incurred by the Blitz Personal Injury Trustee described in section 4.5 above, or (vi) this section 6.5(c).
- **6.6** Severability. Should any provision in this Agreement be determined to be unenforceable, such determination shall in no way limit or affect the enforceability and operative effect of any and all other provisions of this Agreement.
- **6.7** <u>Notices.</u> Notices to Blitz Personal Injury Claimants asserting Blitz Personal Injury Trust Claims shall be given electronically to the e-mail address of such person in each case as provided on such person's claim form submitted to the Blitz Personal Injury Trust with respect to his or her Blitz Personal Injury Trust Claim.
 - (a) Any notices or other communications required or permitted hereunder to the following parties shall be in writing and delivered at the e-mail addresses designated

below, or mailed by registered or certified mail, return receipt requested, postage prepaid, addressed as follows, or to such other address or addresses as may hereafter be furnished in writing to each of the other parties listed below in compliance with the terms hereof.

To the Blitz Personal Injury Trust: <u>blitzclaim@litigationkc.com</u>; <u>adr@richardcohen.net</u>; dhaltiwanger@rpwb.com

To the Blitz Personal Injury Trustee: <u>adr@richardcohen.net</u>; <u>blitzclaim@litigationkc.com</u>; dhaltiwanger@rpwb.com

To the Blitz Personal Injury TAC: <u>blitzclaim@litigationkc.com</u>; dhaltiwanger@rpwb.com

- (b) All such notices and communications shall be effective, if electronically transmitted, when the communication is received at the designated addresses and confirmed by the recipient by return transmission.
- 6.8 <u>Successors and Assigns.</u> The provisions of this Agreement shall be binding upon and inure to the benefit of the Protected Parties, the Blitz Personal Injury Trust, the Blitz Personal Injury Trustee, and their respective successors and assigns, except that neither the Protected Parties, the Blitz Personal Injury Trust, the Blitz Personal Injury Trustee, may assign or otherwise transfer any of its, or their, rights or obligations under this Agreement except, in the case of the Blitz Personal Injury Trust and the Blitz Personal Injury Trustee, as contemplated by section 2.1 above.
- 6.9 Entire Agreement; No Waiver. The entire agreement of the parties relating to the subject matter of this Agreement is contained herein and in the Plan Documents referred to herein, and this Agreement and such documents supersede any prior oral or written agreements concerning the subject matter hereof. No failure to exercise or delay in exercising any right, power or privilege hereunder shall operate as a waiver thereof, nor shall any single or partial exercise of any right, power or privilege hereunder preclude any further exercise thereof or of any other right, power or privilege. The rights and remedies herein provided are cumulative and are not exclusive of rights under law or in equity.
- **6.10 <u>Headings.</u>** The headings used in this Agreement are inserted for convenience only and do not constitute a portion of this Agreement, nor in any manner affect the construction of the provisions of this Agreement.

- 6.11 Governing Law. This Agreement shall be governed by, and construed in accordance with, the laws of the State of Delaware, without regard to conflict of law principles.
- 6.12 <u>Debtors' Representations and Cooperation.</u> The USA Debtors are hereby irrevocably designated as the settlors, and are hereby authorized to take any action required of the settlors in connection with this Agreement. Subject to possession of necessary funds to do so, the USA Debtors agree to reasonably cooperate as may be requested by the Blitz Personal Injury Trustee in implementing the goals and objectives of the Blitz Personal Injury Trust.
- 6.13 <u>Dispute Resolution</u>. Any disputes that arises under this Agreement or under the Blitz Personal Injury TDP may be resolved by (i) submission of the matter to an alternative dispute resolution ("ADR") process mutually-agreeable to the parties involved with such dispute, or (ii) application to the Bankruptcy Court for a judicial determination of the matter. Any review by the Bankruptcy Court of an agreed upon non-binding ADR procedure shall be *de novo*.
- **6.14** Enforcement and Administration. The provisions of this Agreement and the Blitz Personal Injury TDP shall be enforced by the Bankruptcy Court pursuant to the Plan. The parties hereby further acknowledge and agree that the Bankruptcy Court shall have exclusive jurisdiction over the settlement of the accounts of the Blitz Personal Injury Trustee and over any disputes hereunder not resolved by ADR in accordance with section 6.13 above.
- **6.15** Effectiveness. This Agreement shall not become effective until it has been executed and delivered by all the parties hereto.
- **6.16** Counterpart Signatures. This Agreement may be executed in any number of counterparts, each of which shall constitute an original, but such counterparts shall together constitute but one and the same instrument.

IN WITNESS WHEREOF, th	e parties have executed this Agreement this	da	ay of
January, 2014.			

BLITZ USA, INC., BLITZ RE HOLDINGS, LLC, BLITZ ACQUISITION, LLC and MIAMIOK, LLC f/k/a F3BRANDS

By:

Name and Title:
 LAM 2011 HOLDINGS, L.L.C. and BLITZ ACQUISITION HOLDINGS, INC.
By:
Name and Title:
BLITZ PERSONAL INJURY TRUSTEE
Expiration Date of Initial Term:
Anniversary of the date of this Agreement

Accepted and Agreed:	
BLITZ PERSONAL INJURY TRUST ADVISORY COMMITTEE	
Expiration Date of Initial Term: Anniversary of the date of this Blitz Personal Injury Trust Agreement	
Expiration Date of Initial Term:Anniversary of the date of this Blitz Personal Injury Trust Agreement	
Expiration Date of Initial Term: Anniversary of the date of this Blitz Personal Injury Trust Agreement	
Expiration Date of Initial Term: Anniversary of the date of this Blitz Personal Injury Trust Agreement	

EXHIBIT C

Blitz Liquidating Trust Agreement

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

_)	
In re)	
)	Case No. 11-13603 (PJW)
BLITZ U.S.A., Inc., et al. 1)))	Chapter 11
Debtors.))	Jointly Administered
) .	

BLITZ LIQUIDATING TRUST AGREEMENT

This Blitz Liquidating Trust Agreement (the "Blitz Liquidating Trust Agreement") is made this ______ day of January, 2014 by and among Blitz U.S.A, Inc., Blitz Acquisition, LLC, Blitz RE Holdings, LLC and Miami OK LLC (f/k/a F3 Brands LLC) (each a "Debtor" and, collectively, the "USA Debtors"), and Steven D. Sass, as trustee (the "Blitz Liquidating Trustee") and executed in connection with the Debtors' and Official Committee of Unsecured Creditors' First Amended Joint Plan of Liquidation dated December 18, 2013 (as the same has been or may be amended, the "Plan")² filed in the United States Bankruptcy Court for the District of Delaware (the "Bankruptcy Court").

RECITALS

WHEREAS, on November 9, 2011, each of the Debtors filed a voluntary petition for relief under Chapter 11 of the Bankruptcy Code (the "Chapter 11 Cases"). The Chapter 11 Cases, which are captioned *In re Blitz U.S.A.*, *Inc. et al.*, Case No 11-13603 (PJW) are being jointly administered by the Bankruptcy Court; and

WHEREAS, the Debtors have liquidated substantially all of their assets under the provisions of the Bankruptcy Code and intend for the Blitz Liquidating Trust to liquidate any remaining assets; and

WHEREAS, the Debtors and the Official Committee of Unsecured Creditors of Debtors filed the Plan on December 18, 2013;

¹ The Debtors in these Chapter 11 Cases, along with the last four digits of each Debtor's federal tax identification number, are: LAM 2011 Holdings (8742); Blitz Acquisition Holdings, Inc. (8825); Blitz Acquisition, LLC(8979); Blitz RE Holdings, LLC (9071) Blitz U.S.A., Inc. (8104); and MiamiOK LLC (2604). The location of the Debtors' corporate headquarters and the Debtors' service address is 309 North Main Street, Miami, OK 74354.

² Capitalized terms not otherwise defined herein shall have the meanings ascribed thereto in the Plan.

WHEREAS, on	, 2014, the Bankruptcy Court enter	red an Order Confirming the
Plan (the "Confirmation		
WHEREAS, the	Effective Date of the Plan occurred on	, 2014; and

WHEREAS, the Plan contemplates, on the Effective Date, (a) the creation of a Blitz Liquidating Trust (the "Blitz Liquidating Trust") and the creation of the beneficial interests in the Blitz Liquidating Trust solely for the benefit of holders of Allowed General Unsecured Claims against the USA Debtors (collectively, the "Blitz Liquidating Trust Beneficiaries" and, each individually, a "Beneficiary"), and (b) the Blitz Liquidating Trust will be vested with the BAH Settlement Payment, Cash and all of the assets remaining in the USA Debtors' Estates on the Effective Date, including, without limitation, the sum of \$1.54 million due from Wal-Mart to Blitz USA (collectively, the "Blitz Liquidating Trust Assets") to be liquidated and distributed to the Blitz Liquidating Trust Beneficiaries, as set forth in the Plan; and

WHEREAS, the Plan contemplates that, pursuant to Treasury Regulation Section 301.7701-4(d), the Blitz Liquidating Trust shall be created for the purpose of: (a) administering the Blitz Liquidating Trust Assets; (b) resolving all Disputed Claims; (c) pursuing any causes of action, and (d) making all Distributions to the Blitz Liquidating Trust Beneficiaries provided for under the Plan, with no objective to continue or engage in the conduct of a trade or business, except to the extent reasonably necessary to and consistent with the liquidating purpose of the Blitz Liquidating Trust and the Plan; and

WHEREAS, the Blitz Liquidating Trust is intended to qualify as a "grantor trust" for U.S. federal income tax purposes, pursuant to Sections 671-677 of the Internal Revenue Code of 1986, as amended (the "IRC"), with the Blitz Liquidating Trust Beneficiaries to be treated as the grantors of the Blitz Liquidating Trust and deemed to be the owners of the Blitz Liquidating Trust Assets (subject to the rights of creditors of the Blitz Liquidating Trust), and consequently, the transfer of the Blitz Liquidating Trust Assets to the Blitz Liquidating Trust shall be treated as a deemed transfer of those assets from the USA Debtors and their Estates to the Blitz Liquidating Trust Beneficiaries to the Blitz Liquidating Trust for federal income tax purposes.

NOW, THEREFORE, pursuant to the Plan and the Confirmation Order, in consideration of the premises, the mutual agreements of the parties contained herein, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged and affirmed, it is hereby agreed as follows:

ARTICLE I DECLARATION OF TRUST

1.1 Creation and Purpose of the Blitz Liquidating Trust. The USA Debtors and the Blitz Liquidating Trustee hereby create the Blitz Liquidating Trust for the primary purpose of liquidating and distributing the Blitz Liquidating Trust Assets to the Blitz Liquidating Trust Beneficiaries in accordance with the Plan, the Confirmation Order, and applicable tax statutes, rules, and regulations, and in an expeditious but orderly manner, with no objective to continue or

engage in the conduct of a trade or business. In particular, the Blitz Liquidating Trustee shall (a) make continuing efforts to collect and reduce the Blitz Liquidating Trust Assets to Cash, and (b) make timely Distributions and not unduly prolong the duration of the Blitz Liquidating Trust.

- 1.2 Declaration of Trust. In order to declare the terms and conditions hereof, and in consideration of the confirmation of the Plan, the USA Debtors and the Blitz Liquidating Trustee have executed this Blitz Liquidating Trust Agreement and, effective on the Effective Date, the USA Debtors hereby irrevocably transfer to the Blitz Liquidating Trust, all of the right, title, and interests of the USA Debtors in and to the Blitz Liquidating Trust Assets, to have and to hold unto the Blitz Liquidating Trust and its successors and assigns forever, under and subject to the terms of the Plan and the Confirmation Order for the benefit of the Blitz Liquidating Trust Beneficiaries and their successors and assigns as provided for in this Blitz Liquidating Trust Agreement and in the Plan and Confirmation Order.
- 1.3 Vesting of Estate Assets. On the Effective Date, pursuant to the terms of the Plan, the Blitz Liquidating Trust Assets shall be vested in the Blitz Liquidating Trust, which also shall be authorized to obtain, liquidate, and collect all of the Blitz Liquidating Trust Assets in the possession of third parties and pursue all causes of action, if any. Subject to the provisions of the Plan, all such Blitz Liquidating Trust Assets shall be delivered to the Blitz Liquidating Trust free and clear of interests, Claims, Liens, or other encumbrances of any kind. The Blitz Liquidating Trustee shall be conclusively entitled to rely on the legality and validity of such transfers. Moreover, on the Effective Date, all privileges with respect to any Blitz Liquidating Trust Assets, including the attorney/client privilege, to which the USA Debtors are entitled shall be automatically vested in, and available for assertion by the Blitz Liquidating Trustee on behalf of the Blitz Liquidating Trust. To the extent any of the foregoing does not automatically occur on the Effective Date or is not effectuated through the Confirmation Order or this Agreement, the USA Debtors shall, on the Effective Date, execute such other and further documents as are reasonably necessary to effectuate all of the foregoing.
- 1.4 Funding of the Trust. The Blitz Liquidating Trust shall be funded, on the Effective Date, with the Blitz Liquidating Trust Assets, as provided for in the Plan and in the Confirmation Order.
- 1.5 Acceptance by Blitz Liquidating Trustee. The Blitz Liquidating Trustee hereby accepts the trust imposed upon it by this Blitz Liquidating Trust Agreement and agrees to observe and perform that trust on and subject to the terms and conditions set forth in this Blitz Liquidating Trust Agreement, the Plan, and the Confirmation Order. In connection with and in furtherance of the purposes of the Blitz Liquidating Trust, the Blitz Liquidating Trustee hereby accepts the transfer of the Blitz Liquidating Trust Assets.
- 1.6 Name of the Liquidating Trust. The Blitz Liquidating Trust established hereby shall be known as the "Blitz Liquidating Trust".
 - 1.7 Appointment of Oversight Committee. On or before the Effective Date, the Creditors' Committee shall designate the members of an Oversight Committee. The Blitz Liquidating Trustee shall confer with, and obtain approval of, the Oversight Committee

as set forth in this Blitz Liquidating Trust Agreement.

ARTICLE II THE LIQUIDATING TRUSTEE

- **2.1** Appointment. The Blitz Liquidating Trustee has been selected by the Creditors' Committee pursuant to the provisions of the Plan and has been appointed as of the Effective Date. The Blitz Liquidating Trustee's appointment shall continue until the earlier of (a) the date that the Blitz Liquidating Trust is dissolved in accordance with this Agreement, or (b) the date of the Blitz Liquidating Trustee's resignation, death, dissolution, removal or liquidation.
- Agreement, the Plan, or the Confirmation Order, the Blitz Liquidating Trustee may control and exercise authority over the Blitz Liquidating Trust Assets, over the acquisition, management, and disposition thereof, and over the management and conduct of the business of the Blitz Liquidating Trust. Nothing in this Agreement shall be deemed to prevent the Blitz Liquidating Trustee from taking, or failing to take, any action that, based upon the advice of counsel, it determines it is obligated to take (or fail to take) in the performance of any fiduciary or similar duty which the Blitz Liquidating Trustee owes to the Blitz Liquidating Trust Beneficiaries or any other Person or Entity. No Person dealing with the Blitz Liquidating Trust shall be obligated to inquire into the Blitz Liquidating Trustee's authority in connection with the acquisition, management, or disposition of Blitz Liquidating Trust Assets. Without limiting the foregoing, but subject to the Plan, the Confirmation Order, and other provisions of this Blitz Liquidating Trust Agreement, the Blitz Liquidating Trustee shall be expressly authorized to, with respect to the Blitz Liquidating Trust and the Blitz Liquidating Trust Assets:
 - (a) Hold legal title to any and all Blitz Liquidating Trust Assets.
- (b) Exercise all power and authority that may be or could have been exercised, commence all proceedings that may be or could have been commenced, and take all actions that may be or could have been taken with respect to the Blitz Liquidating Trust Assets by any officer, director, shareholder or other party acting in the name of the USA Debtors or their Estates with like effect as if duly authorized, exercised, and taken by action of such officers, directors, shareholders or other party.
- (c) Open and maintain bank accounts on behalf of or in the name of the Blitz Liquidating Trust, calculate and make Distributions, and take other actions consistent with the Plan and the implementation thereof, including the establishment, re-evaluation, adjustment, and maintenance of appropriate reserves, including the USA Debtors Contingent Claims Cash Reserve and the USA Debtors Disputed Claims Reserve, in accordance with the terms of the Plan.
- (d) Receive, manage, invest, supervise, and protect the Blitz Liquidating Trust Assets, subject to the limitations provided herein.
- (e) Subject to the applicable provisions of the Plan and this Blitz Liquidating Trust Agreement, collect and liquidate all Blitz Liquidating Trust Assets pursuant to the Plan.

- (f) Conduct an analysis of Administrative Expense Claims (including fee applications of Bankruptcy Professionals, *provided however*, any analysis of such fee applications shall be subject to and in accordance with the BAH Settlement Term Sheet) against the USA Debtors, Priority Claims against the USA Debtors, Secured Claims against the USA Debtors and General Unsecured Claims against the USA Debtors, and prosecute objections thereto or settling or otherwise compromising such Claims if necessary and appropriate.
- (g) Supervise and administer the commencement, prosecution, settlement, compromise, withdrawal or resolution in any manner approved by the Bankruptcy Court of all Disputed Claims and the Distributions to the Blitz Liquidating Trust Beneficiaries and creditors of the Blitz Liquidating Trust, in accordance with this Blitz Liquidating Trust Agreement, the Plan, and the Confirmation Order.
- (h) Subject to Article IV of this Blitz Liquidating Trust Agreement, commence, prosecute, compromise, settle, withdraw, abandon, or resolve in any manner approved by the Bankruptcy Court any causes of action.
- (i) Seek a determination of tax liability under Section 505 of the Bankruptcy Code; file, if necessary, any and all tax and information returns required with respect to the Blitz Liquidating Trust; make tax elections for and on behalf of the Blitz Liquidating Trust; and pay taxes, if any, payable for and on behalf of the Blitz Liquidating Trust.
- (j) Pay all lawful expenses, debts, charges, taxes and liabilities of the Blitz Liquidating Trust.
- (k) Make Distributions to the Blitz Liquidating Trust Beneficiaries, and to creditors of the Blitz Liquidating Trust as provided for, or contemplated by, the Plan, the Confirmation Order, and this Blitz Liquidating Trust Agreement.
- (l) Withhold from the amount distributable to any Person or Entity such amount as may be sufficient to pay any tax or other charge which the Blitz Liquidating Trustee has determined, based upon the advice of its agents and/or professionals, may be required to be withheld therefrom under the income tax laws of the United States or of any state or political subdivision thereof.
- (m) Enter into any agreement or execute any document or instrument required by or consistent with the Plan, the Confirmation Order or this Blitz Liquidating Trust Agreement and perform all obligations thereunder.
- (n) If any of the Blitz Liquidating Trust Assets are situated in any state or other jurisdiction in which the Blitz Liquidating Trustee is not qualified to act as trustee, nominate and appoint a person duly qualified to act as trustee in such state or jurisdiction and require from each such trustee such security as may be designated by the Blitz Liquidating Trustee in its discretion; confer upon such trustee all the rights, powers, privileges, and duties of the Blitz Liquidating Trustee hereunder, subject to the conditions and limitations of this Blitz Liquidating Trust Agreement, except as modified or limited by the Blitz Liquidating Trustee and except where the conditions and limitations may be modified by the laws of such state or other jurisdiction (in

which case, the laws of the state or other jurisdiction in which such trustee is acting shall prevail to the extent necessary); require such trustee to be answerable to the Blitz Liquidating Trustee for all monies, assets and other property that may be received in connection with the administration of all property; and remove such trustee, with or without cause, and appoint a successor trustee at any time by the execution by the Blitz Liquidating Trustee of a written instrument declaring such trustee removed from office, and specifying the effective date and time of removal.

- (o) Subject to Section 3.5 of this Blitz Liquidating Trust Agreement, purchase and carry all insurance policies and pay all insurance premiums and costs it deems reasonably necessary or advisable.
- (p) Implement, enforce, or discharge all of the terms, conditions, and all other provisions of, and all duties and obligations under, the Plan, the Confirmation Order, and this Blitz Liquidating Trust Agreement.
- (q) Employ and compensate such professionals and other agents as are necessary and appropriate in furtherance of the Blitz Liquidating Trustee's duties.
- (r) Undertake all administrative functions remaining in the USA Debtors' Chapter 11 Cases, including the preparation and filing of post-Effective Date operating reports for the USA Debtors and the ultimate closing of the USA Debtors' Chapter 11 Cases.
- (s) Invest the Blitz Liquidating Trust Assets transferred to the Blitz Liquidating Trust, the proceeds thereof, or any income earned by the Blitz Liquidating Trust in Permissible Investments, provided however, that the scope of any such Permissible Investments shall be limited to include only those investments that a "liquidating trust" within the meaning of Treasury Regulation Section 301.7701-4(d) may be permitted to hold, pursuant to the Treasury Regulations or any modification in the Internal Revenue Services ("IRS") guidelines, whether set forth in IRS rulings, revenue procedures, other IRS pronouncements or otherwise.
- (t) Take all other actions consistent with the provisions of the Plan which the Blitz Liquidating Trustee deems reasonably necessary or desirable to administer the Blitz Liquidating Trust and the Plan.
 - 2.3 Limitations on the Blitz Liquidating Trustee. Notwithstanding anything under applicable law, this Blitz Liquidating Trust Agreement or the Plan to the contrary, the Blitz Liquidating Trustee shall not do or undertake any of the following:
- (a) Take, or fail to take, any action that would jeopardize treatment of the Blitz Liquidating Trust as a "liquidating trust" for federal income tax purposes.
- (b) Receive transfers of any listed stocks or securities, any readily-marketable assets or any operating assets of a going business, except as is absolutely necessary or required under the Plan and the Confirmation Order; provided, however, that in no event shall the Blitz Liquidating Trustee receive any such investment that would jeopardize treatment of the Blitz Liquidating Trust as a "liquidating trust" for federal income tax purposes.

- (c) Exercise any investment power other than the power to invest in demand and time deposits in banks or savings institutions, or temporary investments such as short term certificates of deposit or Treasury bills or other investments that a "liquidating trust" within the meaning of Treasury Regulation Section 301.7701-4(d) may be permitted to hold, pursuant to the Treasury Regulations or any modification in the IRS guidelines, whether set forth in IRS rulings, revenue procedures, other IRS pronouncements or otherwise.
- (d) Receive or retain any operating assets of a going business, a partnership interest in a partnership that holds operating assets, or fifty percent (50%) or more of the stock of a corporation with operating assets, except as is absolutely necessary or required under the Plan and the Confirmation Order; <u>provided</u>, <u>however</u>, that in no event shall the Blitz Liquidating Trustee receive or retain any such asset or interest that would jeopardize treatment of the Blitz Liquidating Trust as a "liquidating trust" for federal income tax purposes.
- (e) Notwithstanding any of the foregoing, the Blitz Liquidating Trustee shall not be prohibited from engaging in any trade or business on its own account, provided that such activity does not interfere with the Blitz Liquidating Trustee's administration of the Blitz Liquidating Trust.

2.4 Compensation of Blitz Liquidating Trustee and its Agents and Professionals.

- (a) The Blitz Liquidating Trustee shall be entitled to receive reasonable compensation (fees and reasonable out-of-pocket expenses, if any incurred in connection with this Blitz Liquidating Trust Agreement prior to the Effective Date and retroactive to January 6, 2014, as well for the performance of its duties after the Effective Date, as negotiated between the Blitz Liquidating Trustee and the Oversight Committee, plus the reimbursement of all reasonable out-of-pocket expenses incurred in connection with this Blitz Liquidating Trust Agreement. Any successor to the Blitz Liquidating Trustee shall also be entitled to reasonable compensation in connection with the performance of its duties, which compensation may be different from the terms provided herein, plus the reimbursement of reasonable out-of-pocket expenses.
- (b) The Blitz Liquidating Trustee's fees shall be paid by wire transfer on the first day of each month.
- (c) The Blitz Liquidating Trustee (only with respect to its fees and expenses incurred prior to the Effective Date and the reimbursement of its reasonable out-of-pocket expenses incurred after the Effective Date) and each of it agents and professionals (unless any such agents or professionals and the Blitz Liquidating Trustee agree to different treatment) seeking compensation or reimbursement shall serve a statement on the Blitz Liquidating Trustee and the Oversight Committee will have fifteen (15) days from the date such statement is received to review the statement and object to such statement by serving an objection on the party seeking compensation (including the Blitz Liquidating Trustee) setting forth the precise nature of the objection and the amount at issue. At the expiration of the fifteen (15) day period, and without further order of the Bankruptcy Court (except as provided herein), the Blitz Liquidating Trustee shall pay from the Blitz Liquidating Trust Assets, or the proceeds or income thereof, 100% of the amounts requested, except for the

portion of such fees and expenses to which any objection has been made. The parties shall attempt to consensually resolve objections, if any, to any statement. If the parties are unable to reach a consensual resolution of any such objection, the party who received an objection to its fees and expenses may seek payment of such fees and expenses by filing a motion with and obtaining an order from the Bankruptcy Court and providing notice to the Blitz Liquidating Trustee [and the Oversight Committee]. If the Blitz Liquidating Trustee or its agents or professional fails to submit a statement, it shall be ineligible to receive payment of fees and expenses therefore as provided in this Blitz Liquidating Trust Agreement until the statement is submitted.

- General Duties, Obligations, Rights, and Benefits of the Liquidating Trustee. 2.5 The Blitz Liquidating Trustee shall have all duties, obligations, rights, and benefits assumed by, assigned to or vested in the Blitz Liquidating Trust under the Plan, the Confirmation Order, this Blitz Liquidating Trust Agreement and any other agreement entered into pursuant to or in Such duties, obligations, rights and benefits include, without connection with the Plan. limitation, all duties, obligations, rights and benefits relating to the collection and liquidation of the Blitz Liquidating Trust Assets, administration of Claims, satisfaction of claims of creditors of the Blitz Liquidating Trust, the pursuit of causes of action, if any, Distributions to Blitz Liquidating Trust Beneficiaries, administration of the Blitz Liquidating Trust and any other duties, obligations, rights, and benefits reasonably necessary to accomplish the purpose of the Blitz Liquidating Trust under the Plan, the Confirmation Order, this Blitz Liquidating Trust Agreement, and any other agreement entered into pursuant to or in connection with the Plan. Without limiting the duties, obligations, rights, and benefits of the Blitz Liquidating Trustee under this Section or any other provision of this Blitz Liquidating Trust Agreement, the Blitz Liquidating Trustee shall have all duties, obligations, rights, and benefits assigned to the Blitz Liquidating Trustee under the Confirmation Order.
- 2.6 USA Debtors Contingent Claims Cash Reserve. The Blitz Liquidating Trustee may establish, from time to time fund, and administer a reserve (the "USA Debtors Contingent Claims Cash Reserve"), that shall consist of Cash in an amount which shall be the good faith estimate of the total Distributions to be made on account of all Contingent Claims against the USA Debtors as of the Effective Date. The USA Debtors or the Blitz Liquidating Trustee as applicable, shall seek Bankruptcy Court approval of such Contingent Claims Cash Reserve on the Effective Date or as soon thereafter as practicable. The Blitz Liquidating Trustee shall be authorized to make distributions from the USA Debtors Contingent Claims Cash Reserve in satisfaction of such Contingent Claims, once Allowed, in accordance with this Blitz Liquidating Trust Agreement, the Plan, and the Confirmation Order.
- 2.7 USA Debtors Disputed Claims Cash Reserve. The Blitz Liquidating Trustee may establish, from time to time fund, and administer a reserve (the "USA Debtors Disputed Claims Cash Reserve"), that shall consist of Cash in an amount which shall be the good faith estimate of the total Distributions to be made on account of all Disputed Claims against the USA Debtors as of the Effetive Date, as determined by the Blitz Liquidating Trustee. The Blitz Liquidating Trustee shall be authorized to make distributions from the USA Debtors Disputed Claims Cash Reserve in satisfaction of such Disputed Claims, once Allowed, in accordance with this Blitz Liquidating Trust Agreement, the Plan, and the Confirmation Order.

- 2.8 Replacement of the Blitz Liquidating Trustee. The Blitz Liquidating Trustee may resign at any time upon thirty (30) days' written notice delivered to the Bankruptcy Court [and the Oversight Committee], provided that such resignation shall only become effective upon the appointment of a permanent or interim successor Blitz Liquidating Trustee. The Blitz Liquidating Trustee may also be removed by the Bankruptcy Court upon application [by the Oversight Committee] and after notice and a hearing. In the event of the resignation or removal of the Blitz Liquidating Trustee, a successor Blitz Liquidating Trustee shall be appointed by the Bankruptcy Court based upon submissions from interested parties (including the Oversight Committee, Blitz Liquidating Trustee or any Blitz Liquidating Trust Beneficiary). Upon its appointment, the successor Blitz Liquidating Trustee, without any further act, shall become fully vested with all of the rights, powers, duties, and obligations of its predecessor, except as provided in Section 2.4(a) above, and all responsibilities of the predecessor Blitz Liquidating Trustee relating to the Blitz Liquidating Trust shall be terminated. In the event the Blitz Liquidating Trustee's appointment terminates by reason of death, dissolution, liquidation, resignation, or removal, such Blitz Liquidating Trustee shall be immediately compensated for all reasonable fees and expenses accrued through the effective date of termination, whether or not previously invoiced. The provisions of Article V of this Blitz Liquidating Trust Agreement shall survive the resignation or removal of any Blitz Liquidating Trustee.
- 2.9 Blitz Liquidating Trust Continuance. The death, dissolution, liquidation, resignation, or removal of the Blitz Liquidating Trustee shall not terminate the Blitz Liquidating Trust or revoke any existing agency created by the Blitz Liquidating Trustee pursuant to this Blitz Liquidating Trust Agreement or invalidate any action theretofore taken by the Blitz Liquidating Trustee, and the provisions of this Blitz Liquidating Trust Agreement shall be binding upon and inure to the benefit of the successor Blitz Liquidating Trustee and all its successors or assigns.

ARTICLE III LIABILITY OF BLITZ LIQUIDATING TRUSTEE

Standard of Care; Exculpation. Neither the Blitz Liquidating Trustee, nor any director, officer, affiliate, employee, employer, professional, successors, assigns, agent, or representative of the Blitz Liquidating Trustee (each, an "Exculpated Party" and collectively, the "Exculpated Parties") shall be liable for any losses, claims, damages, liabilities, obligations, settlements, proceedings, suits, judgments, causes of action, litigation, actions, or investigations (whether civil or administrative and whether sounding in tort, contract or otherwise), penalties, costs, and expenses, including reasonable fees and disbursements (collectively referred to herein as "Losses"), whether or not in connection with litigation in which any Exculpated Party is a party, or enforcing this Blitz Liquidating Trust Agreement (including these exculpation provisions), as and when imposed on the Blitz Liquidating Trustee, incurred, caused by, relating to, based upon or arising out of (directly or indirectly) the Blitz Liquidating Trustee's execution, delivery, and acceptance of or the performance or nonperformance of its powers, duties and obligations under this Blitz Liquidating Trust Agreement, the Plan, or the Confirmation Order or as may arise by reason of any action, omission or error of an Exculpated Party; provided, however, that the foregoing limitation shall not apply to any Losses suffered or incurred by any holder of an Allowed Claim of a Blitz Liquidating Trust Beneficiary that are found in a final

judgment by a court of competent jurisdiction (not subject to further appeal) to have resulted from the fraud, gross negligence or willful misconduct of such Exculpated Party. Every act taken or omitted, power exercised or obligation assumed by the Blitz Liquidating Trust or any Exculpated Party pursuant to the provisions of this Blitz Liquidating Trust Agreement shall be held to be taken or omitted, exercised, or assumed, as the case may be, by the Blitz Liquidating Trust or any Exculpated Party acting for and on behalf of the Blitz Liquidating Trust and not otherwise; provided, however, that none of the foregoing Entities or Persons are deemed to be responsible for any other such Entities' or Persons' actions or inactions. Except as provided in the first proviso of the first sentence of this Section 3.1, every Person, firm, corporation, or other Entity contracting or otherwise dealing with or having any relationship with the Blitz Liquidating Trust or any Exculpated Party shall have recourse only to the Blitz Liquidating Trust Assets for payment of any liabilities or other obligations arising in connection with such contracts, dealings or relationships and the Blitz Liquidating Trust and the Exculpated Parties shall not be individually liable therefore. In no event shall the Blitz Liquidating Trustee be liable for indirect, punitive, special, incidental, or consequential damage or loss (including but not limited to lost profits) whatsoever, even if the Blitz Liquidating Trustee has been informed of the likelihood of such loss or damages and regardless of the form of action.

3.2 Indemnification.

- The Blitz Liquidating Trustee and any director, officer, affiliate, employee, employer, professional, successor, assign, agent, or representative of the Blitz Liquidating Trustee (each, an "Indemnified Party" and collectively, the "Indemnified Parties") shall be defended, held harmless, and indemnified from time to time by the Blitz Liquidating Trust against any and all Losses, including, without limitation, the costs for counsel or others in investigating, preparing, defending, or settling any action or claim, whether or not in connection with litigation in which any Indemnified Party is a party, or enforcing this Blitz Liquidating Trust Agreement (including these indemnity provisions), as and when imposed on the Blitz Liquidating Trustee, incurred, caused by, relating to, based upon or arising out of (directly or indirectly) the Blitz Liquidating Trustee's execution, delivery, and acceptance of or the performance or nonperformance of its powers, duties, and obligations under this Blitz Liquidating Trust Agreement, the Plan, or the Confirmation Order or as may arise by reason of any action, omission, or error of an Indemnified Party; provided, however, such indemnity shall not apply to any such Losses to the extent it is found in a final judgment by a court of competent jurisdiction (not subject to further appeal) to have resulted from the fraud, gross negligence, or willful misconduct of such Indemnified Party. Satisfaction of any obligation of the Blitz Liquidating Trust arising pursuant to the terms of this Section shall be payable only from the Blitz Liquidating Trust Assets, shall be advanced prior to the conclusion of such matter and such right to payment shall be prior and superior to any other rights to receive a distribution of the Blitz Liquidating Trust Assets.
- (b) The Blitz Liquidating Trust shall promptly pay to the Indemnified Party the expenses set forth in subparagraph (a) above upon submission of invoices therefore on a current basis. Each Indemnified Party hereby undertakes, and the Blitz Liquidating Trust hereby accepts its undertaking, to repay any and all such amounts so paid by the Liquidating Trust if it shall ultimately be determined that such Indemnified Party is not entitled to be indemnified therefore under this Blitz Liquidating Trust Agreement.

- 3.3 No Liability for Acts of Successor/Predecessor Liquidating Trustees. Upon the appointment of a successor Blitz Liquidating Trustee and the delivery of the Blitz Liquidating Trust Assets to the successor Blitz Liquidating Trustee, the predecessor Blitz Liquidating Trustee and any director, officer, affiliate, employee, employer, professional, agent, or representative of the predecessor Blitz Liquidating Trustee shall have no further liability or responsibility with respect thereto. A successor Blitz Liquidating Trustee shall have no duty to examine or inquire into the acts or omissions of its immediate or remote predecessor and no successor Blitz Liquidating Trustee shall be in any way liable for the acts or omissions of any predecessor Blitz Liquidating Trustee unless a successor Blitz Liquidating Trustee expressly assumes such responsibility. A predecessor Blitz Liquidating Trustee shall have no liability for the acts or omissions of any immediate or subsequent successor Blitz Liquidating Trustee for any events or occurrences subsequent to the cessation of its role as Blitz Liquidating Trustee.
- 3.4 Reliance by Blitz Liquidating Trustee on Documents or Advice of Counsel. Except as otherwise provided in this Blitz Liquidating Trust Agreement, the Blitz Liquidating Trustee, any director, officer, affiliate, employee, employer, professional, agent, or representative of the Blitz Liquidating Trustee may rely, and shall be protected from liability for acting or failing to act, upon any resolution, certificate, statement, instrument, opinion, report, notice, request, consent, order, or other paper or document reasonably believed by the Blitz Liquidating Trustee to be genuine and to have been presented by an authorized party. Neither the Blitz Liquidating Trustee nor any director, officer, affiliate, employee, employer, professional, agent, or representative of the Blitz Liquidating Trustee shall be liable for any action taken or omitted or suffered by the Blitz Liquidating Trustee, in reasonable reliance upon the advice of counsel or other professionals engaged by the Blitz Liquidating Trustee in accordance with this Blitz Liquidating Trust Agreement. The Blitz Liquidating Trustee shall be fully indemnified by the Blitz Liquidating Trust for or in respect of any action taken, suffered or omitted by it and in accordance with such advice or opinion.
- 3.5 Insurance. The Blitz Liquidating Trustee, may purchase, using the Blitz Liquidating Trust Assets, and carry all insurance policies and pay all insurance premiums and costs the Blitz Liquidating Trustee deems reasonably necessary or advisable, including, without limitation, purchasing any errors and omissions insurance with regard to any Losses it may incur, arising out of or due to its actions or omissions, or consequences of such actions or omissions, other than as a result of its fraud or willful misconduct, with respect to the implementation and administration of the Plan or this Blitz Liquidating Trust Agreement. [The Blitz Liquidating Trustee shall obtain the approval of the Oversight Committee prior to purchasing any such insurance policies].
- 3.6 The provisions of this Article III shall survive the termination of this Blitz Liquidating Trust Agreement and the death, resignation, removal, liquidation, dissolution, or replacement of the Blitz Liquidating Trustee.

ARTICLE IV GENERAL PROVISIONS CONCERNING ADMINISTRATION OF THE LIQUIDATING TRUST

- **4.1 Liquidating Trust Reserves.** The Blitz Liquidating Trustee may, at its discretion, establish the Blitz Liquidating Trust Reserves as set forth in Section 2.6 and 2.7 of this Blitz Liquidating Trust Agreement.
- 4.2 Register of Blitz Liquidating Trust Beneficiaries. The Blitz Liquidating Trustee shall maintain at all times a register of the names, distribution addresses, amounts of Allowed Claims, and the ratable interests in the Blitz Liquidating Trust of the Blitz Liquidating Trust Beneficiaries (the "Register"). The initial Register shall be delivered to the Blitz Liquidating Trustee by the USA Debtors and shall be based on the list of holders of General Unsecured Claims against the USA Debtors maintained by KCC as of the Effective Date and prepared in accordance with the provisions of the Plan and the Confirmation Order. The Blitz Liquidating Trustee shall file the initial Register on the docket of the USA Debtors' Chapter 11 Cases. All references in this Blitz Liquidating Trust Agreement to holders of beneficial interests in the Blitz Liquidating Trust shall be read to mean holders of record as set forth in the Register maintained by the Blitz Liquidating Trustee and shall exclude any beneficial owner not recorded on such Register. The Blitz Liquidating Trustee shall cause the Register to be kept at its office or at such other place or places as may be designated by the Blitz Liquidating Trustee from time to time.

4.3 Books and Records.

- (a) On the Effective Date, the USA Debtors shall transfer and assign to the Blitz Liquidating Trust full title to, and the Blitz Liquidating Trust shall be authorized to take possession of, all of the books and records of the USA Debtors except those that pertain to (a) the Blitz Personal Injury Trust Claims, and (b) the Blitz Personal Injury Trust Assets, including but not limited to, insurance policies, self-insured retentions, deductibles, retrospective premiums, dividend payments, procurement of insurance, and the submission or payment of insurance claims and any other books and records transferred in accordance with Section 4.9 of the Plan. The Blitz Liquidating Trust shall have the responsibility of storing and maintaining books and records transferred hereunder until the USA Debtors' Chapter 11 Cases are closed, after which time such books and records may, to the extent not prohibited by applicable law, be abandoned or destroyed without further Bankruptcy Court order. For the purpose of this Section 5.3, books and records include computer generated or computer maintained books and records and computer data, as well as electronically generated or maintained books and records or data, along with books and records of the USA Debtors maintained by or in possession of third parties and all of the claims and rights of the USA Debtors in and to their books and records, wherever located.
- (b) The Blitz Liquidating Trustee also shall maintain in respect of the Blitz Liquidating Trust and the Blitz Liquidating Beneficiaries books and records relating to the Liquidating Trust Assets and any income or proceeds realized therefrom and the payment of expenses of and claims against or assumed by the Blitz Liquidating Trust in such detail and for

such period of time as may be necessary to enable it to make full and proper reports in respect thereof. Except as expressly provided in this Blitz Liquidating Trust Agreement, the Plan, or the Confirmation Order, or as may be required by applicable law (including securities law), nothing in this Blitz Liquidating Trust Agreement is intended to require the Blitz Liquidating Trust to file any accounting or seek approval of any court with respect to the administration of the Blitz Liquidating Trust, or as a condition for making any payment or distribution out of the Blitz Liquidating Trust Assets. Blitz Liquidating Trust Beneficiaries shall have the right upon thirty (30) days' prior written notice delivered to the Blitz Liquidating Trustee to inspect the Blitz Liquidating Trust's books and records, including the Register, provided such Blitz Liquidating Trust Beneficiary shall have entered into a confidentiality agreement in form and substance reasonably satisfactory to the Blitz Liquidating Trustee. Satisfaction of the foregoing condition notwithstanding, if (a) the Blitz Liquidating Trustee determines in good faith that the inspection of the Blitz Liquidating Trust's books and records, including the Register, by any Beneficiary would be detrimental to the Blitz Liquidating Trust or (b) such Blitz Liquidating Trust Beneficiary is a defendant (or potential defendant) in a pending (or potential) action brought by the Blitz Liquidating Trust, the Blitz Liquidating Trust may deny such request for inspection. The Bankruptcy Court shall resolve any dispute between any Beneficiary and the Liquidating Trustee under this Section 5.3.

Transfer of Privileged Information and Confidential Information. On the Effective Date or as soon as practicable thereafter as is reasonably practicable, except for the Blitz Personal Injury Privileged Information (which shall be treated as set forth in Section 4.10 of the Plan), the Privileged Information of the USA Debtors shall be transferred, assigned, given over to, and shall vest exclusively in the Blitz Liquidating Trustee. Further, with regard to any such privileges, (i) they are transferred or contributed for the sole purpose of enabling the Blitz Liquidating Trustee to perform its duties to administer the Blitz Liquidating Trust and for no other reason, (ii) they are vested solely in the Blitz Liquidating Trustee, and not in the Blitz Liquidating Trust or any other entity, committee or subcomponent of the Blitz Liquidating Trust, or any other person (including counsel) who has been engaged by, represents or has represented any Blitz Personal Injury Trust Claimant or any person who alleges or may allege a claim directly or indirectly relating to or arising from the Debtors' products, premises or operations, (iii) they shall be preserved and not waived, (iv) for the avoidance of doubt (if any), any such transfer or contribution shall have no effect on any right, claim or privilege of any person other than the Debtors, and (v) no information subject to a privilege or a prior assertion thereof shall be publicly disclosed by the Blitz Liquidating Trustee or the Blitz Liquidating Trust or communicated to any person not entitled to receive such information nor in a manner that would diminish the protected status of any such information. To the extent not subject to an applicable privilege or immunity in accordance with the foregoing, the Blitz Liquidating Trustee and any of his or her Representatives shall maintain the confidentiality of all Confidential Information and such Confidential Information may only be disclosed to the following Persons: (i) the Blitz Liquidating Trustee and outside counsel to the Blitz Liquidating Trustee; (ii) experts, consultants or non-legal professionals who actively assist the Blitz Liquidating Trustee in the analysis, valuation and/or litigation of any Claim against the Blitz Liquidating Trust or the Blitz Liquidating Trust Assets; (iii) the Blitz Personal Injury Trustee and outside counsel to the Blitz Personal Injury Trustee solely to the extent necessary to comply with Section 4.9 and/or 5.11 of the Plan; (iv) paralegal, stenographic, technical, clerical, document management and secretarial personnel employed by any of the foregoing; (v) the Bankruptcy Court and court personnel,

including stenographic, video, or audio reporters; (vi) any person identified on the face of any such Confidential Information as an author or recipient thereof; (vii) any person who is determined to have been an author and/or previous recipient of the Confidential Information, but is not identified on the face thereof; and (viii) during depositions or trial testimony (or preparation therefor), witnesses to whom disclosure is reasonably necessary, *provided*, *however* that none of the foregoing Entities is, has not been engaged by, represents or has represented any holder of a Blitz Personal Injury Claim or any other person who has or may assert a claim directly or indirectly relating to, based upon or arising from the Debtors' products, premises or operations.

- 4.5 Final Accounting of Blitz Liquidating Trustee. The Blitz Liquidating Trustee (or any such successor Blitz Liquidating Trustee) shall within one hundred twenty (120) days after the termination of the Blitz Liquidating Trust or the death, dissolution, liquidation, resignation, or removal of the Blitz Liquidating Trustee, render an accounting containing the following information:
 - (a) A description of the Blitz Liquidating Trust Assets.
- (b) A summarized accounting in sufficient detail of all gains, losses, receipts, disbursements and other transactions in connection with the Blitz Liquidating Trust and the Blitz Liquidating Trust Assets during the Blitz Liquidating Trustee's term of service, including their source and nature.
- (c) The ending balance of all Blitz Liquidating Trust Assets as of the date of the accounting, including the Cash balance on hand and the name(s) and location(s) of the depository or depositories where the Cash is kept.
 - (d) All known liabilities of the Blitz Liquidating Trust.
 - (e) All pending actions.
 - 4.6 Filing of Accounting. The final accounting described in Section 4.5 shall be filed with the Bankruptcy Court and all Blitz Liquidating Trust Beneficiaries shall have notice that the final accounting has been filed and an opportunity to have a hearing on the approval of the accounting and the discharge and release of the Blitz Liquidating Trustee.

ARTICLE V BENEFICIAL INTERESTS AND BENEFICIARIES

- 5.1 Trust Beneficial Interests. Each holder of an Allowed General Unsecured Claim against the USA Debtors, shall be entitled to receive beneficial interests in accordance with the treatment of such Claim under the Plan, and shall be entitled to distributions as set forth in the Plan.
- 5.2 Interest Beneficial Only. Ownership of a beneficial interest in the Blitz Liquidating Trust shall not entitle any Blitz Liquidating Trust Beneficiary to any title in or to the

Blitz Liquidating Trust Assets or to any right to call for a partition or division of the Blitz Liquidating Trust Assets or to require an accounting.

- 5.3 Evidence of Beneficial Interest. Ownership of a beneficial interest in the Blitz Liquidating Trust shall not be evidenced by any certificate, security, or receipt or in any other form or manner whatsoever, except as maintained on the books and records of the Blitz Liquidating Trust by the Blitz Liquidating Trustee, which may be the Register.
- 5.4 Exemption from Registration. The parties hereto intend that the rights of the holders of the beneficial interests arising under this Blitz Liquidating Trust Agreement shall not be "securities" under applicable laws, but none of the parties hereto represents or warrants that such rights shall not be securities or shall be entitled to exemption from registration under applicable securities laws. If such rights constitute securities, the parties hereto intend for the exemption from registration provided by Section 1145 of the Bankruptcy Code and by other applicable law to apply to their issuance under the Plan.
- 5.5 Transfers of Beneficial Interests. Beneficial interests in the Blitz Liquidating Trust shall be nontransferable except upon death of the interest holder or by operation of law. The Blitz Liquidating Trust shall not have any obligation to recognize any transfer of Claims occurring after the Record Date. Only those holders of Claims of record stated on the transfer ledgers as of the close of business on the Record Date, to the extent applicable, shall be entitled to be recognized for all purposes hereunder.
- 5.6 Absolute Owners. The Blitz Liquidating Trustee may deem and treat the Blitz Liquidating Trust Beneficiary reflected as the owner of a beneficial interest on the Register as the absolute owner thereof for the purposes of receiving Distributions and payments on account thereof for federal and state income tax purposes and for all other purposes whatsoever.
- 5.7 Change of Address. A Blitz Liquidating Trust Beneficiary may, after the Effective Date, select an alternative distribution address by filing a notice with the Bankruptcy Court (copy served on the Blitz Liquidating Trustee) identifying such alternative distribution address. Absent such notice, the Blitz Liquidating Trustee shall not recognize any such change of distribution address. Such notification shall be effective only upon receipt by the Blitz Liquidating Trustee.
- 5.8 Effect of Death, Dissolution, Incapacity, or Bankruptcy of Beneficiary. The death, dissolution, incapacity, or bankruptcy of a Beneficiary during the term of the Liquidating Trust shall not operate to terminate the Liquidating Trust during the term of the Liquidating Trust nor shall it entitle the representative or creditors of the deceased, incapacitated or bankrupt Beneficiary to an accounting or to take any action in any court or elsewhere for the distribution of the Liquidating Trust Assets or for a partition thereof nor shall it otherwise affect the rights and obligations of the Beneficiary under this Liquidating Trust Agreement or in the Liquidating Trust.
- 5.9 Standing. Except as expressly provided in this Blitz Liquidating Trust Agreement, the Plan or the Confirmation Order, a Blitz Liquidating Trust Beneficiary does not

have standing to direct the Blitz Liquidating Trustee to do or not to do any act or to institute any action or proceeding at law or in equity against any party upon or with respect to the Blitz Liquidating Trust Assets.

ARTICLE VI PROCEDURES FOR RESOLVING AND TREATING DISPUTED CLAIMS

6.1 Incorporation of Plan Provisions. As of the Effective Date, the Blitz Liquidating Trust shall assume responsibility for all Claims matters against the USA Debtors, other than Blitz Personal Injury Claims, as set forth in Section 5.7 of the Plan. In accordance with the Plan, the Blitz Liquidating Trust shall establish an appropriate USA Debtors Disputed Claims Cash Reserve and a USA Debtors Contingent Claims Cash Reserve pending resolution, as set forth in the Plan, of all contested matters and adversary proceedings concerning Disputed Claims and Contingent Claims against the USA Debtors.

6.2 USA Debtors Disputed Claims Cash Reserve.

- (a) Establishment of USA Debtors Disputed Claims Cash Reserve. On the Effective Date, the Blitz Liquidating Trustee shall establish a separate USA Debtors Disputed Claims Cash Reserve and deposit Cash in an amount which shall be the good faith estimate of the total Distributions to be made on account of all Disputed Claims against the USA Debtors, as determined by the Blitz Liquidating Trustee.
- (b) Amounts to Be Reserved. The Blitz Liquidating Trustee shall reserve the Cash (or other property allocated for Distribution on account of each Disputed Claim) based upon the asserted amount of each such Disputed Claim or such lesser amount as may be estimated by the Blitz Liquidating Trustee or the Bankruptcy Court in accordance with the Plan. All Cash or other property allocable to Disputed Claims hereunder shall be distributed by the Blitz Liquidating Trustee from the USA Debtors Disputed Claims Cash Reserve on the initial Distribution date (or such other date on which Distributions are made pursuant to the Plan and this Blitz Liquidating Trust Agreement). To the extent that the property placed in a USA Debtors Disputed Claims Cash Reserve consists of Cash, that Cash shall be deposited in an interest-bearing account at a qualified institution, consistent with the terms and limitations of this Blitz Liquidating Trust Agreement.
- Claim shall be distributed by the Blitz Liquidating Trustee from the USA Debtors Disputed Claims Cash Reserve on the first Distribution Date after the Claim is Allowed. Distributions shall be made only to the extent of the aggregate distributions that the holder of any such Allowed Claim would have received had such Claim been Allowed as of the Effective Date (less any taxes paid with respect to amounts held in the USA Debtors Disputed Claims Cash Reserve). Distributions to each holder of a Disputed Claim that has become an Allowed Claim (and to the extent that the holder of the Disputed Claim has not received prior distributions on account of that Claim) shall be made in accordance with the provisions of the Plan.
- (d) Termination of USA Debtors Disputed Claims Cash Reserves. The USA Debtors Disputed Claims Cash Reserve shall be closed and extinguished by the Blitz

Liquidating Trustee when all Distributions and other dispositions of Cash or other property required to be made therefrom under the Plan and this Blitz Liquidating Trust Agreement have been made. Upon closure of the USA Debtors Disputed Claims Cash Reserve, all Cash and other property held in that Disputed Claims Cash Reserve shall revest in the Blitz Liquidating Trust as a part of the general Blitz Liquidating Trust Assets and such Cash and property shall be used to pay the fees and expenses of the Blitz Liquidating Trust in accordance with this Blitz Liquidating Trust Agreement, and thereafter distributed on a Pro Rata basis to holders of Allowed General Unsecured Claims against the USA Debtors.

- (e) Limitation of Liability for Funding the USA Debtors Disputed Claims Cash Reserve. The Blitz Liquidating Trustee shall have no duty to fund any USA Debtors Disputed Claims Cash Reserve beyond the good faith estimate as contemplated in the Plan.
- (f) Transmittal of Distributions and Notices. Any property or notice which a person is or becomes entitled to receive pursuant to the Plan and this Blitz Liquidating Trust Agreement may be delivered by regular mail, postage prepaid, in an envelope addressed to that person's address listed in the Register. Property distributed in accordance with this subsection shall be deemed delivered to such person regardless of whether such property is actually received by that person. Notice given in accordance with this subsection shall be effective only upon receipt.

ARTICLE VII DISTRIBUTIONS

- 7.1 Distributions to Beneficiaries from Blitz Liquidating Trust Assets. All payments to be made by the Blitz Liquidating Trust to any Blitz Liquidating Trust Beneficiary shall be made only in accordance with the Plan, the Confirmation Order and this Blitz Liquidating Trust Agreement and from the Blitz Liquidating Trust Assets (or from the income and proceeds realized from the Blitz Liquidating Trust Assets) net of the Blitz Liquidating Trust Expense Reserve, USA Debtors Disputed Claims Cash Reserve, USA Debtors Contingent Claims Cash Reserve and other reserves established by the Blitz Liquidating Trustee, if any, and only to the extent that the Blitz Liquidating Trust has sufficient Blitz Liquidating Trust Assets (or income and proceeds realized from the Blitz Liquidating Trust Assets) to make such payments in accordance with and to the extent provided for in the Plan, the Confirmation Order, and this Blitz Liquidating Trust Agreement.
- 7.2 **Distributions; Withholding.** The Blitz Liquidating Trustee shall make Distributions to holders of Allowed Claims as provided in the Plan and, following the Distribution required under the Plan, the Blitz Liquidating Trustee shall make Distributions to holders of Allowed General Unsecured Claims of the USA Debtors (including distributions of all net Cash (including net Cash proceeds)) on quarterly Distribution dates, to the extent possible; provided, however, that the Blitz Liquidating Trust may retain and supplement from time to time a reserve (the "Blitz Liquidating Trust Reserve") in such amount (a) as is reasonably necessary to meet contingent liabilities and to maintain the value of the Blitz Liquidating Trust Assets during the term of the Blitz Liquidating Trust; (b) to pay reasonable administrative expenses including, without limitation, the compensation and the reimbursement of reasonable, actual and

necessary costs, fees, and expenses (including attorneys' fees and expenses and other professionals' fees and expenses) of the Blitz Liquidating Trustee in connection with the performance of their duties in connection with this Blitz Liquidating Trust Agreement; and (c) to satisfy all other liabilities and claims of creditors of the Blitz Liquidating Trust incurred or assumed in respect of the Blitz Liquidating Trust (or to which the Blitz Liquidating Trust Assets are otherwise subject) in accordance with the Plan, the Confirmation Order and this Blitz Liquidating Trust Agreement. All such distributions shall be made as provided, and subject to any withholding or reserve, in this Blitz Liquidating Trust Agreement, the Plan or the Confirmation Order. Additionally, the Blitz Liquidating Trustee may withhold from amounts distributable to any Blitz Liquidating Trust Beneficiary any and all amounts, determined in the Blitz Liquidating Trustee's sole discretion, to be required by any law, regulation, rule, ruling, directive, or other governmental requirement. In addition, all distributions under this Blitz Liquidating Trust Agreement shall be net of the actual and reasonable costs of making such distributions.

- 7.3 No Distribution Pending Allowance. No payment or Distribution shall be made with respect to any Claim to the extent it is a Disputed Claim and/or a Contingent Claim unless and until such Disputed Claim and/or Contingent Claim becomes an Allowed Claim, except for distributions into a USA Debtors Disputed Claims Cash Reserve or USA Debtors Contingent Claims Cash Reserve in accordance with the Plan, Confirmation Order, and this Blitz Liquidating Trust Agreement.
- 7.4 Distributions after Allowance. Distributions to each holder of a Disputed Claim and/or a Contingent Claim, to the extent that such Claim ultimately becomes an Allowed Claim, shall be made in accordance with the provisions of the Plan, Confirmation Order, and this Blitz Liquidating Trust Agreement.
- 7.5 Non-Cash Property. Subject to Section 2.3 hereof, any non-Cash property of the Blitz Liquidating Trust may be sold, transferred or abandoned by the Blitz Liquidating Trustee [upon prior written notice to, and consent of, the Oversight Committee]. Notice of such sale, transfer, or abandonment shall be provided to the holders, if any, of Secured Claims holding liens on such non-Cash property. If, in the Blitz Liquidating Trustee's reasonable judgment, such property cannot be sold in a commercially reasonable manner, or the Blitz Liquidating Trustee believes, in good faith, such property has no value to the Blitz Liquidating Trust, the Blitz Liquidating Trustee shall have the right to abandon or otherwise dispose of such property, including by donation of such property to a charity designated by the Blitz Liquidating Trustee, [after written notice to, and consent of, the Oversight Committee]. Except in the case of fraud, willful misconduct, or gross negligence, no party in interest shall have a cause of action against the Blitz Liquidating Trustee or any director, officer, employee, consultant, or professional of the Blitz Liquidating Trustee arising from or related to the disposition of non-Cash property in accordance with this Section.
- 7.6 Undeliverable Distributions. If any Distribution is returned as undeliverable, the Blitz Liquidating Trust may, in its discretion, make such efforts to determine the current address of the holder of the Claim with respect to which the Distribution was made as the Blitz Liquidating Trustee deems appropriate, but no Distribution to any holder shall be made unless

and until the Blitz Liquidating Trust has determined the then-current address of the holder, at which time the Distribution to such holder shall be made to the holder without interest. Amounts in respect of any undeliverable Distributions made by the Blitz Liquidating Trust shall be returned to, and held in trust by, the Blitz Liquidating Trust until the Distributions are claimed or are deemed to be unclaimed property under section 347(b) of the Bankruptcy Code ("<u>Unclaimed Property</u>").

- 7.7 Unclaimed Property. Except with respect to property not Distributed because it is being held in the USA Debtors Disputed Claims Cash Reserve or the USA Debtors Contingent Claims Cash Reserve, Distributions that are not claimed by the expiration of six months from the time the Distribution is delivered shall be deemed to be Unclaimed Property and shall vest or revest in the Blitz Liquidating Trust, and the Claims with respect to which those Distributions are made shall be automatically canceled. After the expiration of that six month period, the claim of any Person or Entity to those Distributions shall be discharged and forever barred. Nothing contained in the Plan or this Blitz Liquidating Trust Agreement shall require the Blitz Liquidating Trust to attempt to locate any holder of an Allowed Claim. All funds or other property that vests or revests in the Blitz Liquidating Trust pursuant to this Section 7.7 shall be distributed by the Blitz Liquidating Trustee to the other holders of Allowed General Unsecured Claims of the USA Debtors in accordance with the provisions of the Plan and this Blitz Liquidating Trust Agreement. A Claim, and the Unclaimed Property distributed on account of such Claim, shall not escheat to any federal, state, or local government or other entity by reason of the failure of its holder to claim a Distribution in respect of such Claim.
- 7.8 Time Bar to Cash Payments by Check. Checks issued by the Blitz Liquidating Trust on account of Allowed Claims shall be null and void if not negotiated within 90 days after the date of issuance thereof. Requests for the reissuance of any check that becomes null and void pursuant to the Plan and this Section 7.8 shall be made directly to the Blitz Liquidating Trustee by the holder of the Allowed Claim to whom the check was originally issued. Any Claim in respect of such voided check shall be made in writing on or before the date that is six months after the Distribution was made. After that date, all Claims in respect of void checks shall be discharged and forever barred and the proceeds of those checks shall revest in and become property of the Blitz Liquidating Trust as Unclaimed Property in accordance with Section 347(b) of the Bankruptcy Code and be distributed as Unclaimed Property.
- 7.9 Withholding Taxes and Expenses of Distribution. Any federal, state, or local withholding taxes or other amounts required to be withheld under applicable law shall be deducted from Distributions hereunder. All holders of Claims shall be required to provide the Blitz Liquidating Trustee with any information necessary to effect the withholding of such taxes, including a completed W-9 form. In addition, all Distributions to be made by the Blitz Liquidating Trust under the Plan shall be net of the actual and reasonable costs of making such Distributions.
- 7.10 **Distributions on Non-Business Days.** Any payment or Distribution due on a day other than a Business Day shall be made, without interest, on the next Business Day.

- 7.11 No Distribution in Excess of Allowed Amount of Claim. Notwithstanding anything to the contrary herein, no Holder of an Allowed Claim shall receive in respect of such Claim any Distribution in excess of the Allowed amount of such Claim.
- 7.12 Setoff and Recoupment. The Blitz Liquidating Trust may, but shall not be required to, setoff against, or recoup from, any Claim and the Distribution to be made pursuant to the Plan in respect thereof, any claims or defenses of any nature whatsoever that any of the USA Debtors, their Estates or the Blitz Liquidating Trust may have against the holder of such Claim, but neither the failure to do so nor the allowance of any Claim under the Plan shall constitute a waiver or release by the USA Debtors, their Estates or the Blitz Liquidating Trust of any claim, defense, right of setoff, or recoupment that any of them may have against the holder of any Claim.

ARTICLE VIII TAXES

- 8.1 Income Tax Status. Consistent with Revenue Procedure 94-45, 1994-28 I.R.B. 124, the Blitz Liquidating Trust shall be treated as a liquidating trust pursuant to Treasury Regulation Section 301.7701-4(d) and as a grantor trust pursuant to IRC Sections 671-677. As such, the Blitz Liquidating Trust Beneficiaries will be treated as both the grantors and the deemed owners of the Blitz Liquidating Trust. Any items of income, deduction, credit, and loss of the Blitz Liquidating Trust shall be allocated for federal income tax purposes to the Blitz Liquidating Trust Beneficiaries.
- Tax Returns. In accordance with IRC Section 6012 and Treasury Regulation 8.2 Section 1.671-4(a), the Blitz Liquidating Trust shall file with the IRS annual tax returns on Form 1041. In addition, the Blitz Liquidating Trust shall file in a timely manner such other tax returns, including any state and local tax returns, as are required by applicable law and pay any taxes shown as due thereon out of the Blitz Liquidating Trust Assets (or the income or proceeds thereof). Within a reasonable time following the end of the taxable year, the Blitz Liquidating Trust shall send to each Blitz Liquidating Trust Beneficiary a separate statement setting forth the Blitz Liquidating Trust Beneficiary's share of items of income, gain, loss, deduction or credit and will instruct each such Blitz Liquidating Trust Beneficiary to report such items on their federal income tax returns. The Blitz Liquidating Trust may provide each Blitz Liquidating Trust Beneficiary with a copy of the Form 1041 for the Blitz Liquidating Trust (without attaching any other Blitz Liquidating Trust Beneficiary's Schedule K-1 or other applicable information form) along with such Blitz Liquidating Trust Beneficiary's Schedule K-1 or other applicable information form in order to satisfy the foregoing requirement. The Blitz Liquidating Trust shall allocate the taxable income, gain, loss, deduction, or credit of the Blitz Liquidating Trust with respect to each Blitz Liquidating Trust Beneficiary.
- 8.3 Withholding of Taxes and Reporting Related to Blitz Liquidating Trust Operations. The Blitz Liquidating Trust shall comply with all withholding and reporting requirements imposed by any federal, state, local, or foreign taxing authority, and all Distributions made by the Blitz Liquidating Trust shall be subject to any such withholding and reporting requirements. To the extent that the operation of the Blitz Liquidating Trust or the

liquidation of the Blitz Liquidating Trust Assets creates a tax liability, the Blitz Liquidating Trust shall promptly pay such tax liability out of the Blitz Liquidating Trust Assets (or the income or proceeds thereof) and any such payment shall be considered a cost and expense of the operation of the Blitz Liquidating Trust payable without Bankruptcy Court order. The Blitz Liquidating Trust may reserve a sum, the amount of which shall be determined by the Blitz Liquidating Trustee, sufficient to pay the accrued or potential tax liability arising out of the operations of the Blitz Liquidating Trust or the operation of the Blitz Liquidating Trust Assets. The Blitz Liquidating Trustee, on behalf of the Liquidating Trust, may enter into agreements with taxing authorities or other governmental units for the payment of such amounts as may be withheld. Any federal, state, or local withholding taxes or other amounts required to be withheld under applicable law shall be deducted from Distributions hereunder. All Blitz Liquidating Trust Beneficiaries shall be required to provide any information necessary to effect the withholding of such taxes.

- 8.4 Valuations. As soon as reasonably practicable after the Effective Date, the Blitz Liquidating Trustee (to the extent that the Blitz Liquidating Trustee deems is necessary or appropriate in his or her sole discretion) shall value the Blitz Liquidating Trust Assets based on the good faith determination of the value of such Blitz Liquidating Trust Assets. The valuation shall be used consistently by all parties (including the USA Debtors, the Blitz Liquidating Trustee, and the Blitz Liquidating Trust Beneficiaries) for all federal income tax purposes. The Bankruptcy Court shall resolve any dispute regarding the valuation of the Blitz Liquidating Trust Assets.
- 8.5 Treatment of USA Debtors Disputed Claims Cash Reserves. Notwithstanding any other provision of this Blitz Liquidating Trust Agreement to the contrary, subject to definitive guidance from the IRS or a court of competent jurisdiction to the contrary, the Blitz Liquidating Trust shall (i) treat any Blitz Liquidating Trust Assets allocable to, or retained on account of, a USA Debtors Disputed Claims Cash Reserve in accordance with Section 8.2 of this Blitz Liquidating Trust Agreement as held by one or more discrete trusts for federal income tax purposes, consisting of separate and independent shares to be established in respect of each Disputed Claim, in accordance with the trust provisions of the IRC (sections 641 et seq.), (ii) treat as taxable income or loss of each USA Debtors Disputed Claims Cash Reserve, with respect to any given taxable year, the portion of the taxable income or loss of the Blitz Liquidating Trust that would have been allocated to the holders of Disputed Claims had such Claims been Allowed on the Effective Date (but only for the portion of the taxable year with respect to which such Disputed Claims are unresolved), (iii) treat as a distribution from the USA Debtors Disputed Claims Cash Reserve any increased amounts distributed by the Blitz Liquidating Trust as a result of any Disputed Claims resolved earlier in the taxable year, to the extent such distributions relate to taxable income or loss of the USA Debtors Disputed Claims Cash Reserve determined in accordance with the provisions hereof, and (iv) to the extent permitted by applicable law, report consistent with the foregoing for state and local income tax purposes. All Blitz Liquidating Trust Beneficiaries shall report, for income tax purposes, consistent with the foregoing. In the event, and to the extent, any Cash retained on account of Disputed Claims in the USA Debtors Disputed Claims Cash Reserve is insufficient to pay the portion of any such taxes attributable to the taxable income arising from the assets allocable to, or retained on account of, Disputed Claims, such taxes shall be (i) reimbursed from any subsequent Cash amounts retained on account of

Disputed Claims, or (ii) to the extent such Disputed Claims have subsequently been resolved, deducted from any amounts distributable by the Blitz Liquidating Trustee as a result of the resolutions of such Disputed Claims.

8.6 Expedited Determination of Taxes. The Blitz Liquidating Trust may request an expedited determination of taxes of the Blitz Liquidating Trust, including the USA Debtors Disputed Claims Cash Reserve, under section 505(b) of the Bankruptcy Code for all returns filed for the Blitz Liquidating Trust for all taxable periods through the termination of the Blitz Liquidating Trust.

ARTICLE IX TERMINATION OF BLITZ LIQUIDATING TRUST

- 9.1 Termination of Blitz Liquidating Trust. The Blitz Liquidating Trustee shall be discharged and the Blitz Liquidating Trust shall be terminated, at such time as (a) all Disputed Claims have been resolved, (b) all of the Blitz Liquidating Trust Assets have been liquidated, (c) all duties and obligations of the Blitz Liquidating Trustee hereunder have been fulfilled, (d) all Distributions required to be made by the Blitz Liquidating Trustee under the Plan and this Blitz Liquidating Trust Agreement have been made, and (e) the Chapter 11 Cases of the USA Debtors have been closed; <u>provided</u>, <u>however</u>, that in no event shall the Blitz Liquidating Trust be terminated later than the term of the Blitz Liquidating Trust under Section 9.2 of this Blitz Liquidating Trust Agreement, as such term may be extended pursuant to Section 9.2.
- Maximum Term. The term of the Blitz Liquidating Trust shall end no later than 9.2 the third (3th) anniversary of the Effective Date (the "Initial Blitz Liquidating Trust Term"); provided, however, that the Blitz Liquidating Trustee may, subject to the further provisions of this Section 9.2, extend the term of the Blitz Liquidating Trust for such additional period of time as is necessary to facilitate or complete the recovery and liquidation of the Blitz Liquidating Trust Assets as follows: within the six (6) month period prior to the termination of the Initial Blitz Liquidating Trust Term, the Blitz Liquidating Trustee may file a notice of intent to extend the term of the Blitz Liquidating Trust with the Bankruptcy Court and upon approval of the Bankruptcy Court of such extension, the term of the Blitz Liquidating Trust shall be so extended. The Blitz Liquidating Trust may file one or more such extension notices, each notice to be filed within the six (6) month period prior to the termination of the extended term of the Blitz Liquidating Trust (all such extensions, collectively, are referred to herein as the "Supplemental Blitz Liquidating Trust Term"). Notwithstanding anything to the contrary in this Section 9.2, however, the Blitz Supplemental Liquidation Term may not exceed three (3) years without a favorable letter ruling from the IRS that any further extension would not adversely affect the status of the Blitz Liquidating Trust as a liquidating trust for federal income tax purposes. In addition, the provisions of this Section 9.2 shall be without prejudice to the right of any party in interest under Section 1109 of the Bankruptcy Code to petition the Bankruptcy Court, for cause shown, to shorten the Supplemental Blitz Liquidating Trust Term.
- 9.3 Events Upon Termination. At the conclusion of the term of the Blitz Liquidating Trust, the Blitz Liquidating Trustee shall distribute the remaining Blitz Liquidating

Trust Assets, if any, to the Blitz Liquidating Trust Beneficiaries, in accordance with the Plan, the Confirmation Order, and this Blitz Liquidating Trust Agreement.

9.4 Winding Up, Discharge, and Release of the Blitz Liquidating Trustee. For the purposes of winding up the affairs of the Blitz Liquidating Trust at the conclusion of its term, the Blitz Liquidating Trustee shall continue to act as Blitz Liquidating Trustee until its duties under this Blitz Liquidating Trust Agreement have been fully discharged or its role as Blitz Liquidating Trustee is otherwise terminated under this Blitz Liquidating Trust Agreement and the Plan. Upon a motion by the Blitz Liquidating Trustee, the Bankruptcy Court may enter an order relieving the Blitz Liquidating Trustee, its agents and employees of any further duties, discharging, and releasing the Blitz Liquidating Trustee and releasing its bond, if any.

ARTICLE X MISCELLANEOUS PROVISIONS

- 10.1 Amendments. The Blitz Liquidating Trustee may modify, supplement, or amend this Agreement in any way that is not inconsistent with the Plan or the Confirmation Order [upon prior written notice to, and consent of, the Oversight Committee]. The Blitz Liquidating Trustee shall seek Bankruptcy Court approval of any material modification, supplement, or amendment.
- 10.2 Waiver. No failure by the Blitz Liquidating Trust, the Blitz Liquidating Trustee, or its agents, professionals and employees to exercise or delay in exercising any right, power, or privilege hereunder shall operate as a waiver, nor shall any single or partial exercise of any right, power, or privilege hereunder preclude any further exercise thereof, or of any other right, power, or privilege.
- 10.3 Cumulative Rights and Remedies. The rights and remedies provided in this Blitz Liquidating Trust Agreement are cumulative and are not exclusive of any rights under law or in equity.
- 10.4 No Bond Required. Notwithstanding any state law to the contrary, the Blitz Liquidating Trustee (including any successor Blitz Liquidating Trustee) shall be exempt from giving any bond or other security in any jurisdiction.
- 10.5 Irrevocability. This Blitz Liquidating Trust Agreement and the Blitz Liquidating Trust created hereunder shall be irrevocable, except as otherwise expressly provided in this Blitz Liquidating Trust Agreement.
- 10.6 Tax Identification Numbers. The Blitz Liquidating Trustee may require any Blitz Liquidating Trust Beneficiary to furnish to the Blitz Liquidating Trustee its social security number or employer or taxpayer identification number as assigned by the IRS and the Blitz Liquidating Trustee may condition any Distribution to any Blitz Liquidating Trust Beneficiary upon the receipt of such identification number.
- 10.7 Relationship to the Plan. The principal purpose of this Blitz Liquidating Trust Agreement is to aid in the implementation of the Plan and, therefore, this Blitz Liquidating Trust Agreement incorporates and is subject to the provisions of the Plan and the Confirmation Order.

In the event that any provision of this Blitz Liquidating Trust Agreement is found to be inconsistent with a provision of the Plan or the Confirmation Order, the provisions of the Plan or the Confirmation Order, as applicable, shall control. In the event that any provision of the Plan is found to be inconsistent with a provision of the Confirmation Order, the Confirmation Order shall control.

- 10.8 Division of Blitz Liquidating Trust. Under no circumstances shall the Blitz Liquidating Trustee have the right or power to divide the Blitz Liquidating Trust unless authorized to do so by the Bankruptcy Court.
- 10.9 Applicable Law. This Blitz Liquidating Trust shall be governed by and construed in accordance with the laws of the State of Delaware, without giving effect to rules governing the conflict of laws.
- 10.10 Retention of Jurisdiction. Notwithstanding the Effective Date, and to the fullest extent permitted by law, the Bankruptcy Court shall retain exclusive jurisdiction over the Blitz Liquidating Trust after the Effective Date, including, without limitation, jurisdiction to resolve any and all controversies, suits and issues that may arise in connection therewith, including, without limitation, this Blitz Liquidating Trust Agreement, or any entity's obligations incurred in connection herewith, including without limitation, any action against the Blitz Liquidating Trustee or any professional retained by the Blitz Liquidating Trustee, in each case in its capacity as such. Each party to this Blitz Liquidating Trust Agreement hereby irrevocably consents to the exclusive jurisdiction of the Bankruptcy Court in any action to enforce, interpret or construe any provision of this Blitz Liquidating Trust Agreement or of any other agreement or document delivered in connection with this Blitz Liquidating Trust Agreement, and also hereby irrevocably waives any defense of improper venue, forum non conveniens or lack of personal jurisdiction to any such action brought in the Bankruptcy Court. Each party further irrevocably agrees that any action to enforce, interpret, or construe any provision of this Blitz Liquidating Trust Agreement will be brought only in the Bankruptcy Court. Each party hereby irrevocably consents to the service by certified or registered mail, return receipt requested, of any process in any action to enforce, interpret or construe any provision of this Blitz Liquidating Trust Agreement.
- 10.11 Severability. In the event that any provision of this Blitz Liquidating Trust Agreement or the application thereof to any person or circumstance shall be determined by the Bankruptcy Court to be invalid or unenforceable to any extent, the remainder of this Blitz Liquidating Trust Agreement, or the application of such provision to persons or circumstance, other than those as to which it is held invalid or unenforceable, shall not be affected thereby, and such provision of this Blitz Liquidating Trust Agreement shall be valid and enforced to the fullest extent permitted by law.
- 10.12 Limitation of Benefits. Except as otherwise specifically provided in this Blitz Liquidating Trust Agreement, the Plan or the Confirmation Order, nothing herein is intended or shall be construed to confer upon or to give any person other than the parties hereto and the Blitz Liquidating Trust Beneficiaries any rights or remedies under or by reason of this Blitz Liquidating Trust Agreement.

10.13 Notices. All notices, requests, demands, consents, and other communications hereunder shall be in writing and shall be deemed to have been duly given to a person, if delivered in person or by facsimile with an electromagnetic report of delivery or if sent by overnight mail, registered mail, certified mail, or regular mail, with postage prepaid, to the following addresses:

If to the Blitz Liquidating Trustee:

with a copy to:

If to a Blitz Liquidating Trust Beneficiary:

To the name and distribution address set forth in the Register with respect to such Beneficiary.

The parties may designate in writing from time to time other and additional places to which notices may be sent.

- 10.14 Further Assurances. From and after the Effective Date, the parties hereto covenant and agree to execute and deliver all such documents and notices and to take all such further actions as may reasonably be required from time to time to carry out the intent and purposes of this Blitz Liquidating Trust Agreement, and to consummate the transactions contemplated hereby.
- 10.15 Integration. This Blitz Liquidating Trust Agreement, the Plan, and the Confirmation Order constitute the entire agreement with, by and among the parties thereto, and there are no representations, warranties, covenants, or obligations except as set forth herein, in the Plan and in the Confirmation Order. This Blitz Liquidating Trust Agreement, together with the Plan and the Confirmation Order, supersede all prior and contemporaneous agreements, understandings, negotiations, and discussions, written or oral, of the parties hereto, relating to any transaction contemplated hereunder. Except as otherwise provided in this Blitz Liquidating Trust Agreement, the Plan or Confirmation Order, nothing herein is intended or shall be construed to confer upon or give any person other than the parties hereto and the Blitz Liquidating Trust Beneficiaries any rights or remedies under or by reason of this Blitz Liquidating Trust Agreement.
- 10.16 Interpretation. The enumeration and Section headings contained in this Blitz Liquidating Trust Agreement are solely for convenience of reference and shall not affect the meaning or interpretation of this Blitz Liquidating Trust Agreement or of any term or provision hereof. Unless context otherwise requires, whenever used in this Blitz Liquidating Trust Agreement the singular shall include the plural and the plural shall include the singular, and words importing the masculine gender shall include the feminine and the neuter, if appropriate, and vice versa, and words importing persons shall include partnerships, associations, and corporations. The words herein, hereby, and hereunder and words with similar import, refer to

this Blitz Liquidating Trust Agreement as a whole and not to any particular Section or subsection hereof unless the context requires otherwise. Any reference to the "Blitz Liquidating Trustee" shall be deemed to include a reference to the "Blitz Liquidating Trust" and any reference to the "Blitz Liquidating Trust" shall be deemed to include a reference to the "Blitz Liquidating Trustee" except for the references in Sections 4.3 and 4.4, and such other provisions in which the context otherwise requires.

10.17 Counterparts. This Blitz Liquidating Trust Agreement may be signed by the parties hereto in counterparts, which, when taken together, shall constitute one and the same document. Delivery of an executed counterpart of this Blitz Liquidating Trust Agreement by facsimile or email in pdf format shall be equally effective as delivery of a manually executed counterpart.

IN WITNESS WHEREOF, the parties hereto have either executed and acknowledged this Blitz Liquidating Trust Agreement, or caused it to be executed and acknowledged on their behalf by their duly authorized officers or representatives, all as of the date first above written.

USA DEBTORS:	
BLITZ U.S.A., INC,	
Ву:	
Name: Title:	
BLITZ ACQUISITION, LLC	
By:	
Name: Title:	
BLITZ RE HOLDINGS, LLC	<u>.</u>
By:	
Name: Title:	
MIAMIOK LLC (f/k/a F3 Bra	ands LLC)
manifest Ede (fival 5 bio	and EEC)
Th.	
By:Name:	
Title:	
as Blitz Liquid	ating Trustee
Ву:	
Name:	
Title:	

EXHIBIT D

Notice of Entry of Confirmation Order

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re:	Chapter 11
BLITZ U.S.A., Inc., et al., ¹	Case No. 11-13603 (PJW)
Debtors.	(Jointly Administered)
	Re: Docket Nos.

NOTICE OF (A) ENTRY OF ORDER CONFIRMING DEBTORS' AND OFFICIAL COMMITTEE OF UNSECURED CREDITORS' FIRST AMENDED JOINT PLAN OF LIQUIDATION, AND (B) BAR DATES FOR FILING ADMINISTRATIVE EXPENSE CLAIMS

AND REJECTION DAMAGES CLAIMS

TO: ALL PARTIES IN INTEREST

PLEASE TAKE NOTICE THAT:

- 1. On [January 30], 2013, the United States Bankruptcy Court for the District of Delaware (the "Bankruptcy Court") entered its Order Confirming Debtors' and Official Committee of Unsecured Creditors' First Amended Joint Plan of Liquidation (the "Confirmation Order"). Unless otherwise defined herein, capitalized terms used in this Notice shall have the meanings ascribed to such terms in the Debtors' and Official Committee of Unsecured Creditors' First Amended Joint Plan of Liquidation, dated December 18, 2013 (the "Plan").
- 2. Copies of this Confirmation Order and the Plan may be obtained by accessing http://www.kccllc.net/Blitz
- 3. The Plan will become effective in accordance with its terms on the date on which all conditions to the effective date of the Plan as set forth in section 11.2 of the Plan have been satisfied or waived as provided in section 11.5 of the Plan (the "Effective Date"). The Blitz Liquidating Trustee will file a notice of the occurrence of the Effective Date with the Bankruptcy Court, mail a copy thereof to all parties on the Master Service List maintained in these cases, and post a copy at http://www.kccllc.net/Blitz
- 4. In accordance with section 2.3 of the Plan, other than (a) Administrative Expense Claims for which the Bankruptcy Court established a previous Bar Date, including 503(b)(9) Claims and Administrative Expense Claims arising during the period commencing on the Petition Date and continuing through and including July 13, 2012, which were subject to a Bar Date of July 13, 2012, and (b) Fees of Bankruptcy Professionals, any and all requests for payment or proofs of Administrative

The Debtors in these chapter 11 cases, along with the last four digits of each Debtor's federal tax identification number, include: LAM 2011 Holdings, LLC (8742); Blitz Acquisition Holdings, Inc. (8825); Blitz Acquisition, LLC (8979); Blitz RE Holdings, LLC (9071); Blitz U.S.A., Inc. (8104); and MiamiOK LLC (2604). The location of the Debtors' corporate headquarters and the Debtors' service address is: 309 North Main Street, Miami, OK 74354.

Expense Claims must be filed with the Bankruptcy Court no later than the first Business Day that is at least forty-five (45) days after the Effective Date, unless otherwise ordered by the Bankruptcy Court. Objections to any Administrative Expense Claims must be filed by the first Business Day that is seventy-five (75) days after the Effective Date, which objection date may be extended by application to the Bankruptcy Court.

- 5. In accordance with section 8.2 of the Plan, if the rejection of a contract or lease pursuant to any provision of the Plan results in a Claim by the non-Debtor party or parties to such contract or lease, such Claim shall be forever barred and shall not be enforceable against the Estates, the Blitz Liquidating Trust, the Blitz Personal Injury Trust, their successors or properties, unless a Proof of Claim is filed and served on Blitz Liquidating Trustee or the BAH Plan Administrator, as applicable, within thirty (30) days of the notice of entry of the Confirmation Order.
- 6. As required by paragraph 55 of the Confirmation Order and the Blitz Personal Injury TDP, all materials required by the Personal Injury POC form (including medical records and expert reports) and any materials to be considered in support of a Special Circumstances claim must be sent electronically by the holders of Blitz Personal Injury Claims to the Blitz Personal Injury Trust, at blitzclaim@litigationkc.com, dhaltiwanger@rpwb.com and adr@richardcohen.net so as to be received on or before February 28, 2014 by 5:00 p.m. Central Standard Time. Materials received after this time will not be considered by the Blitz Personal Injury Trust.

Dated:	ſ	٦.	2014

RICHARDS, LAYTON & FINGER, P.A.

Daniel J. DeFranceschi Michael J. Merchant One Rodney Square 920 North King Street Wilmington, Delaware 19801 Telephone: (302) 651-7700

Counsel to Blitz Acquisition, LLC, Blitz RE Holdings, LLC, Blitz U.S.A., Inc. and MiamiOK LLC (f/k/a F3 Brands LLC)

YOUNG CONAWAY STARGATT & TAYLOR, LLP

Sean M. Beach Rodney Square 1000 North King Street Wilmington, Delaware 19801 Telephone: (302) 571-6600

Counsel for LAM 2011 Holdings, LLC and Blitz Acquisition Holdings, Inc.

WOMBLE CARLYLE SANDRIDGE & RICE, LLP

Francis A. Monaco, Jr. 222 Delaware Avenue, Suite 1501 Wilmington, Delaware 19801 -and-

LOWENSTEIN SANDLER LLP

Jeffrey D. Prol, Esq. 65 Livingston Avenue Roseland, New Jersey 07068

Counsel to the Official Committee of Unsecured Creditors